

Head Office: Università degli Studi di Padova

Department of General Psychology (DPG)

Ph.D. COURSE IN: Brain Mind and Computer Science (BMCS)

CURRICULUM: Computer Science for Societal Challenges and Innovation

SERIES: XXXVI

A USER-CENTRED APPROACH TO RELAXATION AND ANXIETY MANAGEMENT IN CLINICAL AND NON-CLINICAL SETTINGS:

THE USE OF CUSTOMIZED VIRTUAL REALITY SCENARIOS EXPERIENCED INDEPENDENTLY OR IN COMBINATION WITH WEB-BASED RELAXATION TRAINING

Un approccio incentrato sull'utente per promuovere il rilassamento e la gestione dell'ansia in contesti clinici e non-clinici: indagine del ruolo di scenari virtuali personalizzati erogati in modalità autonoma o in combinazione con training di rilassamento in modalità *web-based*

Thesis written with the financial contribution of Fondazione Bruno Kessler (FBK)

Coordinator: Prof. Anna Spagnolli

Supervisor: Prof. Caterina Novara

Co-Supervisor: Prof. Silvia Gabrielli

Ph.D. student: Dr. Susanna Pardini

The underlying principles are that therapy shouldn't just focus on diagnosing and treating disorder, maladjustment, suffering, etc. [...] Therapy should recognize, use and build a patient's existing strengths and resources rather than view the client as flawed and disordered. (Hefferon & Boniwell, 2011, p. 233) The research projects were funded by the Fondazione Bruno Kessler (Trento) based on the agreement with the Brain, Mind and Computer Science PhD program – University of Padova (art.2, comma 2, lett. a; DM 45/2013).

The purchase of hardware components (Alien Workstations, Oculus-Quest 2 tools) has been funded by Fondazione VRT (Valorizzazione Ricerca Trentina), Grant for the project "ViRLab_Salute", 2nd call of grant Next Generation 2021.

Table of Contents

ABSTRACT	7
INTRODUCTION	11
PART 1 – THEORETICAL OVERVIEW	15
CHAPTER 1. RELAXATION: A COGNITIVE AND BEHAVIORAL FRAMEWORK PERSPECTIVE	17
1.1 The Cognitive and Behavioral Theory of Relaxation1.2 The Progressive Muscle Relaxation Program (PMR): a standardized somatically oriented relaxation technique that works	17 19
1.3 Guided Imagery: an effective relaxation technique to manage anxiety and promote relaxation1.4 Complementary methods for relaxation: the bolstered effect in the PMR and Guided Imagery combined administration	.21 26
1.5 Why have PMR and Guided Imagery, among the relaxation techniques, been considered in the current Ph.D. thesis?	28

2.1 What are the advantages of Virtual Reality technology in promoting relaxation and managing	
anxiety?	30
2.2 Using Virtual Reality for Relaxation in Clinical and Non-Clinical Populations	37
2.3 The Customization of Virtual Reality Scenarios to enhance Relaxation	42
2.4 Main research gaps to be further investigated	44

PART 2 – THE RESEARCH STUDIES	16
CHAPTER 3. CUSTOMIZE REALISTIC VIRTUAL REALITY SCENARIOS FOR RELAXATION	47
STUDY 1. INVESTIGATING THE USER-EXPERIENCE OF CUSTOMIZED VIRTUAL REALI	ΓҮ
ENVIRONMENTS FOR RELAXATION: A PROOF-OF-CONCEPT STUDY	47
3.1 Theoretical background and aims of the study	47
3.2 Methodology	49
3.3 Results	.58
3.4 Discussion	67
3.5 Conclusions	.69
CHAPTER 4. CAN CUSTOMIZED, RELAXING VIRTUAL REALITY SCENARIOS ENHANCE EFFICACY IN THE TRAINING OF PROGRESSIVE MUSCLE RELAXATION AN INNOVATIVE COMPLEMENTARY METHOD	V? 71
STUDY 2. CUSTOMIZED, REALISTIC VIRTUAL REALITY SCENARIOS INTEGRATED WITH THE PROGRESSIVE MUSCLE RELAXATION TRAINING: A PILOT RANDOMIZED	
CLINICAL I KIAL AND PROOF OF PRINCIPLE	12

4.1 Theoretical background.4.2 Aims and hypotheses.	72
4.3 Methodology	77
4.4 Results	90
4.5 Discussion	132
CHAPTER 5. EXPLORING THE ROLE OF CUSTOMIZED VIRTUAL REALITY SCENARIOS FOR RELAXATION AND ANXIETY MANAGEMENT IN PATIENTS W MEDICAL CONDITIONS	TTH 140
STUDY 3. CUSTOMIZED VR SCENARIO TO MANAGE ANXIETY IN PATIENTS WITH	
COGNITIVE IMPAIRMENT: A PROOF-OF-CONCEPT AND FEASIBILITY STUDY	141
5.1 Rationale and theoretical background	141
5.2 Aims	143
5.3 Methodology	144
5.4 Results	152
5.5 Discussion	165
5.6 Conclusions	169

PART 3 – GENERAL DISCUSSION AND FUTURE PERSPECTIVES	172
CHAPTER 6. GENERAL DISCUSSION AND FUTURE PERSPECTIVES	173
6.1 Summary of results6.2 Take-home messages and future perspectives	

REFERENCES	
ABBREVIATIONS	212
ACKNOWLEDGMENTS	213

ABSTRACT

Virtual reality (VR) is widely used to treat various mental diseases and helps facilitate multisensory stimulation, sense of presence, and achievement of relaxation. Recent research shows that VR scenarios representing, for instance, visual and auditory elements of natural relaxing environments can facilitate the learning of relaxation techniques, such as Body Scan and Progressive Muscle Relaxation Training (PMR). The customization of the VR scenarios could be an element that further facilitates relaxation, allowing participants to experience more realistic emotional conditions, increasing relaxation, sense of presence, and perception of security in the virtual context. Due to a lack of research in the field the main objective of the current Ph.D. thesis is to investigate if customized, relaxing VR scenarios, used alone or in association with standardized relaxation training protocols, allow the management of anxiety. The first study (Chapter 3) investigated anxiety symptoms, user experience, and individual preferences regarding exposure to a customized and non-customized relaxing VR environment combined with a Body Scan audio track-based guiding session. The study was based on a crossover design and a mixed-method approach with a convenience sample of 20 individuals from a non-clinical population. Findings indicated that customized scenarios could have a role in enhancing the subjective perception of relaxation, immersivity, and pleasure compared to noncustomized virtual settings.

The second pilot study (Chapter 4) further investigated the role of standard exposure to relaxing Guided Imagery (GI) with PMR to enhance the perception of relaxation and the role of VR in overcoming possible limitations of in-imagination exposure compared to the usual procedure. The study is a randomized pilot study on a total of 108 individuals from a university sample. The study involved three following experimental conditions: "PMR training by Zoom & VR exposure", "PMR training by Zoom & GI exposure", "PMR training by Audio-track & VR exposure". The objective was to understand if the use of VR helps to 1) enhance relaxation and the decrease of anxiety, and 2) recall the image and being immersed in the relaxing scenario in a latter self-directed session even without use of VR. Another aim was to investigate the role of two different web-based procedures (by Zoom and based on audio-tracks) on reducing state anxiety during the first learning session of the PMR. The main results indicated that customized virtual scenario can promote a greater reduction of the self-perceived state anxiety compared to the imaginative condition, as well as to recall of the strategies learned during the relaxation protocol. Independently of the type of web-based relaxation training administered before, VR

is more effective than in-imagination exposure in allowing a subjective reduction of the state of anxiety and arousal experienced.

The last proof-of-concept study (Chapter 5) investigated the impact of customized-relaxing VR scenarios to manage anxiety on a sample of 23 participants with cognitive impairment. The primary objective was to assess the impact of VR on self-reported and observational levels of motion-sickness, engagement, and pleasantness in older adults living with cognitive impairment and residing in long term care. Distinct from most studies in this field, the current study intended to provide personalization of visual and audio stimuli over offering a selected number of outdoor settings. For this reason, the secondary aim was to investigate if customized, relaxing virtual environments can positively impact feelings and state anxiety. A third goal was to investigate the usability of the VR apparatus from the perspective of long-term care health staff and understand feasibility in clinical settings. From a general perspective, the exposure to the VR apparatus was well tolerated, and no severe or notable adverse side effects were reported. Participants expressed verbally or manifested with non-verbal behavior a general state of interest, awareness, engagement, and enjoyment during the VR exposure showing appreciation of the experience. Reports collected during the intervention showed that the exposure to a realistic, natural, and above all, personalized virtual environment adapted to users' preferences and needs, prompted reactions, such as reminiscence of melancholic memories, the aesthetic appreciation of what they were seeing, the sense of safeness and protection in an appreciated context. The general increased relaxation state and the reduction in worry/anxiety post-VR, both self-reported and observed, support the hypothesis that VR can serve as a restorative or stress-reducing experience.

The studies shed light on possible future applications of VR relaxation environments for clinical and non-clinical purposes. The customization of VR-based scenarios has been suggested as an element that may promote relaxation, and few studies have studied this topic. Furthermore, in light of the COVID-19 pandemic, innovative VR treatment protocols are increasingly requested and helpful in managing relaxation, also due to the possibility to be administered with greater flexibility, in several different contexts, and independently from the presence of the therapist. The introduction of VR and web-based training could have a positive impact on reducing psychotherapy sessions' need and cost, as it allows a partial self-management of the treatment. If the combined administration of a customized VR relaxation scenario and PMR is effective in obtaining a better subjective perception of relaxation with greater autonomy. For this reason, exploring alternative and effective ways of providing PMR could

extend the treatment administration options, especially in situations where the standard procedure is more difficult to be realized. Moreover, the third study of the present thesis (Chapter 5) contributed to the growing body of literature focusing on empirically evaluating the acceptability and potential impact of immersive VR in long-term care settings, with specific considerations for frail older adults with varying degrees of cognitive, sensory, and mobility impairments. Differently from previous work, we evaluated the effect of customizing the VR experience in this setting and use case. This represents a valuable step forward compared to the standard "one-size-fits-all" approach adopted in most of the studies in the field of VR. At the same time, while personalization could improve acceptability by accommodating and tailoring the VR environment in light of patients' preferences, it might lead to potential challenges in its deployment. Since customized and user-centred approaches require more involvement and effort by those responsible for administering the therapy, it is also critical that we seek their input on the design of the interventions to ensure their adoption and sustainability over time. In this study, we considered the viewpoints of the healthcare staff, which brought valuable insight into the patient's previous history and usual emotional patterns and reactions. Despite some limitations, this research contributes to advancing scientific knowledge related to the customization of VR experiences and their related impact on acceptability and potential clinical efficacy in frail older adults.

INTRODUCTION

The study of user experience in VR scenarios has attracted the interest of the scientific community despite, on our knowledge, most published works remaining on a theoretical level and lack empirical evidence. Furthermore, VR started to be deployed in association with different types of standard relaxation programs. These programs usually included more directive relaxation techniques that imply greater involvement of the physiological response (e.g., Progressive Muscle Relaxation), as well as fewer directive procedures which may rely mostly on the cognitive functioning control (e.g., Meditation) or the physiological feedback (e.g., Autogenic Training) (e.g., [1-6]). The Progressive Muscle Relaxation training (PMR) protocol is based on how our nervous and skeletal systems work. Over-activation of the sympathetic nervous system is primarily responsible for excessive skeletal muscle activity, and PMR is an efficacious technique in reducing activation. This process has an indirect positive effect on other physiological parameters (e.g., heart rate frequency). The training consists of learning how to tense and release several muscle groups from the bottom to the top of the body. Evidence suggests that stress-competing relaxation techniques are among the most effective training programs to enhance psychological resources and reduce psychological distress [7]. There are different types of protocols based on their duration, including abbreviated procedures consisting of 5-6 sessions [8,9]. The participant is requested to learn to discriminate between tension and relaxation of the various muscle districts and to learn, recognize, and maintain a state of relaxation. The holding and releasing activities allow proprioceptive stimulation from peripheral sensory afferents to the higher nervous system regions via the spinal cord and the brainstem as a "bottom-up" pathway [10-12]. Also based on these processes, the PMR is considered one of the most effective and adaptable techniques for relaxation, anxiety management, stress, and depression reduction considering different kinds of target populations (e.g., non-clinical individuals, patients with cancer, chronic pain, osteoarthritis, heart disease, chronic headache, anxiety, depression) [13-17]. Complementary methods that imply the administration of integrated cognitive and behavioral procedures are studied to manage anxiety, stress, depression, and pain, particularly in cancer patients. Notable is the integration of the PMR training and the Guided Imagery (GI) used to enhance and facilitate the effect of relaxation, reducing anxiety and depression symptoms [17-19]. There are several advantages to combining standard relaxation techniques with VR technologies [6]. VR tool is easily

applicable, engaging, user-friendly, and provides comprehensive testing, training, and treatment possibilities. In addition, VR technologies can be characterized as tools of accurate control and measurement [20]. Indeed, VR enables the user's full involvement, controlling for interfering external stimuli [21,22]. Based on evidence, we intended to investigate if the concurrent exposure in a customized VR relaxing scenario, rather than in the GI only, and the administration of the PMR procedure promote a better effect on anxiety reduction, allowing a sense of relaxation, engagement, and psychological well-being.

Moreover, there are controversial data about the PMR administered in vivo or via audio recording [23,24]. To the best of our knowledge, promising outcomes have been found on the efficacy of online relaxation training (e.g., Brief Mindfulness-Based Intervention, PMR; [25,26]), but no recent evidence exist regarding the comparison between training administered in a remote modality or with an audio recording. In light of the potential benefits of using new technologies in administering relaxation procedures, we also intended to explore the PMR's efficacy in a remote situation via Zoom and audio.

In Chapters 3 and 4 of the thesis, the standard relaxation techniques administered in combination with the virtual scenarios are the PMR and a short Body Scan session derived from PMR and adapted to be administered in a single session.

The background underlined in Chapter 5 is based on the growing evidence of the positive impact of being immersed in natural environments on physical and mental health [27-29]. In healthcare facilities, allowing older adults to have recreational and pleasant experiences in natural environments, especially if they have a cognitive and physical impairment, although challenging, can be beneficial [30]. However, participating in organized outdoor activities may prove difficult for this population due to their reduced motor ability, the fear of injury during movement, as a side effect of medication, and weather conditions. Overall, evidence shows that even if unable to engage directly in a natural environment, visible exposure to nature scenes can be beneficial, reducing hospital length-of-stay and drug consumption [31-33]. Based on these findings, it is desirable to identify alternative strategies for providing older adults with exposure to natural scenes while accounting for barriers to mobility, autonomy, and safety concerns. A potentially innovative method to overcome these obstacles could be the deployment of VR. Indeed, VR technologies have been demonstrated to reduce isolation and increase engagement across several populations [34]. VR systems consist of technologies that provide the user with multiple sensory inputs through, for example, visual and auditory displays and can have varying degrees of immersion.

Immersive visual information is generally conveyed through displays, where the interface allows exposure to content that can be explored in 360 degrees, as with Head-Mounted Displays (HMDs). Auditory information can be provided via headphones, some of which are embedded directly into the HMD. VR-based interventions have been successfully deployed for addressing many clinical conditions, such as specific phobias, social anxiety, panic disorder, post-traumatic stress disorder, and depressive disorders [35]. Evidence suggests that immersive virtual scenarios can improve the quality of life in older adults' in healthcare facilities, highlighting a positive impact on behavioral and psychological symptoms. VR was also found to overcome some barriers that can hinder participation in recreational activities, such as the difficulty of moving in the environment. To our knowledge, few studies have investigated the feasibility and effectiveness of using immersive virtual environments for managing anxiety and promoting a state of relaxation in populations of elderly individuals with impaired physical and cognitive functions [36-39]. Moreover, there is a lack of research on customizing virtual scenarios for therapeutic purposes. Some published research has highlighted how using VR HMD promotes the management of depressive and anxious symptoms in older people [40,41]. Appel et al. [36] concluded that exposure of people with cognitive and physical impairments to realistic and natural immersive scenarios in virtual reality through an HMD is safe and feasible.

These findings encourage future studies to investigate the role of virtual reality scenarios' customization used alone or in combination with other standardized relaxation techniques to understand if these types of novel procedures facilitate relaxation and the management of anxiety.

A brief overview of the rationale underlying the three studies is presented in Chapters 1 and 2 of the thesis. Chapter 3 reports the first feasibility study conducted, aimed at investigating the experience of users to obtain preliminary data about preferences, pleasure, satisfaction, engagement, immersivity, sense of presence and the subjective perception of relaxation and state-anxiety when exposed to a standard and a customized relaxing VR scenario. Chapter 4 presents the next phase of the research informed by the results of the previous study. It illustrates a pilot study regarding the role of relaxing customized VR environments in improving relaxation and managing anxiety if administered in integration with PMR and compared to the guided imagination technique, a standard intervention commonly administered in combination with PMR.

Chapter 5 presents a proof-of-concept study on a clinical sample in which the primary objective was to evaluate the impact of VR on self-reported and observational levels of motion-sickness,

engagement, and pleasantness in older adults living with cognitive impairment and residing in long-term care. Specifically, the current study intended to provide customization of visual and audio stimuli, offering a selected number of outdoor settings. For this reason, the secondary aim is to investigate if customized, relaxing virtual environments can positively impact feelings and state anxiety. A third goal is to investigate the usability of the VR apparatus from the perspective of health staff working in long-term care to understand better the acceptability and potential adoption of our solution in these clinical settings. The Chapter 6 provides a summary of the overall results emerging from the above-mentioned three studies, with particular focus on implications and future perspectives of using customized virtual scenarios for promoting relaxation and reducing anxiety in clinical and non-clinical populations. Overall, the current study provides a gateway for future research to explore the benefits and effects of virtual scenarios as a means of treatment and therapy in various populations.

PART 1 ⊣⊢ THEORETICAL OVERVIEW

CHAPTER 1.

RELAXATION: A COGNITIVE AND BEHAVIORAL FRAMEWORK PERSPECTIVE

1.1 The Cognitive and Behavioral Theory of Relaxation

The American Psychological Association (APA) defined Relaxation as 1) abatement of intensity, vigor, energy, or tension, resulting in the calmness of mind, body, or both; 2) the return of a muscle to its resting condition after a period of contraction [42].

Throughout history, one of the most extended theories that first explained the processes implied in relaxation is the *Arousal-Reduction Model*, based on the conceptualization of relaxation as the reduction in arousal [43-45]. Since the importance of determining other variables that allow conceptualizing relaxation, other extended models have been developed over time, suggesting an alternative idea defined as a transactional process involving not only arousal but also factors helpful in understanding the peculiar conditions that influence stressful situations (external stimuli, coping resources, cognitive appraisal) (Figure 1), included anxiety, as a negative emotion that mediate between stimuli and the physiological reactions pattern [46,47].





Source: Jensen et al. [48].

Beliefs, and cognitive components, in general, were started to be investigated as the focal points on which to act for facilitating relaxation. Smith et al. [49] identified three cognitive processes to be considered as the "active ingredients" of any type of relaxation activity: 1) *Focusing* the attention on stimuli for a defined period; 2) The ability to desist from being committed to other concurrent goal-directed activity; 3) The capacity of tolerating and accepting uncertainty, unfamiliar, or paradoxical experiences.

In general, the cognitive-behavioral model of relaxation comprises a series of subsequent steps that imply initial and gradual management of relaxation; the learning and deployment of the mentioned cognitive processes (focusing, passivity, and receptivity), the capacity to learn and distinguish the tension and relaxation states; the ability to associate and use other more symbolic strategies able to allow the learned-relaxation technique [49].

All of these cognitive and behavioral strategies for relaxation can be extended to different types of techniques that are adapted based on the type of approach, such as physical (e.g., Breathing), unrestrictive to mental strategies (e.g., Somatic Focusing, Thematic Imagery), restrictive to mental strategies (e.g., Contemplation), [50, 51].

Progressive Muscle Relaxation technique (PMR) can be included in the physical approach category since it is less demanding in terms of mental and cognitive strategies compared to techniques such as hypnosis training, and it can be considered a simple and flexible relaxation strategy [52]. The PMR fits well with the above-mentioned cognitive and behavioral features since it requires (1) focusing on and discriminating sensations of muscles tension and tension release, (2) taking a passive attitude after generating tension, and (3) tolerating temporary variegate feelings and emotions such as boredom, frustration, and discomfort until relaxation skills begin to develop [52].

On the other hand, techniques such as Guided Imagery (GI) imply more internalized and topdown processing (e.g., focusing on mental content, physical passivity, and receptivity) [49]. From a cognitive load requested perspective, contemplation, and meditation (e.g., Mindfulness-Based Stress Reduction training) should be considered among the most demanding in terms of cognitive resources [49].

Choosing between different techniques, or combining two or more relaxation strategies, is fundamental because it allows one to identify the best type of intervention according to the users' needs, preferences, cognitive features (e.g., the ability to extensively focus attention on a specific activity) and other aspects that impact the effectiveness of relaxation intervention, such as personality characteristics and emotional functioning (e.g., distress tolerance, emotional regulation). All the impacting features have been considered in choosing a relaxation intervention for each individual.

1.2 The Progressive Muscle Relaxation training: a standardized somatically oriented relaxation technique that works

Progressive Muscle Relaxation training (PMR), also called the "Jacobson Relaxation method" or "Progressive Muscle Relaxation", is an evidence-based, actively engaging relaxation technique developed by the physiologist Edmund Jacobson [53] in which the main aim is to learn tension and relaxation of the different muscles districts in order to acquire a strategy to allow relaxation and reduce anxiety and mental stress [54].

PMR involves physiological effects opposite to those carried out by psychological stress and anxiety. It is used to promote the reduction of the Sympathetic Nervous System based on evidence that overactivation of the sympathetic autonomic nervous system contributes to excessive skeletal muscle activity. The earlier studies assumed that the reduction of skeletal muscle activity allows decreasing the activity of the sympathetic nervous system, in correspondence with the negative feedback from the skeletal muscle proprioceptors to the ascending reticular activation system and hypothalamus. Then, resting muscles send slight or no feedback information from the skeletal muscle proprioceptors to central brain structures, decreasing autonomic activation (e.g., [55-56]), with a corresponding increase of the Parasympathetic Nervous System activity; in this model, research observed decreasing heart rate and blood pressure, oxygen consumption, sweat gland activity, motor-physical activity, and a variation of brain waves patterns (e.g., [57-60]).

Due to these neuro-functional processes, we can therefore specify that the PMR is based on "top-down" and "bottom-up" neuronal processing [61]. To resume, the "top-down" way sends input from the central nervous system areas to peripheral areas to contract muscles and gradually release the tension. The "bottom-up" process is involved during the active holding and releasing of bodily tension that produces proprioceptive stimulation from peripheral muscles to the brain via the spinal cord and the brainstem. This exchange of sensory, emotional, and cognitive information should help strengthen the awareness and learning of the muscle districts' subjective state of tension and relaxation so that it can be replicated more automatically to stimulate relaxation and deactivation even in conditions experienced as stressful by individuals. For a general overview of the Sympathetic and Parasympathetic systems involved in these processes, see Figure 2.





Source: Singh et al. [61].

Since PMR does not involve particularly complex cognitive resources, its application can be extended to any person with different health conditions. The PMR technique involves a number of sessions established based on the purpose and level of detail to be devoted to the musculature of each body district. The first part of the training involved sessions in which participants must actively contract and release muscles of the dedicated body district, focusing on the feelings generated focally [62]. As stated, the number of sessions can vary. For example, the initial training in all muscle groups may require more than 100 sessions over several months or even years [63]. The PMR has been modified to make it more effective and easier to apply (e.g., [55, 64]). Joseph Wolpe [65] deployed PMR as a counterconditioning method to manage phobias in the systematic desensitization technique. Also, to facilitate participant compliance during the training, subsequent revised versions of PMR were suggested by Paul [66] and formalized by Bernstein and Borkovec (e.g., [56]). Revised PMR consisted, for example, of the reduction of the tension-release cycle times, the tension-release practice considering all 16 muscle groups in every training session, and the introduction of useful metaphors to help participants understand better the rationale that underlies the PMR training (e.g., the analogy with "pendulum" movement to explain the importance of exercising tension before releasing muscles). Abbreviated forms of PMR are advantageous in terms of saving cost and time. A meta-analysis considered studies that have deployed the revised Bernstein and Borkovec protocol [55], concluding that abbreviated PMR could effectively treat various disorders [26].

The effects of the PMR technique have been widely demonstrated in numerous studies (e.g., [67, 68, 69]). For instance, empirical evidence showed that PMR is effective in reducing tension and anxiety in different types of target samples that experience anxiety related to a stressful job activity and have a medical or psychological disorder (e.g., non-clinical individuals, patients with cancer, chronic pain, osteoarthritis, heart disease, chronic headache, anxiety, depression, asthma, headache, tinnitus) (e.g., [70-75]).

1.2.1 The PMR to manage anxiety disorders

Anxiety can be defined as an emotion manifested through heterogeneous cognitive and behavioral symptoms and signs (e.g., tension, worried thoughts, and physical changes like increased blood pressure and heart rate frequency). It is considered a future-oriented, long-acting response broadly focused on a diffuse threat [76]. The core feature that sets out anxiety as a pathological condition is that it is based on unreal or distorted beliefs about the threat or worry stimuli. Moreover, anxiety goes from an adaptive emotional reaction to a pathological one when it becomes disproportionate to the stimulus and/or persists above the adaptive limits, leading to a weakened and pervasive everyday functioning [47]. Anxiety disorders are among the most prevalent psychiatric disorders, and anxiety symptoms can be observed in various mental and organic disorders. Relaxation techniques constitute one of the therapeutic approaches to managing anxiety, comprised the Jacobson's training, which is effective for multiple health conditions (e.g., 78). As stated above, one of the main applications of PMR is to manage anxious symptoms related to different diseases and stressful situations in daily life (e.g., 79-83).

1.3 Imagining natural, relaxing environments based on the Guided Imagery technique: An effective relaxation strategy to manage anxiety and promote relaxation

To our knowledge, four main theories help explain why interaction with nature benefits humans. The *Attention Restoration Theory* (ART) [84, 85], the *Stress Reduction Theory* (SRT) [86, 87], the *Perceptual Fluency Account* (PFA) [88], and the *Prospect-Refuge theory* [89].

- The *Attention Restoration Theory* assumes the presence of two types of attention processes. One is named "directed attention" and is commonly required to voluntarily direct attention on effortful tasks for extended periods (e.g., tasks during the working activity). The other is named "effortless attention", and it arises in a more involuntary, effortless, and automatic manner since intrinsically interesting or exciting stimuli induce it. It has been seen that the natural environment has the power to elicit effortless attention, enabling individuals to restore the ability to direct attention. These processes are supported by natural environments requiring less top-down directed attention and restoring the cognitive system [84,85], (Figure 3).





Source: Kaplan, S. [84].

- The *Stress Reduction Theory*, proposed by Ulrich et al. [87], suggests that natural environments allow positive emotions and decrease negative thoughts and emotions, with a consequent impact on positive changes in physiological activities. Ulrich et al. [87] argue that humans are evolutionarily equipped to respond quickly to threatening stimuli in natural settings and that individuals will be ready to quickly acquire restorative responses that may strongly apply in natural settings. Positive affect evoked by some environmental stimuli blocks negative affect and thoughts and enables more rapid, complete psychophysiological stress recovery.

- The *Perceptual Fluency Account* links our positive affective responses to natural stimuli to the ease of processing such stimuli and assumes that the restoration of attention and stress reduction are by-products of this processing fluidity [88]. In this context, is postulated that fluency induces less effortful processing and then increases positive affect, which increases attention.

- The *Prospect-refuge theory* suggests that people prefer landscapes that offer both prospects (a clear field of view) as well as places to hide [90]. Supporting this theory, research has shown that nature walks that had high prospects led to higher cognitive restoration compared with nature walks that had low prospects [91].

Figure 4. Graphical representation of the Prospect-refuge theory. A) strong refuge; B) prospect-refuge balanced; C) Prospect.



Source: Conrad (1993), unpublished Master-Thesis; Dosen, & Ostwald [89].

Also based on these framework theories, a considerable amount of research investigated both the bottom-up (e.g., perceptual features of the natural scenarios) and the top-down processes as well (e.g., individuals' features, dispositions, preferences, and needs), showing that exposure to natural environments has positive benefits for psychological health and well-being, and it constitutes an effective intervention for the management of anxiety symptoms (e.g., [92-93]).

Exposure to natural environments may not always be possible, and some contexts and situations make contact with nature more challenging, for example, living in long-term facilities, in high-density urban environments, or where therapy is undertaken in a confined room without direct access to the natural world [94]. To overcome these obstacles, different strategies are usually adopted. Considering the similarity between the imagery and the perceptual systems, a validated standard intervention that has shown to be helpful in increasing positive affective states (e.g., [95]) is the creation of mental imagery based on personalized natural and relaxing scenarios. This type of intervention is commonly known as Guided Imagery (GI), a widely structured method flexibly used in cognitive and behavioral therapy [96]. GI involves instructional guidance that invokes sensory experiences (visual, auditory, haptic, and taste-smell experiences) and allows behavioral and physiological responses [97, 98]. The instructional guidance and the strong focus on the engagement of participants help

gain greater perceptual detail of the images generated, which creates a more realistic mental representation during the relaxation exercise [99]. GI has been used as an effective intervention for anxiety by generating relaxing states through mental processes [100-103]. An explicit addition of the natural environment to the GI process might overcome difficulties in physical access to nature and enhance relaxation and deactivation. Interventions using GI of naturalistic scenarios might be an accessible and cost-effective intervention for anxiety reduction and lend support to the growing evidence of the benefits of nature on mental well-being. Due to the sensory system's elicitation, literature has highlighted that GI events could be experienced as real events [104]. Further, there is evidence that visual mental imagery and visual perception share similar representations and are similarly processed [105, 106]. Indeed, imagining scenarios compete with perception processes if both are elicited, suggesting they use common cognitive processes (e.g., visual perception is affected once a visual image is held) [95, 107-108]. Moreover, even though some sensory processes function similarly in both imagery and perception, as has been revealed [109].

Figure 5. An illustrative representation of the pattern of similarity between visual perception and visual imagery in frontal, parietal, temporal, and occipital regions.



Source: Ganis et al. [109].

Numerous studies across a wide range of populations have demonstrated a link between GI and anxiety [110-117]. The evidence-based literature assumed that imagery is compelling for anxiety symptoms because anxiety is more likely to be responsive to perceptual-sensory representation than representational systems (e.g., language) that evolved later in the developmental process. And that both imagery and the real experience facilitate similar immediate perceptual experiences that directly influence corresponding emotional brain systems [101]. In the same scientific paper, Holmes and Mathews [101] stated that imagery acts as an 'emotional amplifier' with the capacity to modify emotional states. A 2016 study by Ji et al. [98] strongly indicates that GI can be deployed as a technique to treat anxiety symptoms. It has also demonstrated that anxiety-reducing effects can emerge from the imagery of the natural world without incorporating suggestive relaxation cues, a common element in guided-imagery scripts. In one study conducted to evaluate the effects of GI on intraoperative anxiety for patients undergoing abdominal surgery under spinal anesthesia, GI techniques were shown to significantly reduce anxiety [118]. Another study investigating the effects of 20 minutes of guided imagery on preoperative anxiety significantly reduced anxiety and cortisol levels [119]. Yet another study showed that hospital nurses working during the COVID-19 pandemic trained in GI, compared to controls, significantly decreased anxiety [120].

More than that, the use of GI is revealed outside of the medical setting. One study evaluating nature-versus-urban-based guided imagery as an intervention for anxiety found a significant decrease in state anxiety among adults imagining urban and natural settings [94]. The effect was most substantial for nature-based guided imagery. Another study showed significant increases in self-forgiveness scores following seven five-minute guided imagery sessions [121]. The Nguyen et al. [94] study aimed to investigate the effect of a nature-based GI intervention on state anxiety reduction and highlighted that GI could reduce anxiety, especially if naturalistic imagined scenarios were experienced, independently of gender and the ability to imagine imagined scenarios. These results confirmed how powerful exposure to naturalistic imagery scenarios is in facilitating a state of relaxation. Highlighting the flexibility of GI usage that can be undertaken not only during therapeutic sessions, overcoming accessibility barriers of taking up psychotherapy (e.g., therapy costs), and supporting the effect of being exposed to a naturalistic environment on psychological well-being [94].

Figure 6. An illustrative representation of a Guided Imagery session with a brief description of related features and consequences due to its deployment.



Source: https://connectcare3.com/resources/guided-imagery/

1.4 Complementary methods for relaxation: The bolstered effect in the PMR and GI combined administration

Studies have demonstrated the effectiveness of complementary training that combines the effects of PMR and GI in intensifying relaxation and reducing distress, anxiety, depression, quality of life, and the perception of pain in patients with various medical conditions or during pregnancy [17,18,122-126]. This complementary approach can promote a higher sense of relaxation during training sessions, but it should also be integrated with recalling a state of relaxation during everyday activities.

The combined method introduced by Baird et al. [17] is composed of 8 steps:

1) Setting a comfortable and protected natural environment in which to relax;

- 2) Recreating an imagined scene involving different senses;
- 3) Guiding the user in focusing on breathing;
- 4) Using passive progressive relaxation techniques;
- 5) Recalling the imagined scene to direct the attention on pleasant stimuli;
- Feeling a release from pain and activation (e.g., imagining a decrease of painful sensations);
- 7) Imagining to move into the scene by continuing to use focused breathing and relaxation;
- 8) Ending the session ("Breathe deeply. When you are ready, allow the image to fade. Stretch and open your eyes").

To resume, combining PMR and GI techniques amplify the effect of shifting attention from anxious and painful stimuli, driving attention to pleasant information, and promoting relaxation. Indeed, physical relaxation makes visualization easier, reducing responsiveness to stress as it transforms stressful situations from adverse reactions of fear and anxiety into positive images of well-being [127, 128]. As GI positively impacts emotional symptoms and PMR is more physical, administering these techniques can enhance the effect on relaxation [129].

When PMR and GI are concurrently administered, the relaxation session starts with PMR to help participants focus attention and facilitate the muscles' release. Then the therapist suggests an image of a relaxing, calm, or comforting place (e.g., sunset or moonlight, sitting on a hot or warm beach, floating on the water or through space) [127].

This therapy helps to deviate the attention from the physical and psychological discomfort by staying concentrated on pleasant imagery, which can decrease anxiety and pain, reduce analgesic intake, and bring down tension, anguish, fear, heart frequency, and blood pressure, besides promoting psychological well-being, energy, and sleep (e.g., [130, 131]).

Moreover, these techniques encourage self-efficacy and active participation in the treatment process [132, 133]. Relaxation therapy combined with GI is a widely used intervention that can effectively promote clinical results [124].

Therefore, it is fundamental to investigate health interventions, also considering novel and easy-to-deploy procedures to facilitate the application of this effective relaxation training.

1.5 Why have PMR and Guided Imagery, among the relaxation techniques, been considered in the current Ph.D. thesis?

PMR and GI, other than meditation approaches such as Mindfulness, are helpful training techniques for improving physical and psychological health outcomes (e.g., 134). Since many approaches work to manage anxiety and induce relaxation, the best perspectives from which clinicians might start selecting one or more relaxation techniques are the treatment goals and the users' preferences, needs, and abilities. The PMR and the GI relaxation training are commonly experienced as more familiar and tangible than Mindfulness training by individuals that are used to having concrete thinking styles, psychiatric vulnerabilities, or cognitive impairments (e.g., 135). Moreover, previous research highlighted how the PMR is an empirically validated clinical treatment to reduce negative stress responses [24], characterized by other advantages such as the low-cost of this type of intervention, ease of teaching to users with heterogeneous abilities and socio-anagraphic characteristics [136]. Indeed, the PMR has shown successful results in various clinical populations [62, 137-139], non-clinical samples, such as corporate workers and college students [64, 140, 141], and age groups ranging from adolescents to older adults [142, 143]. As stated before, PMR studies with college students have shown positive results, including anxiety reduction [64, 138].

Moreover, several physiological and psychological benefits of PMR, such as its contribution to lowering blood pressure, cortisol, heart rate, fatigue, anxiety, and pain, have been demonstrated in previous studies [64, 139-145]. Considering evidence from literature, the current Ph.D. studies aim is to preliminary investigate the feasibility, acceptance, and the impact on psychological constructs (such as feelings, state and trait anxiety, arousal, depression) in order to inform following more rigorous studies that will be implemented considering people with medical and psychological problems (e.g., chronic pain, cancer, dementia, anxiety and stress disorders).

CHAPTER 2.

THE USE OF VIRTUAL REALITY TO PROMOTE RELAXATION

2.1 What are the advantages of Virtual Reality technology in promoting relaxation and managing anxiety?

The American Psychological Association (APA) defines Virtual Reality (VR) as a "simulated three-dimensional environment created through the memory, graphics, and processes of a computer. It is often used to create simulated environments for activities such as flying a plane or exploring space, which are expensive or dangerous to experience directly. Supporting hardware and software tools, including gloves and head monitors with real-time feedback, are often used to immerse and train humans in this virtual reality"¹. Starting from the last part of the last century, the scientific community began to consider VR a valuable and effective application for diagnosing and treating a wide range of medical and psychological problems [146-148]. The opportunity to deploy VR primarily focused on the cost-benefit analysis of this new technological tool. The cost-benefit balance is essential in evaluating psychological therapy's effectiveness [148]. Cost is about the commitment that an individual or an organization provides in terms of money, time, and emotional involvement, and benefit refers to the effectiveness of a treatment (e.g., the outcome achieved in the shortest time is possible to reach). For example, in treating anxiety, one of the gold-standard techniques used to expose individuals to managing specific phobias is the behavioral exposure technique traditionally deployed gradually, starting from the "in-imagination" to the "in vivo" exposure (Figures 1A-1B).

¹ https://dictionary.apa.org/virtual-reality

Figure 1. Illustrations of the "Imaginal behavioral exposure therapy" (Figure 1A) and the "*In vivo* behavioral exposure therapy" commonly used to treat Panic Disorder and Agoraphobia.



Source: https://cbt4panic.org/3-types-of-graduated-exposure/ [149]

In the in-imagination condition, participants learn to cope with their phobic stimuli through mental images; imagining scenarios could be considered a fairly inexpensive solution in economic and emotional terms, as it implies exposure that usually occurs in the therapist's office and does not put the person in very uncomfortable situations from an emotional point of view, but it may potentially be not effective since individuals may have difficulty imagining scenarios. On the other hand, in the *in vivo* condition, participants are gradually exposed to real phobic environments; this exposure strategy can be considered more effective, overcoming limitations that imagining has concerning effectiveness, but it is a more onerous solution since it implies higher costs in terms of money, time and emotional involvement by individuals. VR can fill these standard exposure modalities' costs and benefits ratio gaps. Indeed, VR shall facilitate exposure to phobic or stressful stimuli more than imagination because it permits to recreate an environment similar to reality and is less invasive than in vivo exposure. Therefore, it is assumed that exposure in a virtual environment can facilitate subsequent *in vivo* exposure due to its potential to facilitate immersiveness and the sense of presence, reducing the gap between imagination and reality.

This cost and benefits model can be translated considering the potentiality of using VR to learn to relax. Indeed, a fortiori due to the Sars-Cov-19 pandemic, innovative VR treatment protocols are increasingly requested and helpful in managing relaxation [150], also due to the possibility to be administered with greater flexibility in several different contexts and independently from the presence of the therapist. Delivering an entire relaxation protocol (e.g., complementary intervention strategies such as PMR and GI) in the presence of the psychotherapist could

require a considerable investment in economic and temporal terms for users. The introduction of VR could reduce the number of psychotherapy sessions needed and their cost, as it allows partial self-management of part of the treatment.

Furthermore, in the case of patients with medical problems, the integration of VR could facilitate the administration of the relaxation intervention during specific invasive treatments (such as chemotherapy). If the merged administration of a customized VR relaxation scenario and PMR is effective in obtaining a better subjective perception of relaxation than the standard procedure, it could allow the users to do relaxation with greater autonomy. For this reason, assessing the efficacy of the PMR and GI in alternative ways could extend the treatment administration, especially in situations where the standard procedure is more challenging to be realized, based on the principle that "the more vivid and realistic are the relaxing images in the imagination of the client, the easier it will be to obtain a relaxation response" [148].

The main advantages of VR can be summarized as follows [151]:

- <u>Cost</u>: VR exposure is less costly since it does not require the therapist to go to the feared place with the patient.
- <u>Availability</u>: VR permits exposure to situations that are difficult to access or less effective based on imagination. Since VR exposure does not require users to have particular abilities in imagining or visualizing scenarios.
- <u>Engagement</u>: Sense of presence, immersion, interaction, and exposure to valuable stimuli offered by VR improves the engagement of the intervention, favouring participants' adherence to the interventions.
- 4) <u>Control and Self-Efficacy</u>: Exposure to virtual reality allows nearly complete control of everything that happens in the situation experienced by the person in the virtual scenario. In addition, the therapist can still know what is happening in the virtual context. Moreover, feeling in control may increase patients' feelings of self-efficacy.
- 5) *Realism and presence*: VR exposure permits people to feel present, experiencing the VR context as real.
- 6) <u>Going beyond reality</u>: Virtual contexts allow the creation of situations or elements that may be difficult to occur in the real world.
- <u>Safety</u>: In the virtual context, users have the possibility to control the computergenerated scenario, with the consciousness that the technology can be switched off anytime.
- Privacy and confidentiality: VR offers significant advantages of increased privacy and confidentiality.

The next sections of the current Chapter will focus on clarifying virtual reality's main features and potential, proceeding in detail in describing its application in experimental protocols aimed at promoting relaxation.

2.1.1 Software and Hardware equipment components

Virtual reality integrates real-time computer graphics and various sensory inputs to create a close-to-reality computer-generated world with which users can interact. The technological components of VR are composed of *input devices*, *output devices*, and the *virtual environment* [152]. The cores of a VR system are the *simulation and rendering processors* essential for reproducing the actions that will take place in the virtual world by analysing the user input, determining the actions that will take place in the VR scenario, and creating sensory images that depict a virtual world. The main components of the architecture of a VR system are represented in Figure 1.

Figure 1. The architecture of a VR system.



Source: Virtual reality (slideshare.net)

Input devices consist in sensors and trackers that allow users to interact with the virtual environment (e.g., Tracking devices: datagloves, head-positioning sensors, embedded cameras, eye-trackers; Pointing devices: six-degrees-of-freedom mouse, trackball, joystick) [35].

Output devices provide continuous computer-generated information to the user through different sensory modalities, such as visual, auditory, olfactory, and haptic feedback, via input devices like data gloves [35]. To maximize the immersiveness and sense of presence in the

virtual environment, the graphical user interface is commonly mediated through a Head-Mounted Display (HMD), such as a helmet or goggles containing two small television screens and stereo earphones. The user can explore and navigate in the virtual world through motiontracking devices attached to the HMD, which enable the computer to adapt the field of view to the user's movements.

The *virtual environment* is the computer-generated 3D virtual scenario designed to be explored so that users can interact with virtual content.

HMDs are considered the most common immersive VR technology since they permit the focalization of the entire visual contact into the virtual scenarios world in which an internal display replaces different kinds of computer-generated stimuli [35] (Figure 2). The sensors embedded in the HMD allow the perception of the stimuli in different viewing positions.

Figure 2. Example of Oculus Quest 2-HMDs worn by one of the experimenters and Oculus Quest 2 basic equipment.



Oculus Quest 2 - basic equipment

Another advanced and expensive immersive VR technology is, for example, the cave automatic virtual environment (CAVE) [153]. Non-immersive VR systems use high-resolution monitors (desktop or laptop screens) as output devices. Still, they need help to wholly clog the visual channel of the user, and their interactive abilities could be improved (e.g., due to the lack of movement tracking). Virtual environments that are not immersive include 3D video games and 3D computer modelling applications [35].

2.1.2 Virtual contexts' core "active agents": Sense of Presence and other strictlyrelated constructs for the comprehension of VR experience

A successful virtual experience gives users a sense of presence—as though they are physically immersed in the virtual environment. This sensation is achieved by fencing real-world stimuli experienced through computer-generated stimuli. Two of the essential aspects that bear and sustain the power of VR technology are "Sense of Presence", defined as an experiential quality in virtual environments about the perception of interaction with the virtual context as to "be into the VR scenario" rather than the surrounding physical environment [154], and "Immersion" a condition more influenced by the technical aspects of a virtual system that facilitate users in experiencing a position in the virtual context close to the real world. The more the technical components facilitate immersiveness and can sensorially separate the user from the physical world, the easier it is to feel a sense of presence [155,156].

Heterogeneous perspectives exist about the Immersion construct. As stated above, some models considered immersion an objective characteristic of a virtual system [157]. Others define immersion as a psychological condition derived from the perception of being included in an environment with which to interact [158]. While other conceptualizations suggest an integrated perception in which perceptual immersion and psychological immersion are defined [159, 160].

In any case, immersion is a strictly related construct to the Sense of Presence.

Sense of Presence is a multidimensional construct composed of a dynamic interaction of multisensory data and cognitive processes [161]. One of the first research groups introducing the Sense of Presence concept was Akin et al. [162] group which defined "presence" as a condition that occurs when "*At the worksite, the operator receives sufficient quantity and quality of sensory feedback to provide a feeling of actual presence*". Slater et al. [157] defined presence as the perception of immersion in a virtual environment, which is affected by the users' attention and the subjective perception of reality of the computer-generated environment.
Slater et al. [163] also gave a preliminary conceptualization of differences between immersion and presence, defining immersion as the "fidelity about physical reality generated by the display and interaction system" [161]. Based on the Slater et al. model [164], Media and User Features are considered the two categories of variables that provide for the experience of a user's presence. Media features are characterized by "media form" and "media content" variables. In referring to the "media form", Sheridan [165] introduced three principal categories impacting presence 1) the amount of sensory information; 2) the level of control participants have on the sensor devices; 3) participants' ability to modify the virtual scenario. On the other hand, User Features are about the individual dispositions (e.g., the ability to focus attention), which are counted as inter- and intra-differences in interacting perceptual-motor abilities, mental state-traits variables (e.g., attentional selection, absorption, and dissociation processes) [166], personal needs, preferences, values, involvement [158], past experiences with VR and required tasks, expectations regarding the mediated experience, and susceptibility to motion sickness [e.g., 161]. Moreover, the sense of presence is studied from a sociocultural perspective [167] in which the features of different kinds of experiences characterized by different social contexts impact the possibility of having experiences mediated by personal and culturalmediated values.

Several studies highlighted and allowed the validation of different measures to investigate the sense of presence, starting from multidimensional structured models. Schubert et al. [168] shed light on a three-factor solution composed of "Spatial Presence", "Involvement", and "Realness". Lessiter et al. [169] identified a four-factor solution similar to Schubert et al.'s. (2001), characterized by "Physical Space", "Engagement", "Naturalness", and "Negative Effects".

Considering Sense of Presence as a subjective and internal feeling elicited by sense perceptions (*quale/qualia* in the philosophy literature), assessing it is generally considered extremely difficult [170]. For this reason, heterogeneous methods are usually used to assess this construct and can be classified into "Objective measures" (e.g., behavioral observation and psychophysiological detections) and "Subjective measures" (e.g., self-report questionnaires). Ideally, objective, and subjective features should be complementarily investigated to obtain a more comprehensive status of participants' experience while interacting with virtual environments [161].

Despite the intrinsic limitations (e.g., 171), self-reported questionnaires are essential to obtain information from the user's personal perspective also in the field of new technologies since they are generally considered sensitive and reliable [172]. To limit bias, self-reported measures

should be filled out immediately after exposure [173]. Questionnaires are commonly based on three different forms of evaluation: *scales* (e.g., from 1 to 10, what level of being there did this virtual environment offer?); *paired comparative method* (e.g., which system offered more presence?); *comparative method by similarities among distinct modes* (e.g., put this light as bright as the strength of presence you have experimented in this VR system) [174].

The *Presence Questionnaire* (PQ), developed and validated by Witmer et al. [175], is among the most used self-reported measures. The PQ comprises 32 items on a 7-point Likert scale that measures presence after using a VR tool. The questionnaire gained consistent acceptance. The PQ has four sub-scales: a) "Involvement"; b) "Sensory Fidelity"; c) "Adaptation/Immersion", and d) "Interface Quality".

Another well-known scale is the *ITC-Sense of Presence Inventory (ITC-SOPI)*. This selfreported questionnaire emphasizes the media user experience [169]. The ITC-SOPI, including 44 items measured through a five-point Likert scale, has four factors: "Sense of Physical Space": a sense of being located in a physical space depicted by the media system; "Engagement": a sense of involvement with the content of the mediated environment; "Ecological Validity": a sense of naturalness and believability of the depiction of the environment itself and events within the environment; "Negative Effects": the negative experiences associated to an immersive media, such as eye-strain, headache, sickness.

Other commonly used self-reported measures are the *UCL Presence questionnaire* [176,177], the *Reality Judgement Presence questionnaire* [178], and the *Igroup Presence questionnaire* [179].

On the other hand, objective measuring instruments are not without limits. They have the advantage of not interfering with the virtual experience and allowing it to overcome the limits given by subjective reports but investigate variables not directly related to the sense of presence since it has no physical manifestation objectively measurable [180]. Commonly, objective psychophysiological measures, performance, and postural response assessment [174]. In general, arousal is considered one of the most common physiological correlates of presence (e.g., [170]), and it is investigated through the detection of heart rate frequency and rate-variability, skin's electric conductance, reflex motor behaviors and VR event evoked cortical responses.

2.2 Using Virtual Reality for Relaxation in Non-Clinical and Clinical Populations

As mentioned before, the *Attention Restoration Theory* (ART) sheds light on how exposure to natural contexts can reduce stress, regulate mood, and positively impact daily life activities

[84, 181]. These processes are allowed by the powerful impact natural settings have in providing psychological distance from mental concerns (e.g., a sense of "*being away*"), being effortless, permitting to drive attention to interesting and valuable stimuli ("*fascination*"), supported by an environment of substantial aim ("*extent*"). VR is a technology that can help expose people to virtual naturalistic scenarios, overcoming obstacles that prevent them from visiting real natural environments. Moreover, exposure to pleasant virtual nature scenarios has been shown to provide similar positive effects to actual nature in increasing relaxation, as well as decreasing stress, arousal, and anxiety [182, 183].

Systematic reviews have shown evidence of the application of VR to promote relaxation both in non-clinical and clinical populations [184, 185].

Riches et al. [184], on 19 studies in their systematic review, globally showed that VR could be considered a feasible and acceptable tool to promote relaxation in people without a clinical diagnosis, underlined how virtual environments with pleasant, natural stimuli have the potentiality to enhance relaxation compared to control samples. Riches et al. [184] suggested that the VR tool should be useful to promote relaxation also if it is used at home and at the workplace [185, 186] by considering evidence from studies that highlighted the benefits of VR on relaxation and stress management in worker populations highly exposed to stress [187, 188]. Specifically, studies conducted on healthy young adults (e.g., college students) confirmed that VR is an acceptable, feasible, tolerated, and effective tool for inducing positive emotions and a general state of well-being, highlighting the primary role of immersion, interactivity, VE contents, and sensory modalities in improving the psychological outcomes [189-192]. Usually, the VR contents play a central role in allowing positive emotions [193], and Pavic et al. [194] underlined how most of the studies that obtained effective results involved scenarios derived from natural settings. Several studies confirmed that exposure to virtual nature increases positive emotions, contributes to decreasing negative emotions [183, 189, 192, 195-197], and positively impacts physiological arousal, further confirming its relaxing and restorative properties [183, 195, 198]. Similar benefits of real and virtual nature have been observed when using highly-immersive HMDs [195, 199].

Other studies have investigated the effect of natural virtual environment contents compared to artistic reproduction in a virtual context [200, 201] and virtual crowded urban environments [202]. The superiority of natural virtual scenarios emerged, compared to virtual art settings, in generating the "sublime" effect, heightened positive emotions, and resource restoration [200, 201]. Furthermore, compared to overpopulated urban environments, natural virtual scenarios elicited stronger positive emotions in a sample of young adults [202].

With respect to the preferred sensory modality for inducing positive VR experiences, some studies suggest that it is better to include auditory stimuli other than visual stimuli [198, 203]. Usually, auditory stimuli are composed of music and/or ambient environmental sounds (e.g., birds chirping, waves, etc.), or positive narratives (e.g., 189, 192]. The added value of olfactory and tactile stimulation has been explored in one study, but even if other variegated sensory stimuli could contribute to enhancing positive effects, it seems that auditory and visual information on their own are sufficient in VR for relaxing participants [182].

From the assessment perspective, it has been highlighted that VR's efficacy in inducing targeted emotions has mainly been investigated with self-reported questionnaires (e.g., 204, 205]. Some studies evidenced psychophysiological measures confirming VR's capacity to decrease arousal [206]. As stated, the most common physiological indices investigated in this field are skin conductance and heart rate frequency or variability. Other psychophysiological measures used are electromyography, cortisol blood/urine/saliva levels, or electroencephalography assessment [191, 198, 207, 208].

Considering clinical samples, an increasing number of studies highlighted that virtual reality could be used in various healthcare settings to promote relaxation and manage anxiety in individuals that received a medical or psychological diagnosis such as Generalized Anxiety Disorder, Post Traumatic Stress Disorder, Stress-related problems, Eating disorders, Depression, Psychosis, Attention Deficit Hyperactivity Disorder, Autism Spectrum Disorder (e.g., 209-213].

As specified in Riches et al. [185], most of these studies used a variety of nature-related virtual scenarios, such as beaches, forests, islands, mountains, lakes, and waterfalls that usually required individuals to assume a seated position wearing a headset that reproduced both audio and visual elements.

Figure 3. Example of Oculus Quest 2-HMDs worn by one of the participants during the experimental administration.



In some experimental conditions, to manage and investigate psychophysiological indices, biofeedback was connected with the virtual software and hardware piece of equipment, enabling users to change features of the virtual environment, such as the movement or intensity of a waterfall based on the individuals' actual biometric parameters (e.g., [214]).

Some studies used stressful virtual environments or exercises (e.g., mathematical tasks), against which to compare virtual relaxation environments (e.g., [215]).

In general, positive findings emerged regarding the acceptability and feasibility of using VR as a relaxation tool with clinical populations [216], and encouraging outcomes also emerged from the clinicians' perspective, since the use of VR seems to be able to reduce pressure on clinicians, and healthcare operators in general (e.g., [210]).

VR in a clinical context is generally considered a low-intensity intervention for people with a mental health diagnosis (e.g., [212]), especially if users do not have the possibility of experiencing real natural contexts [36]. Indeed, it is essential to mark how VR could be beneficial for users that lack access to real-world nature (e.g., people who live in healthcare facilities). A growing number of studies have investigated the use of VR for elderly users [217]. In healthcare facilities, allowing older adults to have recreational and pleasant experiences in natural environments, especially if they have a cognitive and physical impairment, although challenging, can be beneficial [30]. Participating in organized outdoor activities may be limited by reduced motor ability, the fear of injury during movement, as a side effect of drugs, and weather conditions. Overall, evidence shows that even if unable to engage directly in a natural

environment, visible exposure to natural scenes can be beneficial, reducing hospital length-ofstay and drug consumption [30-33]. Evidence suggests that immersive virtual scenarios can improve the quality of life in older adults' healthcare facilities, highlighting a positive impact on behavioral and psychological symptoms [218]. Brimelow et al. [219] used a VR immersive scenario with users living in a residential facility characterized by heterogeneous cognitive impairment levels. The authors reported minimal side effects associated with using VR and a positive impact on the psychological well-being and pleasure related to use.

Indeed, some published research has highlighted how using immersive virtual scenarios promotes managing depressive and anxious symptoms in older people [36, 220]. Appel et al. [36] investigated the feasibility of using relaxing virtual environments to promote positive emotions, involvement, and a state of general well-being, as well as managing negative emotions such as boredom, apathy, anxiety, and a sense of isolation. Specifically, discomfort and collateral symptoms potentially associated with using the HMD (such as nausea, headache, eye discomfort, and interference with other medical devices worn) have been assessed, such as the preference for audio and visual stimuli in the VR scenario, their involvement within the virtual context and the impact on the state of anxiety. All participants completed the research without experiencing or reporting adverse side effects related to using the HMD. From a qualitative point of view, positive feedback is associated with the experience and the perception of a greater state of relaxation.

The authors concluded that exposure of people with cognitive and physical impairments to realistic and natural immersive scenarios in virtual reality through an HMD is safe and feasible. These findings encourage future studies to investigate the role of VR scenarios' personalization.

Additional research that inspired the current study was conducted by Moyle et al. [221]. The authors deployed a peculiar VR context named "Virtual Reality Forest", designed to expose users to an immersive context based on an interactive screen to improve the quality of life of people with dementia living in care facilities. In this study, the main aim was to obtain information about users' moods, apathy, and engagement. The scenario was composed of visual and audio stimuli. During the exposure, participants had the possibility to select seasons (spring and autumn) and different types of virtual objects through hand and arm movements. Residents, family members, and staff received positive feedback globally. Participants experienced higher levels of engagement, pleasure, activation, and fear during the exposure to the virtual forest than the control sample. Moreover, the authors affirmed that differences in users' preferences should be considered a main factor limiting the versatility of the VR Forest.

Overall, several studies confirm VR's efficacy for inducing positive emotions and relaxation in healthy middle-aged, elderly users [222, 223], or among residents of long-term healthcare facilities [36, 219, 221]. In order to induce relaxation and manage anxiety and arousal, most studies deployed natural-based virtual environment contents [36, 219, 221, 224], considering their identified benefits and safety of use [36, 202].

Moreover, Riches et al. [185] highlighted that VR seems to have short-term effectiveness in promoting relaxation, and it is more or equally effective if compared to non-virtual relaxation training in reducing stress, suggesting that VR relaxation may complement existing relaxation techniques or offer users other relaxation and stress reduction options.

Indeed, VR can be administered as a standalone tool or combined with other cognitive behavioral therapy strategies with the aim to tailor a treatment to the needs and preferences of each user as a way of creating a treatment that is also sustainable over time.

Even if positive outcomes emerged from using VR as a relaxing tool, there are few studies on customizing virtual scenarios for therapeutic purposes [3]. Moreover, even if studies showed positive outcomes [1, 225, 226, 227], little is known about the impact of the combination of customized virtual reality scenarios and standardized relaxation techniques on users (e.g., Progressive Muscle Relaxation, Mindfulness-Based Stress Reduction, Body Scan) (e.g., 1, 184, 209). In the next paragraph, we report the main evidence-based research on the impact of customized virtual environments to promote relaxation and manage anxiety.

2.3 The Customization of Virtual Reality Scenarios to Enhance Relaxation

With the choice of naturalistic scenarios, customization of the virtual reality context is considered a key approach that promotes relaxation [3], enabling users to experience environments closer to their preferences and needs, increasing their sense of presence, and security [228]. A recent study proposed a personalized and customized VR model for relaxation based on critical variables (e.g., emotional states of people) that were modelled and controlled by an algorithm to customize the VR environment according to users' needs and profiles [5]. However, to the best of our knowledge, few studies have explored the role of the personalization of VR-based scenarios on the user experience and the effect of the integration of relaxation training in VR. One of the most recent studies in this direction showed that being exposed to a virtual environment for relaxation, including a body-scan-guiding audio track, may help improve the participants' relaxation compared to a VR condition including a breathing control audio-track [1].

As stated by Pizzoli et al. [3], a personalized and customized VR context implies the definition of a peculiar "design of the VR environments" that, by starting from the valuable personal life events of users, generates perceptive features to be included in the VR environment to facilitate engagement, reminiscences, and general positive feelings in the user. Such descriptions should be accurate, carefully recorded, and should include multi-sensory details: visual, tactile, auditory, and even olfactory elements, based, for example, on picture, audio, and music stimuli.

Previous studies have shown that embedding autobiographical stimuli into the VR environment helps to promote a subjective and vivid emotional connotation of the experience strengthening the sense of presence [182, 189], and that different emotional states (e.g., relaxation and joy), were evoked through autobiographical stimuli presented in different sensory modalities (picture, audio, music, autobiographical recall), suggesting that autobiographical contents in VR may play a valuable role in subjective emotional experience [3]. In a customized VR environment, the element of recalling the previous safe place would enhance feelings of security and peace, while in emotion regulation training, providing affective connotated stimuli could help emotion regulation and arousal modulation by giving the user the occasion to confront their relevant life events [4]. Verbal instructions can guide relaxation, such as bringing attention to muscle activation and breathing [e.g., 53] or induced through cognitive representations of positive thoughts and stimuli [229, 230] and the customization of VR can be changed based on the relaxing technique employed in the specific intervention, rather it would stress a user-centered environment and autobiographical engagement [4] to enhance the sense of presence and affective involvement.

One of the main advantages of customized virtual scenarios is that they can be deployed in a flexible manner and integrated with the relaxing technique employed [4]. Furthermore, in a relaxing environment, customizing the elements in the virtual scenarios based on preferences and inspiration to safe places visited by the user before can improve the sense of security in the VR context [4].

From a methodological standpoint, the customization of the VR environments should be based on the user experience field to provide information to design virtual scenarios coherently with users' needs [228].

As written above, relevant properties of personal memories would be organized in terms of macro and sub-categories (e.g., perceptual aspects such as form and colour of objects; content aspect, such as the meaning to be communicated through discourses present in the VR; etc.) and the system would operate such properties dynamically during any VR instance, to maintain

a set level of immersion and the emotional reaction by the user, continually monitored through psychophysiological indexes.

To briefly recap, considering the above-mentioned premises, studying the user experience related to customized VR scenarios to promote relaxation has attracted interest in the scientific community although most of the studies published so far adhere to a theoretical perspective. In the following chapters, novel studies investigating the feasibility and applicability of customized virtual scenarios for relaxation will be illustrated to further understand the role of the customization component in VR scenarios.

2.4 Main research gaps to be further investigated

As described in this overview of the recent state-of-the-art studies, due to its descriptive potential, virtual reality technology is widely deployed in the mental health field [35]. The theoretical benefits of virtual reality-based therapy are clear, particularly for anxiety disorders. In this context, VR is valuable for providing exposure to natural and relaxing scenarios to improve relaxation and reduce distress and anxiety (e.g., [183]). Shreds of evidence in this field shed light on its application for clinical and non-clinical populations (e.g., [184, 185]). In general, the use of VR is helpful in overcoming several barriers that standard relaxation procedures present since it is less costly, promotes the availability of relaxing content that could be difficult to generate in imagination, it is able to promote the sense of presence, immersion, and engagement closer to what can be obtained in a real situation but in a safer context [151]. Moreover, VR technologies have the potential to offer a customized experience in virtual scenarios tailored to the needs and preferences of users. This user-centered approach can amplify the positive effect related to the virtual experience, allowing individuals to experience more realistic emotional conditions, like those experienced in everyday life, enhancing relaxation, the sense of presence, and a perception of security in the virtual context (e.g., [3,4]). Jeong et al. [215] preliminary study is one of the first that investigates the feasibility of an innovative VR-based relaxation program in which diaphragmatic breathing and PMR were deployed using a VR application focused on learning the relaxation strategies, reviewing, and doing exercises on what has been learned and practice in an outdoor VR scenario. Outcomes revealed that the virtual program deployed was feasible and shed light showing that, after PMR, the level of tension of participants in the VR condition was significantly lower than what was observed in the control group, suggesting that the relaxation program can be used by healthy

individuals to practice and improve relaxation. Even if at a preliminary stage, the literature showed promising evidence inducing the structuring of research protocols in which virtual reality is included as a means to facilitate the delivery of relaxation protocols that work but can be hindered by various difficulties, including duration and cost. VR-based training can reduce the cost of treatment deployed in a standard modality, increasing the number of training scenarios thanks to the possibility of customizing and introducing elements that arouse a sensory stimulation more similar to reality than can be achieved with guided exposure in the imagination. Moreover, the use of VR and web-based training allow the users to learn from their personal site [231], avoiding the necessity of moving to the therapist's office. Indeed, in term of cost and site, web-based training are useful and effective in promoting self-training or rehabilitation (e.g., [232]) also because can be used in a safe context in which sessions can be accessible in a repetitive manner, whenever users like to practise, facilitating the learning process [215]. Based on current evidence-based studies, future research should further investigate the psychological benefits of natural virtual environments in VR relaxation, taking into account the variety of possible natural stimuli available [184]. Moreover, considering the lack of studies in this field, it is useful to further investigate VR's greater effect in reducing anxiety and promoting relaxation compared to GI.

The next chapter illustrates the results of three different studies aimed at investigating the feasibility and preliminary effectiveness of human exposure to personalized and customized VR scenarios to manage anxiety and promote relaxation. The VR scenarios were experienced alone or integrated with standardized relaxation techniques with the intent to increase and facilitate the deployment of interventions overcoming difficulties or gaps that characterize standard treatments.

PART 2 ⊣⊢ THE RESEARCH STUDIES

CHAPTER 3.

CUSTOMIZED REALISTIC VIRTUAL REALITY SCENARIOS FOR RELAXATION

The current Chapter outlines the first study of the Ph.D. program published in the *International Journal of Environmental Research and Public Health* journal [233].

The research is a Proof-of-Concept study characterized by a cross-over and mixed-methods design primarily aimed at exploring the user experience of a sample of users from the general population exposed to customized and non-customized naturalistic virtual reality scenarios both integrated with a single-session Body Scan technique derived from the PMR. Furthermore, the current study had the objective of preliminary investigating if a customized VR context helps to support engagement, positive emotions, and in decreasing state-anxiety. As stated in Chapter 2, the potentialities that the customization of auditory and visual virtual

elements have in increasing relaxation, engagement, sense of presence, and security have been indicated by previous research. However, the lack of studies in this direction led us to conduct further investigation aimed at extending the state-of-the-art on the impact of customized VR scenarios, suggesting the role of customization as a promising component in facilitating engagement and promoting a higher perceived state of relaxation.

STUDY 1. Investigating the User-Experience of Customized Virtual Reality Environments for Relaxation: A Proof-of-Concept Study

3.1 Theoretical background and aims of the study

As mentioned in Chapter 1, different types of techniques and protocols enabling relaxation exist (e.g., Progressive Muscle Relaxation Technique, body scan, deep breathing, and meditation) and their contribution in promoting a general state of well-being is well recognized (e.g., 234). The Body Scan is a shorter and lower-cost technique based on progressive and gradual learning of moving attention to different parts of the body [235], with a significant impact on the promotion of somatic awareness and the management of distress (e.g., 236).

Even if the Body Scan technique is easy-to-use, standard relaxation procedures in a real context may be less practical to realize in several circumstances, and data suggest that VR can facilitate the learning of relaxation techniques through VR exposure to natural environments [1].

Digital interventions allow the reduction of the costs related to standard psychotherapy treatments (e.g., [237]), and highlight that being exposed to a virtual relaxing environment, including a body-scan-guiding audio track, may help improve the participants' relaxation [1].

Based upon this assumption, we considered that the users' perceptions need to be more deeply assessed. A systematic study of the user experience and usability related to the use of a technological tool is essential to understand and enhance the deployment of hardware and software components (e.g., [238]). The assessment of subjective experience and usability includes various elements that need to be tailored to the intended use of the virtual environment [238]. First, motion simulator disease must be evaluated as an element that can affect the VR experience and could include headaches, salivation, eye strain, nausea, and a general state of discomfort. The assessment of features related to the quality of VR graphics (e.g., resolution, movement of visual elements, shapes, colour contrast), physical performance, and synchronization (e.g., the delay between the user's movement into the virtual context and what the user expected to perceive), as well as user interface layouts is essential since they could influence the subjective experience. The assessment from the users' perspective helps in providing direct feedback from individuals and it permits obtaining different types of information, such as guidance support and a sense of presence [239]. Stanney et al. [240] suggested using the Multi-criteria Assessment of Usability for Virtual Environments (MAUVE) to systematically evaluate users' experience and usability in the VRE context. This evaluation system includes ten criteria divided into two main categories.

The Virtual Environment System Interface comprises:

- *Interaction features*: 1) "Wayfinding" (the possibility for the user to know their position and direction in the environment); 2) "Object Selection"; 3) "Manipulation";
 4) "Navigation" in the virtual environment.
- Multimodal System Output, including "Visual", "Auditory", and "Haptic" features.

The Virtual Environment User Interface is, in turn, divided into:

- *Engagement* (such as sense of presence and immersivity).
- Side Effects (comfort, sickness, after-effects).

To further understand the role of the customization component in VR scenarios, the main aim of our exploratory study was to investigate the experience of users to obtain preliminary data about preferences, pleasantness, satisfaction, engagement, immersivity, sense of presence, and the subjective perception of relaxation and state-anxiety when exposed to a standard or a customized relaxing Virtual Reality Environment. Inspired by the user experience's approach described above and to generate empirical evidence both from qualitative and quantitative data, we obtained information about the physical discomfort related to the heaviness or encumbrance of the head-mounted display, the simulator-sickness, the wayfinding, and the experience of the auditory and visual elements, both in the standard and customized scenarios. The study design was based on a mixed-methods approach inspired by the Obesity-Related Behavioral Intervention Trials (ORBIT) model since it helps in focalizing on the administration of behavioral treatment development at their pre-clinical and pre-efficacy phases. Indeed, the ORBIT framework [241] is generally used for the design (Phase Ib) of digital interventions and their preliminary testing (Phase IIa). In this phase, a proof-of-concept implementation of the customized VR scenario was realized, and preliminary testing was completed to investigate the level of engagement and the user experience with a convenience sample of non-clinical individuals. In this context, it is useful to apply a cross-over experimental design characterized by a within-subjects method where users act as their controls in a pre-post comparison. The sample size can be small because the focus is mainly on the user experience outcome, and sample size estimation is not required. Moreover, the sample can be selected from accessible individuals since this phase will aim to understand if the intervention deserves an increased depth of analysis, improvement, and further testing.

We hypothesized that: (1) relaxing VR scenarios promote relaxation and reduce stateanxiety symptoms; (2) the possibility of customizing the VR environment can be perceived as more engaging for users. Following a human-centered approach to enhancing the effect of standard interventions, we present below the main outcomes of our design and preliminary testing phases.

3.2 Methodology

3.2.1. The Virtual Environment Design

The hardware equipment was composed of an "Alienware m15 Ryzen Edition R5" workstation (CPU 5900HX, 32Gb ram, 1Tb SSD RTX3070 8Gb), a link cable (PC VR; USB 3 Type-C–

5m), and an Oculus Quest 2[®] head-mounted display connected to the workstation (see Figure 1).

Figure 1. Hardware equipment (Fig.A. Oculus Quest 2[®] head-mounted display; Fig.B. USB 3 Type-C–5m; Fig.C. Alienware m15 Ryzen Edition R5).



The Oculus Quest 2 was adopted because it is cheaper and easier to use than similar devices. This goes in the direction of our mission to expand the use of this framework for psychological treatment in daily life. Since, as described in Chapter 2, the auditory and visual stimuli in a virtual context are good enough to obtain a satisfactory relaxation state [242], the software environment was characterized by a series of 360° natural environments composed of auditory stimuli drawn from the freesound database², and the visual stimuli were developed based on Unity®³, Polyhaven⁴, and HDRIhaven⁵ (in Figure 2 a picture depicted the ongoing process of making one of the virtual contexts).

² https://freesound.org/

³ https://unity.com/

⁴ https://polyhaven.com/

⁵ https://polyhaven.com/hdris



Figure 2. A picture depicted the ongoing process of making one of the virtual contexts.

The VR environment was developed by the Digital Health Research Lab at Fondazione Bruno Kessler (FBK) Research Center (Trento, Italy). Figure 3 is a picture of the experimenters throughout the development process of the decision tree characterized by all of the functions that are visible on the PC and were useful to select the customized audio and visual stimuli.

Figure 3. The experimenters during the Decision Tree process.



Based on this set-up, users had the possibility to choose to have their preferred relaxing experience in one of three realistic scenarios that may be selected and customized with the support of a technical operator: a mountain, a marine, and a countryside environment. To build the virtual environments to allow a state of relaxation, we were inspired by indications from

the literature on the main features to follow to build a relaxing context that promotes a perception of well-being (e.g., 1, 195, 243).

For each selected environment, the participants had the option to customize a series of subcategories of components, including different types of sounds (e.g., animal sounds, wind rustle, type of music), visual stimuli (e.g., the presence of people, objects, or animals), the user position in the virtual space, the presence of people, and to change the time of day and weather conditions (examples are shown in Figures 4 and 5).

Figure 4. Example of the marine virtual space from the PC view with the series of icons related to the different customizable variables.



Figure 5. Examples of the different scenarios showed through the Oculus Quest 2 device.





The technical operator and the experimenter could personalize virtual scenarios based on a PC's dashboard interface connected to Oculus Quest 2 via a cable. Specifically, the interface presented a series of icons related to the different customizable variables (such as, music, wind, weather conditions, time of day, and presence of people) (Figure 4). The connection between the PC and the viewer allowed the operator to see what the participant was observing during the experience.

The natural VR scenario was integrated with a Body Scan audio track provided by the HMD that focused on the different parts of the body and sensations experienced in a gradual sequence. The audio track was adapted from the body scan exercise as part of the Mindfulness-Based Stress Reduction program and the PMR [9, 244]. The track lasted 7 min. The body scan technique was deployed in the current research by following the results of Pizzoli et al. [1], which showed that this technique had a more effective impact on relaxation than breathing exercises.

3.2.2. Participants Recruitment and Procedure

The current study is based on a convenience sample of individuals from the general population and was executed between 10 August 2021 and 25 September 2021. The recruitment phase is based on a snowball sampling method, carried out via email. All experimental sessions took place at the Center for Health and Wellbeing–Fondazione Bruno Kessler, Trento (Italy). The procedure phases are resumed in Figure 6.

A total of 27 individuals agreed to participate in the study. After applying the exclusion criteria, 20 were included in the analysis. The exclusion criteria were: 1) participant had a mental disorder diagnosis; 2) a score on the *State-Trait Anxiety Inventory-Y2* (STAI-Y2) indicating moderate or severe trait-anxiety levels (*raw cut point* > 50; [245, 246]; 3) a score > 9 on the *Depressive Anxiety Stress Scale-21* (DASS-21) - Depressive Scale; 4) a score > 6 on the *DASS-21 - Anxiety Scale*; 5) a score > 10 on the *DASS-21 - Stress Scale* [247, 248]. Seven

of the participants were excluded since they did not meet the inclusion and exclusion criteria. In phase T0, individuals signed the informed consent to participate in the research. They also filled out a socio-demographic survey and a series of self-report scales to investigate distress, depressive and anxiety symptoms, the perceived relaxation level, and emotional states. Then, the participants were invited to wear the head-mounted display to be exposed to a short bodyscan-guided relaxation session either with a default-standard VR condition or a customized Virtual Reality Environments (VREs). The order of presentation of each type of VR condition was randomized, and a mixed-methods approach was deployed to collect data. The customized VR experience was personalized according to the participant's preferences on the type of environment, music background, meteorological conditions, presence of objects or other participants, and daytime (first administration phase). Then, in phase T1, the individuals filled out the measures administered during phase T0 to measure their state of anxiety and emotions, in addition to an ad-hoc questionnaire deployed to collect evidence on usability, sense of presence, immersion, engagement, and VRE-related symptoms. Afterward, they were exposed to a second VR experience in a different virtual environment. During phase T2, the participants filled out the same self-report tools of phase T1. Then, the researchers asked the participants what VR conditions they preferred and if they had suggestions for improvements. The assessment procedure took about 45–50 min to be completed, the VRE personalization took approximately 10-15 min, and the Body Scan session lasted about 7 min. All participants knew they could ask to interrupt the experience at any stage (see Figure 6).

Figure 6. Flowchart of the procedure.



<u>Notes</u>: DASS-21=Depression, Anxiety and Stress Scale-21; STAI-Y=State-Trait Anxiety Inventory-Y1 and Y2; SAM=Self-Assessment Manikin; VAS=Visual Analogue Scale.

3.2.3 Materials used in the assessment phases

To investigate the study's primary constructs, we relied on the following self-report questionnaires and open-ended questions, typically used to measure VR scenarios, to collect quantitative and qualitative data:

- The *Socio-demographic survey* consisted of the following variables: gender, age, nationality, mother tongue, level of education, marital status, employment status, type of medical disorder/illness/disease (such as some neuromuscular disorders), or a psychological problem previously diagnosed. Participants were also asked to report if they were taking drugs, had already experienced relaxation activities to manage anxiety in the past, or were familiar with the use of virtual reality devices.
- The *Ad Hoc Questionnaire for Virtual Reality* investigating different aspects of Virtual Reality scenarios and User Experience was composed of the following sections:
 - 1) The Virtual Reality Symptom Questionnaire (VRSQ) [199, 249] is a dichotomous measure aiming to assess symptoms of cyber sickness. Eight items investigate general cybersickness effects, such as fatigue, headache, nausea, and concentration difficulties, and the other five are about visual effects, for example, blurred vision and tired eyes;
 - 2) The 36 items, based on a 7-point Likert scale, were inspired by the *Presence Questionnaire and the Immersive Tendencies Questionnaire* [158] and offered a qualitative set of information. The items investigating aspects, such as the realism of the environments (13 items) (e.g., "*How natural did your interactions with the environment seem*?"), engagement (three items) (e.g., "*I felt involved by the visual elements that surrounded me within the virtual environment*"), immersiveness (nine items) (e.g., "*How immersed were you in the virtual environment experience*?"), the tools' usability and quality of the interface (six items) (e.g., "*How much did the control devices interfere with the performance of assigned tasks or with other activities*?"), and emotional states (eight items) (e.g., "*During the experience, I felt tense*"), were modified to be adapted to the virtual reality environment;
 - 3) At the end of the entire procedure, an open-ended questions assessed if the participant preferred the standard or the customized VRE. Cronbach's alpha for each sub-group was calculated (0.69<Cronbach's alpha<0.88); 4) We

administered the ad hoc satisfaction questionnaire to assess the usefulness of the customized VRE (cVRE) body scan procedure; specifically, based on a dichotomous question, the participants were asked if they considered the body scan audio track helpful or not in promoting relaxation, and two open-ended questions were asked to investigate criticisms or suggestions for improvements.

- The *Depression, Anxiety and Stress Scale-21 (DASS-21)* [247, 248] is a self-report questionnaire of 21-items, seven items per subscale, investigating depression, anxiety, and stress. Items are answered according to the presence and intensity of symptoms in the last week on a 4-point Likert scale (0–3). Based on the original version, DASS demonstrated good internal consistency. In addition, psychometric analyses of the Italian samples showed that the questionnaire investigated general distress plus three additional orthogonal dimensions (anxiety, depression, and stress) and demonstrated good internal consistency, temporal stability, and criterion-oriented, convergent, and divergent validity. The DASS-21 reliability in this study population was adequate, with a Cronbach's alpha of 0.90 for the total score, 0.78 for the depression scale, 0.71 for the anxiety scale, and 0.90 for the stress scale.
- The State-Trait Anxiety Inventory-Y (STAI-Y) [245, 246] is a self-report questionnaire comprising two different 20-item scales investigating state- and trait-anxiety. State-Anxiety could be defined as a transitory anxiety reaction to an event, or a condition perceived as adverse, characterized by feelings of tension, apprehension, nervousness, and worry. Trait-Anxiety is, instead, a more stable feature related to perceiving stressful situations, such as dangerous or threatening situations in general, or as a frequency response in abnormal conditions. Each item is evaluated using a 4-point Likert scale (1–4). The total score for both scales ranges from 20 to 80, with higher scores indicating more severe anxiety. For the Trait-Anxiety score (investigated with the STAI-Y2), Cronbach's alpha was equal to 0.80, indicating a good internal consistency. Regarding the State-Anxiety (measured with the STAI-Y1), Cronbach's alpha indices for all three of the administrations ranged from 0.84 to 0.94.
- The *Self-Assessment Manikin (SAM)* [250, 251] is a non-verbal imagery-based assessment technique that directly measures the pleasure, arousal, and dominance associated with a person's affective reaction to various stimuli. The pencil-and-paper version required placing an "X" either on or between each of the five figures (resulting in a 9-point scale). The meaning of each scale is described to respondents, and they are asked to place the "X" on the figure (or between the figures) that best represents how

they currently feel. *Valence* is depicted from positive (a smiling figure), to neutral to negative (a frowning figure). Arousal ranges from high arousal (eyes wide open) to low arousal (eyes closed). When the same figures were used, the arousal scale also depicts the intensity of arousal with additional imagery over the abdomen area that ranges from high intensity (imagery representing an explosive-like burst) to low intensity (imagery representing a tiny pinprick). Finally, dominance/control ranges from feeling controlled or submissive (a petite figure) to feeling in control or dominant (a huge figure).

The Visual Analogue Scale (VAS) measured relaxation and sense of presence before and after each VRE. Specifically, the participants had to express how relaxed they felt on a line of 10 cm in length (VAS_relax: 0 = not at all relaxed; 10 = completely relaxed; VAS_sense of presence: 0 = absent; 10 = complete). These data collection methods were deployed in parallel for a holistic and multidimensional understanding of the VR experiences. In Table 1, the data collection methods and their analysis are summarized.

Fable 1. Summary	of measuring	instruments and	data analysis	performed.
------------------	--------------	-----------------	---------------	------------

Data collection method	Analysis captured
SOCIO-DEMOGRAPHIC SURVEY	<i>Descriptive analysis.</i> Frequencies, means, and standard deviations related to socio-demographics variables.
VIRTUAL REALITY SYMPTOM QUESTIONNAIRE	<i>Descriptive analysis.</i> Frequencies related to cybersickness variables.
QUESTIONNAIRE INSPIRED BY THE <i>Presence</i> <i>Questionnaire</i> and the <i>Immersive Tendencies</i> <i>Questionnaire</i>	Descriptive analysis. Means and standard deviations. ANOVAs repeated measures to compare the self-reports scores taken at two different time points.
OPEN-ENDED QUESTION – VR SCENARIO PREFERRED	<i>Descriptive analysis.</i> Thematic analysis, Frequencies related to what VR scenario has been preferred.
AD HOC SATISFACTION QUESTIONNAIRE	Descriptive analysis. Frequencies.
Depression, Anxiety and Stress Scale-21 State-Trait Anxiety Inventory-Y Self-Assessment Manikin Visual Analogue Scale	ANOVAs repeated measures and Paired-samples t-tests.

3.3 Results

Anonymized data were processed by using SPSS Statistics version 27 software [252]. Based on the DASS-21 and the STAI-Y distributions, the range of skewness and kurtosis was from -0.974 to 1.408, falling between -2 and +2. For this reason, data are reasonably normally distributed [253, 254]. Moreover, the Kolmogorov–Smirnov and the Shapiro–Wilk tests were carried out for the DASS-21 total score, the STAI-Y2 total score, the STAI-Y1 pre-VR

exposure, the STAI-Y1 post-standard VR exposure, and the STAI-Y2 post-personalized VR exposure; data were not statistically significant (*p-values*>.05), and this constitutes further evidence of the normal data distributions. First, Cronbach's alpha was performed for each selfreport questionnaire's subscales. Secondly, the frequencies, means, and standard deviations were calculated to explore the sociodemographic features. To control the order effect of the VR scenarios (standard VRE versus cVRE), multivariate ANOVAs were conducted considering all of the questionnaires administered at the two different time points after the standard and cVR experience. Descriptive data about users' experiences, preferences, and engagement were derived from the frequencies. Moreover, the participant's responses to the open-ended question relating to the preferred VR scenario were analyzed by SP and SC using thematic analysis and reported as frequencies. Data were analyzed thematically following an inductive, data-driven approach based on the procedure outlined by Braun and Clarke [255]. The data codes were consistently generated, grouped into themes, and applied to the data set to generate frequencies. ANOVAs repeated measures were calculated for the psychological scales assessed at different time points (T0, after the standard or customized VRE). Paired-sample ttests were performed to compare the self-report scores taken at three different time points (T0, after standard or customized VRE) and to investigate the impact of the VRE scenarios on the psychological constructs analyzed. A p-value equal to or less than .05 was considered statistically significant.

3.3.1 Preliminary Analysis

A comparison between groups was made to control the order effect of the standard and customized VR scenarios (see Table 2). No statistical differences emerged, meaning the two VRE scenarios' administration order had no effect.

 Table 2. Comparison between randomization orders.

	Randomization		F	
	order group	M (SD)	(1,18)	р
CAN N TH	1	3.5 (1.1)	1.477	
SAM_V_T1	2	2.8 (1.5)	1.47	ns
SAM A TI	1	5.2 (2.4)	4 13	DE
SAM_A_II	2	7.2 (1.9)	4.15	115
SAM D T1	1	4.8 (2.2)	2.55	ns
D_11	2	6.2 (1.7)		
VAS_RELAX_T1	2	7.8 (1.0) 8 2 (1.1)	0.43	ns
	1	6.6 (2.4)		
VAS_PRESENCE_T1	2	7.9 (2.1)	1.67	ns
	1	2.4 (0.8)	0.44	
SAM_V_12	2	2.1 (1.3)	0.64	ns
SAM A T2	1	5.8 (2.7)	2.22	-
54.112	2	7.4 (2.1)	2.22	115
SAM D T2	1	5.6 (1.7)	1.18	ns
	2	6.4 (1.7)		
VAS_RELAX_T2	1	7.6 (0.9)	2.95	ns
	1	7.6 (0.9)		
VAS_PRESENCE_T2	2	8.4 (0.9)	3.43	ns
	1	28.2 (7.6)		ns
STAI-Y1_T1	2	27.1 (4.5)	0.16	
CTALV1 TO	1	25.2 (3.9)	0.04	
51AI-11_12	2	25.6 (4.7)	0.04	ns
REALISM T1	1	100.2 (14.3)	015	ns
	2	103.2 (19.8)	015	10
REALISM_T2	1	105.1 (17.5)	0.26	ns
	2	108.6 (12.9)		
ENGAGEMENT_T1	2	10.6 (2.2)	0.07	ns
	1	10.3 (2.9)		
ENGAGEMENT_T2	2	12.3 (1.4)	3.82	ns
	1	11.8 (5.5)	1.01	
IMMERSIVITY_11	2	13.9 (3.1)	1.81	ns
IMMERSIVITY T2	1	17.2 (1.5)	1 11	700
IMMERSIVITY_12	2	16.1 (2.4)	1.11	ns
USABILITY T1	1	17.1 (3.1)	3.86	ns
	2	14.1 (3.7)		
USABILITY_T2	2	15.1 (3.4)	0.82	ns
		467(75)		
EMOTIONAL STATE_T1	2	43.80 (8.4)	0.66	ns
	1	45.70 (7.1)		
EMOTIONAL STATE_T2	2	47.30 (6.1)	0.30	ns

Notes: T0 (assessment before VREs); st (assessment after standard VRE); p (assessment after customized VRE); M (SD)= Mean (Standard Deviation); DASS_Tot= Depression, Anxiety and Stress Scale-21_Total score; DASS_D= Depression, Anxiety and Stress Scale-21_Depression scale; DASS_A= Depression, Anxiety and Stress Scale-21_Anxiety scale; DASS_S= Depression, Anxiety and Stress Scale-21_Stress scale; STAI-Y2= State-Trait Anxiety Inventory-Y2 (regarding trait anxiety features); STAI-Y1= State-Trait Anxiety Inventory-Y1 (regarding state anxiety features); SAM_V= Self-Assessment Manikin_Valence; SAM_A= Self-Assessment Manikin_Arousal; SAM_D= Self-Assessment Manikin_Dominance/Control; VAS= Visual Analogue Scale.

The current sample was composed of 16 (80%) women. The mean age of the participants was 34.2 (SD = 10.6; min.: 19; max.: 58), and the mean of the school years was 18.9 (SD = 3.6; min.: 11; max.: 26). Twelve (60%) of the participants were single or non-cohabitant, and eight (40%) were married or cohabitant. Nine (45%) participants were pursuing a Ph.D. program, one was a university student, and ten (50%) were employed. No participant had a diagnosed psychological problem. Seven (45%) participants were familiar with relaxation techniques to manage anxiety and stress symptoms, and six (30%) had previously experienced VR for entertainment. Means and standard deviations of total and subscales self-report scales are described in Table 3.

Questionnaire	M (SD)
DASS_Tot	9.9 (6.9)
DASS_D	2.5 (2.5)
DASS_A	1.7 (1.9)
DASS_S	5.7 (3.6)
STAI_Y2	34.9 (6.1)
STAI_Y1 (T0)	32.7 (9.1)
STAI Y1 (st)	27.7 (6.1)
STAI Y1 (p)	25.4 (4.2)
SAM_V (T0)	3.3 (0.7)
SAM_V (st)	3.2 (1.3)
SAM_V (p)	2.2 (1.1)
SAM_A (T0)	5.5 (1.7)
SAM_A (st)	6.2 (2.4)
SAM_A (p)	6.6 (2.5)
SAM_D (T0)	5.5 (1.4)
SAM_D (st)	5.5 (2.1)
SAM_D (p)	6 (1.7)
VAS_Relax (T0)	5.9 (1.6)
VAS_Relax (st)	8 (1.3)
VAS_Relax (p)	8.2 (1.6)
VAS_Sense of Presence (st)	7.3 (2.3)
VAS_Sense of Presence (p)	8 (1.1)
Realism (st)	101.7 (16.9)
Realism (p)	106.9 (15.1)
Engagement (st)	10.8 (2.6)
Engagement (p)	11.3 (2.5)
Immersivity (st)	12.9 (4.5)
Immersivity (p)	16.6 (2.1)
Usability (st)	15.6 (3.7)
Usability (p)	15.8 (3.4)
Emotional_state (st)	45.3 (7.9)
Emotional state (p)	46.5 (6.4)

Table 3. Means and Standard Deviations related to self-report scales' scores.

<u>Notes</u>: T0 (assessment before VREs); st (assessment after standard VRE); p (assessment after customized VRE); M (SD)= Mean (Standard Deviation); DASS_Tot= Depression, Anxiety and Stress Scale-21_Total score; DASS_D= Depression, Anxiety and Stress Scale-21_Depression scale; DASS_A= Depression, Anxiety and Stress Scale-21_Anxiety scale; DASS_S= Depression, Anxiety and Stress Scale-21_Stress scale; STAI-Y2= State-Trait Anxiety Inventory-Y2 (regarding trait anxiety features); STAI-Y1= State-Trait Anxiety Inventory-Y1 (regarding state anxiety features); SAM_V= Self-Assessment Manikin_Valence; SAM_A= Self-Assessment Manikin_Arousal; SAM_D= Self-Assessment Manikin_Dominance/Control; VAS= Visual Analogue Scale.

3.3.3 Qualitative Results

In the next sub-paragraphs, information about the user acceptance of the hardware- and software- variables characterizing the VR experiences is provided.

- Is the VRE Preferable with or without the Body Scan Audio Track?

The participants' reports regarding the VR experiences and the body scan audio track showed that the body scan was considered helpful in promoting relaxation by 18 (90%) of the participants. Two individuals considered this track useful if administered alone; they reported having difficulty concentrating on the environment and the audio track content simultaneously and expressed interest in trying the experience with the audio track by imagining the relaxing environment.

- Usability, Physical Discomfort in Wearing the Head-Mounted Display, and Simulator Sickness

All of the users affirmed that Oculus was very easy to wear and use since the experimenter provided technical support during the session. Six (30%) individuals recommended improving the Oculus Quest 2 wearability to allow greater grip and more comfort. In the suggestion section, three of them stated that they would prefer to choose the position in which to stay during the activity (e.g., lying down), and also to manage the feeling of heaviness and encumbrance associated with the head-mounted display. Regarding the assessment of the cybersickness symptoms, two (10%) of the participants referred to mild headaches and burning eyes experienced towards the end of the VRE session with the Oculus exposure.

- Graphics Quality, Synchronization, and Wayfinding VRE-Related

Information on the graphic quality was obtained about the perception of the audio and visual elements. All participants reported that audio stimuli, such as animal sounds, wind rustle, and wave movement, felt good and were consistent with the scenario. As for the visual elements, all of the individuals appreciated the colors and shapes of the objects that made up the context, evaluating them as very or totally engaging and very similar to reality. All of the individuals felt able to actively explore the environment as they decided to move their heads and gaze into the virtual context. Moreover, the way they moved around the VR scenarios and the synchronization between their movement and what they saw seemed very (15 participants; 75%) or extremely (five participants; 25%) realistic to them. Moreover, users expressed a need to know their position and direction in the VREs.

- Realism of the Virtual Environments

Regarding the level of realism of the auditory and visual elements offered by the VREs, nine (45%) individuals considered the animal sounds very similar to reality (e.g., the sound of birds), 10 (50%) found the meteorological sounds extremely realistic (such as the wind and raindrops), and nine (45%) of the participants appreciated the way that the visual effects related to the meteorological conditions. Seven (35%) individuals evaluated the placement of the visual elements in the environment as very similar to a real context. Six (30%) of the participants affirmed that the colors, lights, and reflections at the transition from one moment to another of the day were very similar to real stimuli, and nine (45%) considered the movement of birds and clouds to be realistic. Moreover, most of the participants (16; 80%) did not appreciate the presence of humans in the VR context. They instead preferred to imagine the presence of someone close to them during the VR experience. Finally, five (25%) users specified that the perception of the countryside's synchronous movement of the leaves and flowers was artificial, suggesting a less uniform fluctuation.

- Preferred VR Environment between Standard and Personalized

Eighteen (90%) of the individuals preferred the customized environment. All participants expressed a subjective point of view based on their preferences. Overall, four main themes were identified as the most influential factors affecting the participants' choices in the VREs

for relaxation (Table 4): (1) the correspondence with the environment they would have chosen in a real context (relaxing as in reality); (2) the reminiscence; (3) the possibility to choose and control the elements of the VRE (control); (4) the realism of the stimuli (see Table 4).

Main theme	User quotes
	"Especially for the birdsong which is what I feel when I rest in the country" (Particinant 2)
	"I preferred the personalized environment because it reflects more of a context in which I relay in reality" (Participant 9)
	If find it easier to relay in the mountains" (Particinant 10)
	Because even in reality when I have to relay I choose to go to the seaside"
	(Participant 11).
	"The elements around me reflected an environment that generally makes me
	feel good, and relaxed" (Participant 8).
	"If I want to take a break from everything, I usually go to the mountains"
	(Participant 3).
	"I could select a context that generally relaxes me" (Participant 13).
	"I usually feel more comfortable at the seaside, that's why I chose it"
	(Participant 6).
"Relaxing as in reality"	"I prefer the seaside to the countryside in reality. I can associate it more with a sense of relaxation" (Participant 16).
	"There were elements that helped me to relax as in everyday life"
	(Participant 5).
	"This is very similar to my favorite place in reality" (Participant 7).
	"It is similar to the place where I feel better about relaxing" (Participant 17).
	"Because it represents a context that recalls the relaxing places in reality, and
	this was helpful" (Participant 18).
	"I could recreate a more similar environment to the one that really relaxes me"
	(Participant 14).
	"It was similar to the beach I choose to relax" (Participant 12).
	"The seaside is one of the environments that I choose when I want to relax and
	disconnect my head" (Participant 19).
	"The countryside scenario reminds me of the area from which I come. This
"Reminiscence"	makes me feel at home" (Participant 1).
	"The very fact of having been able to choose" (Participant 15).
	"I chose the mountain environment because I associated many beautiful
	memories with it" (Participant 9).
	"Also, because, as a child, I always went to the seaside, and it was beautiful.
	This context somehow reminded me of it" (Participant 19).
	"I decided what to insert in the environment. Then I liked it more!"
"Control"	(Participant 17).
	"First of all because I could decide" (Participant 19).
	"It was my choice" (Participant 16).
	"In the standard context I found more elements similar to reality, I felt more
"Realism of the stimuli"	comfortable" (Participant 20).
	"The seaside seemed more like reality and, although it was not the
	environment I had chosen, in the end I preferred it" (Participant 4).

Table 4. Themes and quotes by users.

Of these 18 users, 16 (89%) said it represented the real context in which they would choose to relax, and they found the possibility of customizing the elements very important, to make the VRE more alike not only the preferred context but also the place in the real world where they would go to relax. Three (17%) individuals reported that the most important thing related to customization was the perception of controlling and deciding the setting, which was felt to be very impactful on their motivation to practice VR relaxation experiences. Four (22%) of the participants reported that the chosen VRE and the way they could personalize it reminded them of the places they experienced in their childhood. Even though they might have found another virtual setting more realistic, they chose the one in which they would relax better as, in their opinion, this had a significant impact on the possibility of experiencing a subjectively adequate level of relaxation. Finally, two (10%) individuals preferred the standard VR environment saying that by finding it more realistic, they felt much more comfortable in that setting.

- Differences between Self-Report Administration after Standard and Customized VR Experiences about Immersivity, Sense of Presence, Realism, Engagement, Usability, Subjective Arousal Perception, Sense of Relaxation, and Pleasantness

Even if the order effect of exposure to the two VR scenarios was controlled, a within-subject design was limited by the possibility that the participants realized the aim of the research, affecting their responses. In this regard, the following analyses were conducted to inform our design, by considering the subjective experience that the participants had when they were exposed to the standard and customized VREs. Comparisons using Multivariate ANOVA, between the assessments conducted after the standard and customized VRE, showed a difference for the Immersivity scale ($F_{(1,19)}=11.33$; *p-value*<.01; $\eta 2=0.37$) with a higher sense of immersivity experienced after the customized VR condition compared to the standard one. No differences emerged for the Usability ($F_{(1,19)}=0.10$; p-value>.05), the VAS Sense of *Presence* ($F_{(1,19)}=2.75$; *p-value*>.05), the *Realism* ($F_{(1,19)}=3.54$; *p-value*>.05), the *Engagement* $(F_{(1,19)}=1.06; p-value>.05)$, and the *Emotional State scales* $(F_{(1,19)}=0.57; p-value>.05)$. Repeated Measures ANOVA, conducted to compare the three assessment phases, showed statistically significant differences between the three assessment time points (T0-T1-T2) for the STAI-Y1 ($F_{(2,18)}=12.21$; *p-value*<.001; $\eta 2=0.58$), the SAM-Valence ($F_{(2,18)}=8.49$; *pvalue*<.05; η 2=0.49), the *SAM-Arousal* (F(2.18)=4.21; *p-value*<.05; η 2=0.25), and the *VAS-relax* ($F_{(2.18)}$ =50.56; *p-value*<.001; η 2=0.85). No differences emerged for the SAM-Dominance $(F_{(2.18)}=1.47; p-value>.05)$. Paired-sample t-tests were conducted as a post-hoc analysis to

understand the origins of the differences between the three assessment time points. As shown from the STAI-Y1 and the VAS relax scale (Table 5), state-anxiety symptoms were higher before both the standard and the customized VR experiences. Indeed, individuals expressed, on average, a lower feeling of relaxation and more intense activation and anxiety levels before the exposure to both of the VR scenarios (t STAI (T0)/STAI (st)=3.20; *p-value*<.01; t STAI (T0)/STAI (p)=4.97; *p-value*<.001; t VAS_rel (T0)/VAS_rel (st)=-6.25; *p-value*<.001; t VAS_rel (T0)/VAS_rel (p)=-2.46; *p-value*<.001; t VAS_rel (T0)/VAS_rel (p)=-2.46; *p-value*<.05). Moreover, higher relaxation levels were experienced after the customized VR experience rather than the standard one (t STAI (st)/STAI (p)=2.20; *p-value*<.05; t VAS_rel (st)/VAS_rel (p)=-9.95; *p-value*<.001). Regarding the Valence of emotions, the participants referred to having experienced a greater degree of pleasantness after the customized VR setting (t SAM_V (T0)/SAM V (p)=3.80; *p-value*<.01; t SAM V (st)/SAM V (p)=2.97; *p-value*<.01).

	t (1,19)	p-value
STAI Y1 (T0) / STAI Y1 (st)	3.20	<.01
STAI Y1 (T0) / STAI Y1 (p)	4.97	<.001
STAI Y1 (st) / STAI Y1 (p)	2.20	<.05
SAM_V (T0)/ SAM_V (st)	0.42	ns
SAM_V (T0) / SAM_V (p)	3.80	<.01
SAM_V (st) / SAM_V (p)	2.97	<.01
SAM_A (T0) / SAM_A (st)	-1.58	ns
SAM_A (T0) / SAM_A (p)	-2.46	<.05
SAM_A (st) / SAM_A (p)	-1.29	ns
VAS_relax (T0) / VAS_relax (st)	-6.25	<.001
VAS_relax (T0) / VAS_relax (p)	-2.52	<.05
VAS_relax (st) / VAS_relax (p)	-9.95	<.001

Table 5. Paired Samples t-Test between psychological constructs administered at three different times.

<u>Notes</u>: T0 (assessment before VREs); st (assessment after standard VRE); p (assessment after customized VRE); M (SD)= Mean (Standard Deviation); STAI-Y1= State-Trait Anxiety Inventory-Y1 (regarding state anxiety features); SAM_V= Self-Assessment Manikin_Valence; SAM_A= Self-Assessment Manikin_Arousal; SAM_D= Self-Assessment Manikin_Dominance/Control; VAS= Visual Analogue Scale.

3.4 Discussion

The primary aim of the current research was to investigate usability, preferences, pleasure, satisfaction, engagement, immersivity, sense of presence, and the subjective perception of relaxation and state-anxiety of users whether being exposed to a standard or a personalized relaxing VREs, concurrently with a body scan audio track. The presence of an audio track integrated with the VR scenario was appreciated by most of the individuals who considered this complementary approach a useful strategy for focusing their attention on the activity and promoting relaxation. This feature deserves further investigation by further studies that should include other conditions in which users experience either the audio track or the VR scenario alone. Regarding the device's usability, wearing the head-mounted display interfered to a limited extent with exposure to the VRE. To inform our future studies, it will be essential to consider some of the participants' recommendations about the possibility of improving the Oculus Quest 2 fit by adopting a peculiar support (e.g., a faceplate head, a strap mount) for the head-mounted display that balances the distribution of the device's perceived weight, reducing interference. Useful self-reported indications were also obtained about the synchronization and wayfinding of the VRE and the auditory stimuli that contribute to rendering visual stimuli and the experience in general, more realistic. One interesting finding concerns the presence of virtual persons in VRE. While the depiction of people was not generally appreciated, most participants reported having a tendency to imagine at least another person in the virtual reality scenario. This underlines the potential contribution of imagination to the recreation of a realistic context of relaxation if the elements also represent human beings. A higher sense of immersivity was referred to by the users when they had the opportunity to customize the VRE. This means that customization has generated a greater feeling of being present in the VR environment [158, 256]. Participants experienced a higher level of comfort after the customized virtual reality experience. Interestingly, based on their self-reporting, the factors that contributed to the pleasure and preference for customized virtual scenarios were: (1) the correspondence of the customized VR with the relaxing context they would have chosen in reality; (2) the reminiscence; (3) the possibility of choosing and controlling elements in the VR context; (4) the realism of the stimuli. Regarding users' preferences for the customized or standard VR scenarios, no differences emerged in our sample between male and female participants, both from a preliminary quantitative analysis and a qualitative assessment. Other studies have shown how useful virtual reality can be to promote relaxation and reduce anxiety [1, 183, 257]. In line with these outcomes, our data suggest that exposure to a VR relaxing

environment promoted the perception of reduced state-anxiety symptoms and facilitated a sense of relaxation, independently of the possibility to customize the VR context per se. Furthermore, when users had the opportunity to customize the virtual scenarios to build their preferred relaxing environment, they felt a more subjective state of relaxation than when they experienced the standard VRE. These findings on customization hold promise as they highlight the importance of deploying a user-centric design approach [3, 5] in adapting the VRE stimuli to fit the needs of individuals and to promote a positive attitude and a better engagement with the relaxation training. Customization can also help make virtual contexts more enforceable than standard VRE. Future studies are expected to shed more light on this. The study of different types of relaxation training techniques (e.g., body scan, diaphragmatic breathing, PMR), embedded in different types of VREs (e.g., customized versus standard) should be further investigated in relation to the control group conditions. At times, even well-established and effective training relaxation programs are not appreciated by the participants. Indeed, in the current research, two individuals reported experiencing boredom and discomfort with the body scan technique, adding that the experience was engaging only thanks to the VRE. Otherwise, they did not appreciate the relaxation training, considering it too dull. Thus, integrating exposure to relaxation training with VRE could also help to better fit users' needs and support motivation to learn a valid relaxation technique that would not be experienced alone. Even if the current research and other studies have begun to investigate the combined role of VRE and relaxation training, so far, the results are still preliminary to prove that a specific relaxation technique, alone or combined with a VRE, is more efficient than others [1]. Therefore, randomized controlled studies should be conducted, comparing different experimental conditions with appropriately sized samples. Since not all of the relaxation training could be suitable for each person, it is empirical to consider individuals' preferences and characteristics by investigating customization and promoting the validation of more specific and appropriate treatment protocols based on the users' needs. Furthermore, the relaxation training described in this study involves an active role of the individual in promoting relaxation. Future studies may evaluate the role played by coping strategies and the Locus of Control [258] in influencing anxiety to understand if having an internal LoC could help to promote a better sense of relaxation by using the learning acquired during the procedure. As well, it would be interesting to determine if coping strategies change over time by comparing the pre-and post-exposure assessment. Another relevant aim for future investigations should be to evaluate in more depth the positive emotions experienced during the procedure and investigate the impact of the vividness of visual imagery on the sense of presence perceived

during VR exposure [259]. This study has some limitations that affect the generalizability of our findings. The sample size limits the studies reliability. The experimental conditions were not compared with a control group, limiting the findings' relevance for clinical practice. Nevertheless, the current study's findings are encouraging, but controlled and randomized trials should be conducted to investigate the efficacy of this type of digital relaxation intervention, including the deployment of a between-subject design approach to control intervention, including the deployment of possible "experimental demand" bias. It would be useful to stratify the participants' sample according to socio-demographic variables, and to increase its size. Future studies could also include validated standardized surveys of the local target population. Ultimately, together with the effective self-report questionnaires administered in this work, future studies may also foresee deploying more objective physiological assessment methods, such as skin conductance and heart rate variability measures to assess relaxation and state-anxiety.

3.5 Conclusions

The described study extends the state-of-the-art research on the impact of customized VREs, suggesting the role of customization as a promising component in facilitating engagement, and promoting a higher perceived state of relaxation. These preliminary results contribute to the planning of future studies investigating the effect of relaxation training in customized VREs on large-scale clinical and non-clinical samples. The efficacy and effectiveness of VRE for relaxation are likely to encourage the deployment of these digital interventions to promote mental well-being, making them more sustainable and scalable in the clinical practice. Overall, the personalization of clinical interventions by deploying a user-centered design approach helps to better consider people's needs to improve their motivation and engagement, with positive implications for their well-being. Our exploratory outcomes are promising for implementing between-subject experimental design studies that should also be focused on deploying VR settings to improve and manage limits related to mental relaxing imagery techniques. This field of investigation could shed more light on the role of VR environments in favoring relaxation and autonomy, reducing costs, and improving the deployment of treatments, particularly needed during pandemic conditions. Access to effective online relaxation training protocols can be beneficial for administering psychological interventions, especially in situations where there are restrictions to face-to-face treatment. In conclusion, the current research's findings provide a better understanding of the user experience and preferences for the customization of VREs for relaxation, paving the way for more extensive future analyses on the role of these solutions in supporting digital interventions for mental well-being.

CHAPTER 4.

CAN CUSTOMIZED, RELAXING VIRTUAL REALITY SCENARIOS ENHANCE EFFICACY IF ADMINISTERED WITH WEB-BASED PROGRESSIVE MUSCLE RELAXATION TRAINING? AN INNOVATIVE COMPLEMENTARY METHOD

The present chapter provides an overview of the second study of the Ph.D. program. The related research protocol is published in the *Journal of Medical Internet Research - Research Protocol* [260], and the results related to the principal aims of the project are under review by the *Journal of Medical Internet Research - Mental Health* (Pardini et al., under review). The current research is registered as a clinical trial in a publically accessible primary register that participates in the WHO International Clinical Trial Registry Platform (ClinicalTrials.gov; registration number: NCT05478941; link: https://clinicaltrials.gov/ct2/show/NCT05478941). The current contribution is about a longitudinal, between-subjects, 3-armed randomized controlled trial aiming to compare three treatment conditions to investigate the impact on state anxiety of Progressive Muscle Relaxation Technique (PMR), associated with a customized, relaxing scenario in Virtual Reality (VR), and the role of VR scenarios in facilitating the recall of relaxing images and the sense of presence in the virtual environment. A secondary objective is to understand if two different web-based relaxation intervention strategies (deployed by Zoom platform or based on a structured series of audio-track) can contribute to reducing anxiety and stress.

As stated in the previous Chapter 1 and 2, VR can facilitate exposure to stressful or relaxing stimuli and enables individuals with difficulties visualizing scenes to be involved in a more realistic sensorimotor experience. It also facilitates multisensory stimulation, the sense of presence, and the achievement of relaxation. VR scenarios representing visual and auditory elements of natural relaxing environments can facilitate the learning of relaxation techniques such as PMR. As aforementioned in Chapter 1, a complementary standardized technique deployed to reduce anxiety symptoms is the integration of PMR and Guided Imagery (GI). Indeed, exposure to a pleasant imaginary environment can help establish an association between a relaxing scenario and the relaxation technique, consequently promoting relaxation. Empirical evidence has shown that VR scenarios can positively impact the effects of relaxation techniques by enabling people to experience emotional conditions in more vivid settings (e.g., [183, 195, 198]). The findings from the literature, including the current study, will help shape the experimental design to be applied to future randomized controlled trials, also considering
clinical samples. Results outline how VR could be considered a more engaging and helpful technique in promoting relaxation and decreasing anxiety levels than GI by making visualization more accessible and helping people face more realistic sensory experiences. Assessing the efficacy of the PMR in alternative delivery modes may extend its applications, especially in situations where the standard procedure is more challenging to be administered. To our knowledge, an equivalent study has yet to be published on this matter.

STUDY 2. CUSTOMIZED, NATURALISTIC VIRTUAL REALITY SCENARIOS COUPLED WITH WEB-BASED PROGRESSIVE MUSCLE RELAXATION TRAINING FOR THE GENERAL POPULATION: A PILOT STUDY

4.1 Theoretical background

4.1.1 Virtual Customized Scenarios as an alternative to Guided Imagery

In working adults and university students, the prevalence of anxiety and stress is increased, predisposing to physical diseases and general repercussions on well-being [261, 262].

The situation has worsened with the SARS-Cov-2 outbreak, which has impacted individuals' general well-being worldwide [263, 264]. Different types of standardized relaxation interventions (e.g., Mindfulness-based Stress Reduction, Jacobson's Progressive Relaxation Techniques, Schultz's Autogenic Training, Abdominal Relaxations, and Visualizations) before, during, and after SARS-Cov-2 have been shown to reduce anxious and stressful symptomatology in university students [264, 265] with a more significant effect when integrated with other complementary techniques such as GI [17, 19]. GI is useful in creating mental imagery and refocusing attention on pleasant and relaxing imagined visual, auditory, tactile, or olfactory sensations, resulting in specific psychological and physiological responses, such as relaxation reduction of the autonomic nervous system responses [17, 266, 267].

As more broadly stated in Chapter 1, the PMR is considered one of the most effective and adaptable techniques for relaxation, anxiety management, and stress reduction considering different kinds of target populations (e.g., non-clinical individuals, patients with cancer, chronic pain, osteoarthritis, heart disease, chronic headache, anxiety, and depression) [1, 6, 22, 233, 268]. Complementary methods that imply the administration of integrated cognitive and behavioral procedures are studied to manage anxiety, stress, depression, and pain, particularly in populations such as patients with cancer. Notably, the integration of PMR and GI enhances

and facilitates the effect of relaxation, reducing anxiety and depression symptoms [1, 3]. The integration of PMR and GI can promote a higher sense of relaxation during training sessions, allowing the application of complementary techniques to cope with stress experienced in daily activities [17, 269-272]. As stated before, the GI technique permits to imagine an undefined number of scenarios that are subjectively perceived as relaxing and safe.

Starting from this background, we hypothesized that the possibility to customize a relaxing scenario, according to individual preferences and in a more immersive context through VR technology, could more effectively promote relaxation and anxiety reduction than using GI only. Various studies demonstrated that being immersed in virtual natural environments facilitates the lowering of anxiety and stress symptoms in college students [17], also in association with different types of relaxation training (e.g., body scan, muscle relaxation training) [18-21].

VR is a usable, engaging, and user-friendly technology that promotes a full multisensory immersion in the virtual context, facilitating the control of disturbing external stimuli [15,16, 4,5]. As stated in Chapter 2, VR is widely used to treat various mental diseases [1], also in association with different types of standardized relaxation training, and it facilitates the sense of presence, immersion, and relaxation [2,3]. Moreover, as previously anticipated in Chapters 2 and 3, the customization of stimuli characterizing VR scenarios could further promote a more realistic and engaging experience, enhancing relaxation, sense of presence, and perception of security in the virtual context (e.g., 3, 233, 268), offering people a more realistic emotional experience, reflecting their needs and preferences [e.g., 4]. Considering the positive impact of GI associated with PMR on allowing relaxation, it may be hypothesized that exposure to a more vivid, closer-to-reality virtual experience may turn out to be helpful in further improving the relaxation learning promoted by the PMR and in decreasing anxiety symptomatology.

4.1.2 Web-based Psychotherapy Interventions to promote relaxation

The first part of the standard PMR training, named "*active PMR*", implies a series of active sessions in which people learn how to tense and release the different muscle groups from the bottom to the top of the body to recognize the subjective state of muscle relaxation; the relaxation of the muscle areas of the body reducing tension interfering with the skeletal muscle activity. These previous active PMR phases are essential and are typically administered face-to-face by a psychotherapist or a healthcare operator trained to conduct the PMR (for a visual representation, see Figure 1).

Figure 1. Representation of what usually happens in the psychotherapist room during the PMR training. The first two images are schematic representations of some of the processes that occur at the muscular level during muscular contraction and relaxation. Specifically, the actin and myosin movement is represented in relaxed versus contracted muscle. Less contact between actin and myosin produces less force (Sources: Lodish, H. (2013). Molecular Cell Biology. UK: W. H. Freeman; https://tommorrison.uk/blog/anatomical-movements).



By considering the effective results of web-based interventions in reducing stress among college students [273] and the relevant advantages that online therapy can offer (e.g., saving costs related to attending psychotherapy sessions), we assumed that assessing the efficacy of PMR in alternative settings may facilitate the administration of treatment when the implementation of the standard procedure is not possible (Figure 2).

Figure 2. Representation of web-based interventions in which the therapist deploys the web-based session via conference call.



The usefulness of web-based relaxation interventions is supported by studies showing how an online procedure led to significant results equal to in-person interventions [274, 275]. One of the first studies investigating the effect of comparing different types of relaxation techniques web-based deployed during the COVID-19 is by Pizzoli et al. [276], in which a series of audio clips were remotely delivered by following three different intervention approaches: 1) auditive natural stimuli (e.g., water sounds), 2) breathing regulation, 3) body scan. Results showed that all three techniques had positive effects on perceived relaxation and stress, concluding that the audio clips were effective in inducing a calmer psychological state, providing novel insights, and suggesting the deployment of further studies to investigate the effect of delivering more than one session treatments. In particular, studies should assess differences in the efficacy of longitudinal exposure to relaxation sessions and correct for the effect of training or habituation low-cost web-based interventions to reduce preoccupation and anxiety in the general population and people with anxiety symptomatology [276].

In investigating the impact of web-based psychotherapy interventions on psychopathological outcomes, there are controversial results regarding a PMR procedure administered in vivo or via audio-tracks [8, 9]. To our knowledge, no recent study addresses the comparison among training sessions administered in vivo, in a remote modality, and via audio-tracks.

Starting from the state of the art knowledge and the impact also due to SARS-CoV-2 in the proliferation and permanence of online interventions, with the help of technological tools that promote autonomy in the psychotherapy administration, we intend to explore PMR's efficacy in a remote situation delivered via both the Zoom platform and audio-tracks deployed by the Moodle platform. Indeed, online treatments promote users' independence in managing psychotherapy techniques, leaving time for other face-to-face psychotherapy activities that require the active involvement of both the patient and the therapist. Assessing the efficacy of PMR in alternative settings may facilitate the administration of treatment where it is impossible to implement the standard procedure. It may also increase the positive impact of abbreviated relaxation sessions and the deployment of a combination of relaxation methods with patients who have chronic pain or must be exposed to invasive medical treatments [18, 67].

In light of the evidence described, we aimed to deploy the active PMR sessions remotely via Zoom or Audio-track, accessible through the Moodle platform, and the last complementary session in the therapist's presence, exposing people to a passive progressive relaxation session with GI or VR.

4.2 Aims and hypotheses

The purposes of the present research project are as follows:

1. Investigating if a customized, relaxing scenario in VR can facilitate relaxation and anxiety operationalized as based on lower state anxiety and heart rate frequency comparing the three experimental conditions.

Hp1: Even if a general reduction in state-anxiety in all three groups is expected, it has been hypothesized that VR is more effective than in-imagination exposure in allowing relaxation and decreasing state anxiety and heart rate (e.g., 199, 277, 278, 279).

2. Single-session web-based relaxation intervention outlined positive effects on perceived relaxation and stress (e.g., 276), suggesting the future deployment of studies aiming to investigate the effect of more than one session treatments to assess differences in the efficacy of longitudinal exposure to relaxation sessions in participants belonging to clinical and general populations [276]. For this reason, our second aim is to investigate the effect of the entire relaxation protocol on state anxiety for all three groups. Specifically, we wanted to see whether one of the two web-based training interventions impacted more on the perceived state of anxiety.

Hp2: It has been hypothesized that a general reduction in state-anxiety in all three groups during the relaxation training can occur [276]. Moreover, we wanted to investigate if groups differed in state anxiety based on the type of relaxation training received.

3. To understand if VR promotes a better sense of presence and engagement in the scenario than GI after Session 6 (T1) and if it helps to recall the image and be immersed in the relaxing scenario in Session 7 (T2).

Hp3: Since a high sense of presence during VR exposure seems to contribute to increasing skill learning (e.g., [280]), it has been assumed that a greater perception of presence and engagement in the VR scenarios and facilitation in recalling the image at T2 can be observed in individuals previously exposed to the VR condition.

4. Investigating if the sense of presence, immersivity in the scenario, and the perception that the virtual environment is realistic induced a more significant reduction in state anxiety after the virtual reality experience.

Hp4: Even if based on different aims, a positive relation between the sense of presence and anxiety has been revealed in studies that explored the efficacy of virtual reality exposure therapy (e.g., [281]). We expected to observe a positive effect of the sense of presence, immersivity, and perception of realistic components in the VR environment on state anxiety after the VR experience.

5. Investigate the differences between and within groups regarding trait anxiety, stress, depression, and coping strategies at T0 and T2.

4.3 Methodology

4.3.1 Ethics Approval

This study protocol involving human participants was General Data Protection Regulation compliant and developed following the Declaration of Helsinki (Italian law 196/2003, European Union General Data Protection Regulation 679/2016). The institutional review board of the Interdepartmental Ethical Committee of Psychology (17 Area) of the University of

Padova (Italy) approved the study protocol on May 28, 2021 (approval number 4213). In February 2022, the trial was tested based on a convenience sample of five volunteers from the general population. Then, it was decided to introduce the following questionnaires: the *Coping Orientation to the Problems Experienced-New Italian Version (COPE-NVI)* [282], the *ITC-Sense of Presence Inventory (ITC-SOPI)*, [169, 199], the *Vividness of Visual Imagery Questionnaire (VVIQ)* [283, 284], and the *Test of Visual Imagery Control (TVIC)* [283, 284]. The amendments and supplementations have been accepted, and the protocol received the final approval of the institutional review board of the Interdepartmental Ethical Committee of Psychology (17 Area) of the University of Padova, Italy (approval number 4701; April 29, 2022). Before their study enrolment, the participants signed a written informed consent form based on a paper-and-pencil form, agreeing to participate in all the study sessions.

They were informed that (1) their data would be confidential, (2) they could omit any information they did not wish to provide, and (3) they could withdraw from the study without providing any explanation.

4.3.2 Eligibility, Recruitment of Study Participants, and Randomization

The recruitment phase started in May 2022. Study participants were recruited in Northeast Italy via social networking websites (e.g., Facebook groups) and by providing information on the research during university lectures. Those interested in participating were asked to make an appointment with the investigators to participate in the first assessment phase (T0), in which the inclusion and exclusion criteria were evaluated, written informed consent was obtained, and a baseline evaluation was made. Eligible participants were adults from the general population (18 years or older), native Italian speakers, owning a PC, and able to use a PC and smartphone. Participants were excluded from the study if they had been diagnosed with a severe mental disorder or medical conditions that could hinder their participation in the study (e.g., neuromuscular disorders, severe psychiatric or neurological disorders, and assumption of drugs that could interfere with heart rate assessment and the subjective relaxation state), or if a psychotherapeutic treatment was ongoing. Eligible participants were randomly allocated to one of three experimental conditions based on a simple blinded randomization via an Excel (Microsoft Inc.) file. Two experimenters, trained in cognitive and behavioral psychotherapy, conducted the relaxation sessions, each administering to 50% of the sample of each group to control for possible biases related to the therapist's personality and competence. The two therapists who administered the experimental procedure received the same PMR training and were supervised by a senior researcher and psychotherapist supervisors.

4.3.3 Study Design

The present study is a longitudinal, between-subjects, 3-armed randomized controlled trial aiming to compare three treatment conditions:

1. Zoom & Guided Imagery (Active Comparator condition), consisting of deploying the PMR training via Zoom and GI exposure. It consists of the following steps:

- a standard behavioral intervention based on four individual PMR sessions via Zoom (s1-s4);
- an in vivo PMR relaxing session and GI conducted by a psychotherapist (T1) after a week from the baseline assessment (T0);

- a follow-up phase (T2) after two weeks, consisting in recovering the GI relaxing scenario and PMR session.

2. Zoom & Virtual Reality (Intervention A) is an experimental condition in which the complementary intervention comprises the PMR administered via Zoom integrated with a customized VR exposure deployed by a head-mounted display (Oculus Quest 2). It is composed of:

- four individual PMR sessions via Zoom (s1-s4);

- an in vivo PMR relaxing session integrated with a customized exposure in a virtual scenario (T1) after a week from T0;

- a follow-up phase with the same activities as the first condition.

3. *Audio-track & Virtual Reality* (Intervention B) is an experimental condition in which the complementary intervention comprises PMR administered through an audio-track integrated with the same customized VR exposure setup in the Zoom & Virtual Reality condition. It consisted of the following components:

- four individual PMR sessions via an audio-track delivered on the Moodle platform (t1-t4) (see Figure 3);

- an in vivo PMR relaxing session with a customized VR exposure (T1) after a week from T0;
- a follow-up phase characterized by the same activities as the other two conditions.

Figure 3. Example of one of the pages in the Moodle platform used by all three groups to fill out self-reported questionnaires and used by participants in the "Audio & VR" group to complete the PMR training sessions (t1-t4).



The assessment phases took place at baseline (T0; Session 1), before and after the four PMR sessions (administered on Zoom or audio-track), at the fifth in-presence session (T1; Session 6), and a week later, the exposure in one of the two complementary relaxation strategies (T2; Session 7).

- ★ The T0 (Baseline; Session 1) assessment phase has a duration of approximately 37 minutes, is the same for all participants, and is administered at the Virtual Reality Laboratory (A10-A11), Department of General Psychology, University of Padova (Italy). In this phase, the following measures are administered:
- a *demographic schedule* (addressing age, gender, nationality, mother tongue, marital status, years of school attendance, employment status, psychological problems or disorders, ongoing psychological treatment, drug use, medical conditions, neuromuscular issues, previous injuries, previous experiences in relaxation practice or anxiety management training, or with Virtual Reality deployed with a head-mounted display), and also, refraining from smoking, intense physical exercise, consuming caffeine for at least 1 hour before testing, and consuming alcohol for at least 6 hours before testing [64];

- a series of *self-report questionnaires* investigating depression, anxiety, stress, quality of life, and distress coping strategies (STAI-Y, DASS-21, PGWBI, and COPE-NVI);
- resting heart rate detection with a Mi Band 2 sensor.
- ★ At each of the *four training sessions (t1-t4; Sessions 2-5)*, to check that there were no potential adverse effects related to contracting and releasing the muscles before and after each relaxation session (t1-t4), the personal level of tension is assessed by using a *Visual Analog Scale (VAS)* from 0 (no tension) to 10 (extreme tension level). The state-anxiety level is evaluated based on the STAI-Y1. The four relaxation sessions are administered 2 to 3 days apart from each other for all three groups. The assessment phase is administered through Moodle, an e-learning platform used for data collection.
- ★ The *T1 phase (at Session 6)* required approximately 60 minutes. Before and after the relaxation session, states of tension and anxiety are assessed using a 0 (no tension) to 10 (extreme tension level) scale, and the state-anxiety level is evaluated based on the STAI-Y1. Then, participants are exposed to a PMR session merged with a VR or a GI procedure. Since the ability to generate vivid visual images is positively associated with the capacity to feel present in a virtual world [259], to control this construct, before the GI or the VR experience, all the participants fill out the *VVIQ* and the *TVIC*. Moreover, at the end of the T1 phase, the VR group filled out the *VRSQ* to monitor VR-related side effects (e.g., sickness), and the *ITC-SOPI* to assess the Sense of Presence, Engagement, Ecological Validity, and Negative Effects experienced to the exposure in the VR or the GI conditions. The Mi Band 2 sensor is used during the entire T1 phase is administered through the Moodle e-learning platform.
- ★ The *T2 phase (at Session 7)* is deployed for approximately 45 minutes. The stateanxiety level is evaluated based on the STAI-Y1. All the users are exposed to a self-GI experience in which those who were part of the VR group are asked to recall the customized VR scenario experienced during the T1 phase. Instead, the GI group retrieved the image participants had used in association with the PMR during the T1 phase. Moreover, at the end of the T2 phase, the VR group only filled out the STAI-Y2, the DASS-21, the PGWBI, the COPE-NVI, and the ITC-SOPI. This assessment phase is administered based on the Moodle e-learning platform. The Mi Band 2 sensor is used during the entire T2 phase administration to detect resting heart rate activity.

4.3.4 The Integrated PMR and Virtual Environment Design

The virtual environment design and the hardware and software equipment are described in Chapter 3, in the context of the first study in which the aim was to investigate the user experience, preferences, and engagement of the exposure to customized, natural, and realistic VR scenarios. Specifically, we adapted the procedure that Baird and Sands [17] introduced to propose an integrated intervention using different new technologies.

The protocol concerns the learning of the adapted abbreviated PMR (inspired by [55]) in five sessions delivered two times a week, as follows:

• Four sessions of active PMR, in which participants are required to perceptively learn the differences between the tension and relaxation of different muscle sections of the body (see Figure 4). Each session has a duration of approximately 25 minutes. The first session is dedicated to the hands, forearms, arms, neck, shoulders, and back; the second to the facial muscles; the third to the diaphragmatic breathing; and the fourth to the abdomen, buttocks, and lower limbs. These sessions are conducted via the Zoom platform and are conducted by a cognitive-behavioral psychotherapist. Each Zoom session consisted of measuring the relaxation state based on measures filled out using the Moodle e-learning platform, sharing general standardized instructions for conducting relaxation and the rationale of the PMR; guided relaxation aimed at relaxation of a dedicated body site, and filling out the same measures the Moodle platform. All sessions were self-managed within the Moodle platform for the participants in condition "Audio-track & VR" who participated in the four PMR sessions through the audio-track. Various activities have been scheduled based on the beginning of the training and the end of the activities planned in previous sessions.

Figure 4. Illustrative examples of the relaxation exercises during the four web-based active PMRs sessions.









3 - Diaphragmatic breathing



4 - Abdomen, buttocks, and lower limbs



Images derived from: FCS Student Life Blog

• The fifth VR session is administered at the University of Padova Laboratory. All participants are asked to wear a smartwatch for heart rate detection. Five measurements (one per minute) are made before the VR experience, 12 during the entire exposure in the virtual environment, and five after the experience in the virtual context. Approximately 12 minutes have been established for the VR experience to avoid potential cybersickness symptoms during the virtual experience. This brief relaxation session is practiced by inexperienced individuals allowing for a decrease in anxiety and negative mood [285, 286].

For an overview of the enrolment, interventions, and assessment phases, see Figures 5-9.

		STUDY PERIOD							
	Enrolment	Allocation		Post-allocation					Close-out
TIMEPOINT	-t,	то	t,	t2	t3	t₄	T1	T2	1 years ± 3 months
ENROLMENT:									
Eligibility screen	х								
Informed consent	х								
[List other procedures]	х								
Allocation		х							
INTERVENTIONS:									
[Active Comparator]			•						
[Intervention A]			•						
[Intervention B]			•						
ASSESSMENTS:			•						
[List baseline variables]	х	х	•						
[List outcome variables]			•						Х

Figure 5. SPIRIT diagram. SPIRIT, Standard Protocol Items: Recommendations for Interventional Trials. Overview of the enrolment, interventions, and assessment phases.

Figure 6. Consolidated Standards of Reporting Trials 2010 flowchart. GI: guided imagery; PMR: progressive muscle relaxation technique; VR: virtual reality.



CONSORT 2010 Flow Diagram



Figure 7. Graphical description of the activities carried out during the Active Comparator condition: PMR training via Zoom and GI exposure.



Figure 8. Graphical description of the activities carried out during Intervention A: PMR training administered via Zoom integrated with a customized virtual reality exposure deployed by wearing a head-mounted display (Oculus Quest 2). GI: Guided Imagery; PMR: Progressive Muscle Relaxation Technique.



Figure 9. Graphical description of the activities carried out during Intervention B: PMR training administered through the Moodle platform via an audio-track integrated with a customized virtual reality exposure deployed by wearing a head-mounted display (Oculus Quest 2). GI: Guided Imagery; PMR: Progressive Muscle Relaxation Technique.



4.3.5 Power Calculation

Sample size estimation was computed with G*Power 3.1 software (Heinrich Heine Universität Düsseldorf) [287]. As a statistical test, ANOVA repeated-measures between-within factors and interaction have been considered. The effect size was set as 0.25, the α as .05, and the Power (1 β error probability) as 0.80. We estimated having to recruit at least 36 participants for each of the three groups. Moreover, sample size was confirmed using the *General Linear Mixed Model Power and Sample Size* program [288].

4.3.6 Data Collection, Storage, and Security

Based on informed consent, participants were informed about the experimental procedure; how data were collected, transmitted, and stored; and who had access to the data. Any digital data, such as responses to web-based questionnaires, were uploaded and saved to a secure server. Any hard-copy documents were stored in a locked cabinet. Both digital data and hard copy documents are retained for ten years after the end of the study and then securely destroyed.

4.3.7 Outcomes measures

Data collection. Measures were administered before and after the relaxation sessions. The heart rate frequency was recorded based on the XIAOMI MiBand2 before the sessions at T1 (Session 6) and T2 (Session 7), to have Baseline data, other than during and after relaxation sessions.

-A *Socio-demographic schedule* was filled in to obtain information about gender, age, mother tongue, marital status, years of education, occupation, psychological, medical, and neuromuscular problems, and use of drugs if participants had relaxation training experience or used virtual reality devices in the past.

- The *State-Trait Anxiety Inventory* - *Y* (*STAI-Y*) [245, 246] is a self-report questionnaire that allows investigating the state and trait anxiety with 40 items on a four-point Likert scale. Items are grouped into two scales focused on how subjects generally feel (trait anxiety) or what they experience at particular times (state anxiety). The reliability and validity properties of the STAI-Y are good in the Italian sample. The internal consistency outcomes are in Table 1.

- The *Depressive Anxiety Stress Scale-21 (DASS-21)* [247, 248] is a self-report questionnaire based on 21 items that provide information about anxiety, depression, and stress symptomatology on a four-point Likert scale from 0 to 3. Internal consistency and convergent, divergent, and criterion-oriented validity are adequate in the original and Italian versions. The internal consistency outcomes are in Table 1.

- The *Coping Orientation to Problems Experienced-Nuova Versione Italiana* (COPE-NVI) [282] is a 60 items self-report questionnaire on a five-point Likert scale that investigates how often people use certain coping strategies with stressful or difficult events. Items are grouped into five subscales referring to different coping strategies: Social Support, Avoidance Strategies, Positive Attitude, Problem-Solving, and Turning to Religion. The present tool is psychometrically valid to measure coping styles in the Italian context. The internal consistency outcomes are in Table 1.

-The *Vividness of Visual Imagery Questionnaire (VVIQ)* [283, 284] is composed of 16 items on a five-point Likert scale and investigates individual differences regarding the ability to imagine visual contexts vividly. The participant is asked to generate four mental images and evaluate their vividness. The internal consistency outcomes are in Table 1.

-The *Test of Visual Imagery Control (TVIC)* [283, 284] is a measure that evaluates individual differences in the ability to intentionally control and modify mental images. For example, participants are asked to mentally visualize a car and then transform the image according to 10 different descriptions. Responses are recorded. The internal consistency outcomes are in Table 1.

-The *ITC-Sense of Presence Inventory (ITC-SOPI)* [169, 199] is a questionnaire of 42 items on a five-point Likert scale that allows for investigating the sense of presence experienced in a virtual reality contest. It comprises four subscales investigating the sense of physical space, the level of engagement experience in the virtual context, the Ecological Validity, and the negative effects of exposure. The internal consistency outcomes are in Table 1.

-The *Psychological General Well-Being Index (PGWBI)* [289, 291] is a self-report questionnaire of 22 items that provides a general subjective assessment of psychological wellbeing. It comprises six sub-scales: Anxiety, Depression, Positivity and Well-being, Selfcontrol, General Health, and Vitality. The scores for all sub-scales can be summarized to provide a summary score, which reaches a maximum of 110 points, representing the best achievable "well-being". The tool's psychometric properties are good for the original version and Italian validation. Considering our sample, an acceptable internal consistency emerged for the total The internal consistency outcomes are in Table 1.

-The *Virtual Reality Symptom Questionnaire (VRSQ)* [249, 292] assesses the general and eyerelated physical symptoms of exposure to a virtual reality environment. The score assigned to each item ranges from 0 to 6, with a maximum total score of 84 (48 for general symptoms and 36 for eye symptoms). Higher scores represent worse symptoms, with 0 corresponding to no adverse effects and 84 to serious adverse effects. Table 1. Demographic features and comparisons considering psychological constructs.

MEASURE		CRONBACH'S	95% CONFIDENCE INTERVAL			
		ALPHA	Lower Bound	Higher Bound		
	STAI-Y2	0.83	0.76	0.88		
	DASS-21- Depression	0.89	0.83	0.90		
	DASS-21- Anxiety	0.65	0.62	0.68		
	DASS-21- Stress	0.84	0.79	0.88		
10	PGWBI-Total	0.73	0.68	0.76		
	COPE-NVI-Social Support	0.86	0.81	0.90		
	COPE-NVI-Avoidance	0.75	0.67	0.84		
	COPE-NVI-Positive Attitude	0.79	0.71	0.84		
	COPE-NVI-Problem Solving	0.80	0.74	0.84		
	COPE-NVI-Trascendental Orientation	0.75	0.70	0.81		
	STAI-Y2	0.83	0.77	0.87		
	STAI-Y1 (before)	0.83	0.75	0.87		
	STAI-Y1 (after)	0.80	0.72	0.86		
	DASS-21- Depression	0.80	0.73	0.88		
T 1	DASS-21- Anxiety	0.62	0.60	0.64		
11	DASS-21- Stress	0.72	0.64	0.80		
	VVIQ	0.90	0.88	0.93		
	TVIC	0.86	0.82	0.88		
	ITC-SOPI-Presence	0.79	0.70	0.86		
	ITC-SOPI-Engagement	0.78	0.70	0.84		
	ITC-SOPI-Ecol validity	0.72	0.66	0.81		
	ITC-SOPI-Negative effects	0.73	0.64	0.83		
	STAI-Y2	0.88	0.81	0.91		
	DASS-21- Depression	0.79	0.76	0.83		
	DASS-21- Anxiety	0.64	0.62	0.66		
7.2	DASS-21- Stress	0.70	0.63	0.77		
12	PGWBI-TOTAL	0.76	0.68	0.85		
	COPE-NVI-Social Support	0.87	0.83	0.90		
	COPE-NVI-Avoidance	0.79	0.71	0.84		
	COPE-NVI-Positive Attitude	0.77	0.71	0.85		
	COPE-NVI-Problem Solving	0.80	0.75	0.85		
	COPE-NVI-Trascendental Orientation	0.74	0.65	0.81		
	ITC-SOPI-Presence	0.93	0.90	0.95		
	ITC-SOPI-Engagement	0.92	0.86	0.95		
	ITC-SOPI-Ecol validity	0.89	0.84	0.93		
	ITC-SOPI-Negative effects	0.81	0.74	0.85		

4.4 Results

4.4.1 Statistical Procedure and Data Analysis

Statistical analyses were performed using SPSS (version 29.0; IBM Corp) [293], RStudio 1.4.1717 [294], and the following packages: *tidyverse* [295], *dplyr* [296], *lsmeans* [297], *lmerTest* [298], *LMERConvenienceFunctions* [299] and *ggplot2* [300]. Linear mixed-effects (LME) models were then applied to the data, with Group, Block (pre/post), and Session as predictors and a random intercept for each subject to account for repeated measures. LME models represent an advantageous statistical approach that is more flexible than classic

ANOVA [301, 302], and since they allow the possibility to include both fixed effects and random effects, which represent variables that are not explicitly manipulated by the experimenter, but that contribute to the characteristics of the observed data, such as individual variability in within-subject designs [303]. On the LME models computed in the current study, we additionally performed an *F-test* using the *Satterthwaite approximation for degrees of freedom* [304] to assess the significance of each predictor. Multiple comparisons were corrected with *False Discovery Rate* (FDR). The total score was compared between the groups with a *Welch's t-test* [305]. Pairwise Comparisons with Bonferroni's confidence interval adjustment were calculated to compare differences between and within groups based on SPSS [293]. To investigate the values of Cohen's d and the effect-size correlation r, using the means and standard deviations of groups, the Effect size calculators by the University of Colorado-Colorado Springs (lbecker.uccs.edu) have been used.

To investigate the normal data distributions of the dependent variables, ranges of skewness and kurtosis are determined (± 2) [253, 306, 307]. The Kolmogorov-Smirnov and the Shapiro-Wilk tests are performed to evaluate the distributions' statistical significance (*p-value*>.05) and normality. The normality distribution has been investigated for all the dependent variables considered in the study assessed at different times and for the three groups individually, and the collected data follows a normal distribution. Frequencies, means, and standard deviations were measured to explore sociodemographic features. In addition, to assess the internal consistency of all the measures administered at T0, T1, and T2, Cronbach's alpha reliability measure and Confidence intervals (95% CI for reliability) have been executed. Multiple linear regression analyses were performed to investigate if the sense of presence, engagement, and realistic perception of the scenario can predict the state anxiety level experienced after the virtual reality exposure.

4.4.2 Socio-demographic features and comparison

The socio-demographic features of the study sample are outlined in Table 2. Since sociodemographic features (e.g., gender) can play a role in the way VR is experienced [e.g., 308-310], a multivariate ANOVA was conducted to investigate whether groups differed in sociodemographic and psychological characteristics before the relaxation training. A difference between groups was found for age, school years, marital status, and for the "Social Support" and "Transcendental Orientation" subscales of the COPE-NVI (see Table 2). To control for the possible impact baseline differences could have on the research outcomes, the effects of these variables were controlled for in the subsequent analyses. Fifteen individuals (41.7%) in the "Zoom & VR group", twelve participants (33.3%) in the "Zoom & GI group", and thirteen (36.1%) in the "Audio-track & VR group" referred to have an anxiety problem. Seven (19.4%), five (13.89%), and 4 (11.1%) participants respectively in the "Zoom & VR", "Zoom & GI", and "Audio-track & VR" groups referred to feeling anxiety related more to specific activities (mainly about university exams). Based on what participants referred to in the T0 assessment, psychological problems were related to the following themes: 1) problematic relationships with parents, 2) low self-esteem, 3) worries about university tests, 4) worries about economic problems, 5) anxiety symptoms experienced in stressful situations, and 6) relational problems. Moreover, the VRSQ was administered to individuals in the two VR groups to assess the possible collateral effects of VR exposure-related nausea. On average, individuals showed light susceptibility to general and eye-related motion sickness levels (see Table 2).

Table 2. Demographic features and comparisons considering psychological constructs.

	Zoom & VR N=36	Zoom & GI N=36	Audio-track & VR N=36	F(2,107)	χ2	Partial ŋ2
Gender (Women)						
(%)	28 (77.8%)	27 (75.0%)	28 (77.8%)		0.10	0.00
Age						
M (SD)	23.83 (6.10)	30.42 (8.36)	29.86 (8.46)	8.05***		0.13
School years						
M (SD)	16.81 (1.31)	18.25 (2.60)	17.72 (2.12)	4.45*		0.08
Marital Status						
single (%)	18 (50.0%)	16 (44.4%)	17 (47.2%)		11.70*	0.07
fiancé non-cohabiting (%)	17 (47.2%)	10 (27.8%)	16 (44.4%)			
married/cohabiting (%)	1 (2.8%)	10 (27.8%)	3 (8.3%)			
Medication						
Yes (%)	8 (22.2%)	5 (13.9%)	10 (27.8%)		2.10	0.06
Psychological Problems						
Yes (%)	15 (41.7%)	12 (33.3%)	13 (36.1%)		0.56	0.05
Medical Problems						
Yes (%)	6 (16.7%)	10 (27.8%)	10 (27.8%)		1.62	0.11
Relaxation training in the past						
Yes (%)	8 (22.2%)	3 (8.3%)	9 (25%)		3.81	0.03
VR in the past						
Yes (%)	10 (27.8%)	8 (22.2%)	13 (36.1%)		1.72	0.08
STAI-Y2-Total	10.05 (2.50)	17 54 9 50	15.00 (5.50)			0.05
M (SD)	48.06 (3.78)	47.56 (3.56)	45.33 (7.50)	2.73		0.05
DASS-21- Depressive	9 90 (9 09)	10.80 (7.24)	10.22 (0.71)	0.52		0.01
DASS 21 Applicate	8.89 (8.08)	10.89 (7.34)	10.22 (9.71)	0.55		0.01
DASS-21- Anxiety M (SD)	0.06 (3.15)	0 22 (2 61)	0.06 (5.77)	0.05		0.01
DASS 21 Stross	9.00 (5.15)	9.55 (5.01)	9.00 (5.77)	0.05		0.01
M (SD)	17 30 (7 42)	15 56 (6.01)	15.04 (8.01)	0.65		0.01
COPE-NVI-Social Support	17.39 (7.42)	15.50 (0.01)	15.54 (8.01)	0.05		0.01
M (SD)	34 92 (6 36)	32 11 (6 47)	31 10 (6 02)	3 13*		0.06
COPE-NVI-Avoidance	54.92 (0.50)	52.11 (0.47)	51.17 (0.02)	5.45		0.00
M (SD)	22.83 (4.91)	23 00 (4 04)	24 67 (5 60)	1.55		0.03
COPE-NVL-Positive Attitude	22.05 (4.71)	25.00 (4.04)	24.07 (5.00)	1.55		0.05
M (SD)	30.78 (5.03)	30 72 (5 26)	29 56 (5 03)	0.66		0.01
COPE-NVI-Problem solving orientation	50.70 (5.05)	50.72 (5.20)	27.50 (5.05)	0.00		0.01
M (SD)	31.97 (5.03)	31.69 (5.27)	33,28 (5,43)	0.93		0.02
COPE-NVI-Trascendental orientation		01103 (0121)		0170		0.02
M (SD)	17.44 (3.49)	17.31 (3.47)	13.08 (3.64)	17.69***		0.25
PGWBI-22-Total	(0.12)					
M (SD)	60.94 (4.20)	63.22 (3.27)	61.75 (4.49)	2.97		0.035
VVIQ-Total		Server of Server and Se		450401/228502		1999-00-00-00-00-00-00-00-00-00-00-00-00-
M (SD)	58.53 (9.37)	58.39	63.06 (9.52)	2.50		0.05
		(11.21)				
VIC-Total		<u> </u>				
M (SD)	41.56 (5.58)	43.44 (5.32)	42.97 (4.61)	1.29		0.02
VRSQ- General Physical Symptoms						
M (SD)	1.39 (1.32)	N/A	1.06 (1.33)	1.14		0.02
VRSQ- Eyes-Related Symptoms						
M (SD)	1.56 (1.32)	N/A	1.03 (0.91)	3.91		0.05

<u>Notes</u>: ^aM=Mean; ^bSD=Standard Deviation; Zoom & VR=Zoom & Virtual Reality Group; Zoom & GI=Zoom & Guided Imagery group; Audio-track & VR=Audio-track & Virtual Reality Group; STAI-Y1=State-Trait Anxiety Inventory (Y2-Trait Anxiety); DASS-21=Depressive, Anxiety, Stress Symptoms Scale-21; COPE-NVI=Coping Orientation to the Problems Experiences-New Italian Version; PGWBI=Psychological General Well-being Index; VVIQ=Vividness of Visual Imagery Questionnaire; TVIC=Test of Visual Imagery Control; VRSQ=Virtual Reality Symptom Questionnaire.

4.4.3 Aim n.1: Investigate if a customized, relaxing scenario in VR can be more effective in reducing state anxiety and heart rate frequency comparing the three groups.

To investigate the differences in state anxiety scores (STAI-Y1) between the three groups before and after the complementary relaxation experience in Session 6 (T1) and Session 7 (T2), a GLM model-based R has been deployed using "group", "block (before/after assessment based on STAI-Y1)", and "session" (T1 and T2) as predictors and a random intercept for each subject to account for repeated measures. In Table 3 are the descriptive analyses for group differences.

Dependent Variables	Time	Group	M ^a (SD) ^b
		Zoom & VR	48.06 (3.52)
	before	Zoom & GI	50.16 (5.16)
		Audio-track & VR	45.33 (5.24)
STAI-Y1 (T1)		Zoom & VR	31.17 (4.54)
	after	Zoom & GI	40.00 (4.28)
		Audio-track & VR	27.83 (5.43)
		Zoom & VR	48.89 (3.04)
	before	Zoom & GI	51.78 (3.99)

Table 3. Descriptive analysis (Mean and Standard Deviation) related to the STAI-Y1 score assessed for each group before and after Session 6 (T1) and Session 7 after one week (T2).

STAI-Y1 (T2)		Audio-track & VR	47.61 (4.15)	
	after	Zoom & VR	32.39 (4.33)	
		Zoom & GI	40.75 (4.74)	
		Audio-track & VR	32.00 (4.76)	

<u>Notes</u>: ^aM=Mean; ^bSD= Standard Deviation; VR= Virtual Reality; GI= Guided Imagery; STAI-Y1= State-Trait Anxiety Inventory (Y1-State Anxiety).

Results showed significant Group main effect ($F_{(2,105)}=54.95$, *p-value*<.001), Block main effect ($F_{(1,315)}=1835.04$; *p-value*<.001), and Session main effect ($F_{(1,315)}=28.15$; *p-value*<.001), as well as significant Block*Group interaction ($F_{(2,315)}=34.69$; *p-value*<.001), (Figure 10), and Session*Group interaction ($F_{(2,315)}=4.30$; *p-value*<.01) (Figure 12).

Pairwise post-hoc contrast of the Group main effect showed that the "Audio-track & VR" group had lower scores than both the "Zoom & VR" (β =1.93; SE=0.74; t_(1,105)=2.61; *p-value*<.05), and the "Zoom & GI" groups (β =7.48; SE=0.74; t_(1,105)=10.10; *p-value*<.001). The "Zoom & VR" group obtained lower state anxiety scores than the "Zoom & GI" group (β =-5.55; SE=0.74; t_(1,105)=-7.49; *p-value*<.001).

Looking at the within-subjects effect from the pairwise post-hoc comparisons of the Block (pre/post)*Group, all the group obtained a decrease in state anxiety after the relaxation sessions ("Zoom & VR": β =16.69; SE=0.59; t_(1,315)=28.25; *p-value*<.001; "Zoom & GI": β =10.60; SE=0.59; t_(1,315)=17.93; *p-value*<.001; "Audio-track & VR": β =16.56; SE=0.59; t_(1,315)=28.01; *p-value*<.001) (see Figure 7). Considering the between-subject effects, individuals in the "Audio-track & VR" group had lower state anxiety scores before starting the relaxation session than the "Zoom & VR" (β =-2; SE=0.85; t_(1,177)=-2.35; *p-value*<.05), and the "Zoom & GI" group (β =-4.50; SE=0.85; t_(1,177)=-5.29; *p-value*<.01). Moreover, before starting with the relaxation session, the "GI & VR" group had higher anxiety scores than the "Zoom & VR"

 $(\beta=2.50; SE=0.85; t_{(1,177)}=2.94; p-value<.01)$. After the relaxation sessions, the "Zoom & VR" group had lower anxiety scores than the "Zoom & GI" (β =-8.60; SE=0.85; t_{(1,177)}=-10.11; p-value<.001), but higher if compared with the "Audio-track & VR" condition (β =1.86; SE=0.85; t_{(1,177)}=2.19; p-value<.05). The "Audio-track & VR" group had lower scores in anxiety after relaxation also than the "Zoom & GI" group (β =-10.46; SE=0.85; t_{(1,177)}=-12.30; p-value<.001).

Figure 10. Plot about the interaction between Block (Time of Administration: Pre/Post)*Group.



Note: STAI state Block*Group interaction results. The black asterisk represents *between-group* differences, while asterisks of the same colour as the lines represent *within-group* differences. **; *p-value*<.01; ***: *p-value*<.001. Bars represent the Standard Error. STAI state=State anxiety is assessed based on the State-Trait Anxiety Inventory (Y1-State Anxiety); Pre=Assessment before the relaxation exposure; Post=Assessment after the relaxation exposure.

Other than the "Audio-track & VR" group had lower score in state anxiety before Session 6 (T1), the fact that the *d-Cohen* index and the *effect size* of the differences between the "Zoom & GI" group and the "Audio-track & VR" group were significantly higher after the relaxation experience permits to hypothesize a greater reduction of state anxiety in the "Audio-track & VR" group than the "Zoom & GI" one after T1 (Table 5). After the relaxation session at T2, the "Zoom & VR" group obtained a greater significant reduction of the perceived state anxiety than the "Zoom & GI" groups (Table 5). Compared with the other two groups, individuals in

the "Audio-track & VR" group obtained lower scores before the relaxation session at Session 7 (T2). But, also in this case, the greater difference between the "Audio-track & VR" group and the "Zoom & GI" group recorded after the relaxation at T2 can be described as a greater reduction of state anxiety in the "Audio-track & VR" group than the "Zoom & GI" one after T2 (Table 5). Otherwise, no differences emerged after Session 6 (T1) and Session 7 (T2) between the level of state anxiety of the two groups that were previously exposed to the virtual environment (Table 5). For a graphical representation of data, see Figures 11A and 11B.

Table 5. Pairwise Comparisons between Groups about differences in state anxiety recorded before and after exposures at T1 and T2 (Confidence interval adjustment: Bonferroni).

Dependent Variables	Groups		Time	Mean Difference (Before - After)	Std. Error	t-test	d-Cohen	effect-size r
STAI-Y1 (TI)	Zaam & VD	Zoom & GI		-2.02	1.05	-1.92	-0.48	-0.23
	Zoom & VR	Audio-track & VR	Before	2.99	1.24	2.41	0.61	0.29
	Audio-track & VR	Zoom & GI		-5.18***	1.19	-4.35	0.93	0.42
	Zoom & VD	Zoom & GI		-9.23***	1.18	-7.82	-2.00	0.71
	Zoom & VR	Audio-track & VR	After	2.63	1.36	1.93	0.67	-0.32
	Audio-track & VR	Zoom & GI		-12.09***	1.31	-9.23	2.49	0.78
	Zoom & VR	Zoom & GI	Before	-2.53*	0.96	-2.64	-0.81	-0.38
		Audio-track & VR		2.63*	1.06	2.48	0.35	0.17
STAI-Y1 (T2)	Audio-track & VR	Zoom & GI		-5.08***	1.02	-4.98	1.02	0.45
	Zoom & VR	Zoom & GI		-8.77***	1.21	-7.25	-1.84	0.68
		Audio-track & VR	After	0.08	1.34	0.06	0.09	-0-04
	Audio-track & VR	Zoom & GI		-8.86***	1.28	-6.92	1.84	0.68

<u>Notes</u>: *=*p*-value<.05; **=*p*-value<.01; ***=*p*-value<.001; VR=Virtual Reality; GI=Guided Imagery; STAI-Y1=State-Trait Anxiety Inventory (Y1-State Anxiety).

Figure 11. (A) Graphical description of the differences between and within subjects considering the STAI-Y1 administered before and after the relaxation session at T1. **(B).** Graphical description of the differences between and within subjects considering the STAI-Y1 administered before and after the relaxation session at T2.



Pairwise post-hoc comparisons of the Session*Group interaction showed that the "Audio-track & VR" group had lower state anxiety scores at T1 rather than T2 (β =-3.22; SE=0.59; t_(1,315)=-

5.45; *p-value*<.001). No differences between T1 and T2 have been revealed for the "Zoom & VR" (β =-1.03; SE=0.59; t_(1,315)=-1.74; *p-value*>.05) and the "Zoom & GI" groups (β =-1.18; SE=0.59; t_(1,315)=-1.99; *p-value*>.05), showing global maintenance of state anxiety levels comparing the two sessions (Figure 12). Looking at the between-subjects effect, the "Zoom & GI" group obtained higher state anxiety scores than the other two groups at T1 ("Zoom & GI" vs. "Zoom & VR": β =5.47; SE=0.85; t_(1,177)=6.43; *p-value*<.001; "Zoom & GI" vs. "Audio-track & VR": β =5.63; SE=0.85; t_(1,177)=9.99; *p-value*<.001), as well as at T2 ("Zoom & GI" vs. "Audio-track & VR": β =6.46; SE=0.85; t_(1,177)=7.59; *p-value*<.001). The "Zoom & VR" group had higher state anxiety scores than the "Audio-track & VR" group at T1 (β =3.03; SE=0.85; t_(1,177)=3.56; *p-value*<.001) (Figure 12).

Figure 12. Plot about Session (T1/T2)* Group interaction.



Note: STAI state Session*Group interaction results. The black asterisk represents *between-group* differences, while asterisks of the same color as the lines represent *within-group* differences. *:*p-value*<.05; **:*p-value*<.01; ***:*p-value*<.001. Bars represent the Standard Error. STAI state=State anxiety is assessed based on the State-Trait Anxiety Inventory (Y1-State Anxiety); T1=Session 6; T2=Session 7.

Heart rate

Differences in heart rate have been investigated between the "Zoom & VR", the "Zoom & GI" and the "Audio-track & VR" groups before, during, and after the complementary relaxation

experience in Session 6 (T1) and Session 7 (T2). A GLM model has been deployed using "group", "block (before/during/after)", and "session" as predictors and a random intercept for each subject to account for repeated measures. In Table 6 are the descriptive analysis of group differences.

Table 6. Descriptive analysis (Mean and Standard Deviation) related to Heart Rate (HR) assessed for each group before and after Session 6 (T1), and Session 7 (T2).

Dependent Variables	Time	Group	M ^a (SD) ^b
		Zoom & VR	75.02 (5.66)
HR (T1)	before	Zoom & GI	77.07 (5.64)
		Audio-track & VR	74.02 (7.22)
	duning	Zoom & VR	68.26 (5.49)
	auring	Zoom & GI	72.86 (5.23)
		Audio-track & VR	64.48 (5.24)
	aftar	Zoom & VR	69.11 (4.98)
	alter	Zoom & GI	72.22 (4.73)
		Audio-track & VR	63.53 (6.36)
		Zoom & VR	77.91 (6.28)
HR (T2)	before	Zoom & GI	77.55 (5.04)
		Audio-track & VR	73.77 (6.66)
	during	Zoom & VR	70.63 (5.45)
	uuring	Zoom & GI	71.87 (5.47)
		Audio-track & VR	65.76 (6.07)
	ofton	Zoom & VR	73.64 (5.77)
	aiter	Zoom & GI	72.80 (5.40)
		Audio-track & VR	65.49 (5.58)

<u>Notes</u>: ^aM=Mean; ^bSD=Standard Deviation; VR=Virtual Reality; GI=Guided Imagery; HR=Heart Rate; T1=Session 6; T2= Session 7.

Heart rate was found to have a significant main effect considering "group" ($F_{(2,105)}=20.71$; *p*-*value*<.001), "block" ($F_{(2,525)}=183.64$; *p*-*value*<.001) (Figure 13), and "session" ($F_{(1,525)}=18.87$; *p*-*value*<.001). Significant Block*Group interaction ($F_{(4,525)}=8.20$; *p*-*value*<.001) (Figure 14), and Session*Group interaction ($F_{(2,525)}=8.53$; *p*-*value*<.01) (Figure 16).

Pairwise post-hoc contrast of the Group main effect showed that the "Audio-track & VR" group had lower scores than both the "Zoom & VR" group (β =-4.59; SE=1; t_(1,105)=-4.58; *pvalue*<.001), and the "Zoom & GI" group (β =-6.22; SE=1; t_(1,105)=-6.21; *p*-value<.001). Data about "session" as a significant main effect are coherent with what emerged for state anxiety, highlighting a general increase in heart rate when data from Session 6 (T1) and Session 7 (T2) are compared. Pairwise post-hoc contrast of the Block main effect highlighted a generally lower level of heart rate during and after the relaxation sessions than before (Before-During: β =6.91; SE=0.40; t_(1,525)=17.17; *p*-value<.001; Before-After: β =-6.43; SE=0.40; t_(1,525)=-15.96; *p*-value<.001) (see Figure 13). This pattern should also be related to the fact that the baseline heart rate was much higher, and it can also be associated with the white coat effect due to the agitation experienced when a person knows that a professional is detecting their heart rate [e.g., 311]. The heart rate recorded during the relaxation session did not differ significantly if compared to the heart rate detected after the completion of the Sessions (During-After: β =-0.49; SE=0.40; t_(1,525)=-1.21; *p*-value>.05).





Note: Heart Rate Block*Group interaction results. The black asterisk represents the *between-block* difference considering "before" and "during", while the pink asterisk represents the *between-block* difference considering

"before" and "after". ***: *p-value*<.001. Bars represent the Standard Error. Before=Heart rate assessment before the relaxation exposure; During= Heart Rate assessment during the relaxation exposure; After=Heart Rate assessment after the relaxation exposure.

Looking at the within-subjects effect from the pairwise post-hoc comparisons of the block (before/during/after)*group interaction, the "Audio-track & VR" group had a reduction in heart rate during and after the relaxation exposure if compared to the heart rate recorded before the sessions (Before vs. During: β =8.77; SE=0.70; t_(1,525)=12.58; *p-value*<.001; Before vs. After: β =9.38; SE=0.70; t_(1.525)=13.46; *p*-value<.001). The same pattern emerged for the "Zoom & GI" group (Before vs During: β =4.95; SE=0.70; t_(1,525)=7.09; *p*-value>.001; Before vs After: β =4.80; SE=0.70; t_(1,525)=6.89; *p*-value<.001). Indeed, the "Audio-track & VR" and the "Zoom" & GI" groups obtained a significant decrease in HR during the relaxation sessions that persisted and did not change after the relaxation sessions. The HR of the "Zoom & VR" group decreased significantly during the relaxation and showed a small but significant increase once the sessions finished (Before vs. During: β =7.02; SE=0.70; t_(1.525)=10.07; *p-value*<.001; Before vs. After: β =5.09; SE=0.70; t_(1,525)=7.30; *p*-value<.001; During vs. After: β =-1.93; SE=0.70; t_(1,525)=-2.77; p-value<.001) (Figure 10). Considering the between-subject effect, individuals in the "Audio-track & VR" group had lower state anxiety scores before starting the relaxation session than the "Zoom & VR" (β=-2.57; SE=1.15; t_(1,180)=-2.23; *p-value*<.05), and the "Zoom & GI" group (β =-3.42; SE=1.15; t_(1,180)=-2.96; *p*-value<.05); these differences have been maintained during and after the relaxation sessions (During "Audio-track & VR" vs. "Zoom & VR": β=-4.32; SE=1.15; t_(1,180)=-3.75; *p-value*<.01; "Audio-track & VR" vs "Zoom & GI": β=-7.24; SE=1.15; t_(1,180)=-6.29; *p-value*<.001; After "Audio-track & VR" vs "Zoom & VR": β=-6.86; SE=1.15; t_(1,180)=-5.96; *p-value*<.001; "Audio-track & VR" vs "Zoom & GI": β=-7.99; SE=1.15; t_(1.180)=-6.94; *p-value*<.001). Differences between the "Zoom & VR" and the "Zoom & GI" emerged during the relaxation and highlighted a greater decrease in HR for the "Zoom & VR" group (β =-2.92; SE=1.15; t_(1,180)=-2.53; *p*-value<.05).

Figure 14. Plot about Block (Time of Administration: pre/during/post)*Group interaction.



Note: Heart Rate Session*Group interaction results. The asterisk of the same color as the lines represents *within-group* differences. ***: *p-value*<.001. Bars represent the Standard Error. Before=Heart rate assessment before the relaxation exposure; During= Heart rate assessment during the relaxation exposure; After= Heart rate assessment after the relaxation exposure.

At Session 6 (T1), "Zoom & VR", and "Audio-track & VR" groups had lower heart rate frequency during the exposure in the virtual environment than individuals in the Guided Imagery condition (Table 7). Moreover, after the exposure a lower heart rate frequency was recorded for the "Audio-track & VR" group, followed by the "Zoom & VR" group, and, lastly by individuals in the "Zoom & GI" condition, that obtain higher heart frequency scores than the participants in the other two VR conditions (Zoom & GI: M=72.22; SD=4.73; Zoom & VR: M=69.11; SD=4.98; Audio-track & VR: M=63.53; SD=6.36) (Table 7).

At Session 7 (T2) individuals in the Audio-track & VR condition had lower scores of heart rate frequency before starting with the relaxation procedure; therefore, it is essential to note that this difference increased during and after the relaxation session (Before Session 7: d-Cohen=0.64; effect size r=0.30; During Session 7: d-Cohen=1.06; effect size r=0.47; <u>After Session 7</u>: d-Cohen=1.33; effect size r=0.65), underlined a higher decrease of heart rate frequency during and following the relaxation session in individuals who in the previous Session 6 (T1) experienced the relaxation in the virtual customized environment (Table 8).

Individuals in the "Audio-track & VR" group obtained lower scores in heart rate frequency during and after the relaxation at Session 7 (T2) than the other two groups. Instead, no differences emerged between the "Zoom & VR" and the "Zoom & GI" groups (see Table 8). In figures 15 A and 15 B are described in a graphical way the differences between groups and the interactions.

Table 7. Pairwise Comparisons between Groups about differences in heart rate recorded before, during and after

 exposures at T1 (Confidence interval adjustment: Bonferroni).

Dependent Variables	Gro	oups	Time	Mean Difference (Before - After)	Std. Error	t-test	d-Cohen	effect-size r
HR (T1)	7. A 17.	Zoom & GI		-2.49	1.63	-1.53	-0.36	-0.18
	Loom & VK	Audio-track & VR	Before	0.77	1.78	0.43	0.15	0.08
	Audio-track & VR	Zoom & GI		-3.25	1.71	-1.9	0.47	0.23
	Zoom & VR	Zoom & GI	During	-4.87***	1.39	-3.50	-0.86	-0.39
		Audio-track & VR		3.19	1.52	2.10	0.70	0.33
	Audio-track & VR	Zoom & GI		-8.05***	1.46	-5.51	1.60	0.62
	Zeem & VD	Zoom & GI		-3.62*	1.42	-2.55	-0.64	-0.30
	Loom & vK	Audio-track & VR	After	5.05**	1.55	3.26	0.98	0.44
	Audio-track & VR	Zoom & GI		-8.67***	1.49	-5.82	1.55	0.61

Table 8. Pairwise Comparisons between Groups about differences in heart rate recorded before, during and after

 exposures at T2 (Confidence interval adjustment: Bonferroni).

Dependent Variables	Groups		Time	Mean Difference (Before - After)	Std. Error	t-test	d-Cohen	effect-size r
	Zoom & VR	Zoom & GI		-0.31	1.59	-0.19	0.06	0.03
		Audio-track & VR	Before	4.14	1.74	2.38	0.64	0.30
	Audio-track & VR	Zoom & GI		4.45*	1.66	2.68	0.64	0.30
	Zoom & VR	Zoom & GI	During	-2.29	1.49	-1.54	-0.23	-0.11
		Audio-track & VR		4.11*	1.63	2.52	0.84	0.39
	Audio-track & VR	Zoom & GI		-6.40***	1.56	-4.10	1.06	0.47
HR (T2)	Zoom & VR	Zoom & GI		-0.23	1.47	-0.16	0.15	0.07
		Audio-track & VR	After	7.69***	1.60	4.81	1.44	0.58
	Audio-track & VR	Zoom & GI		-7.92***	1.53	-5.18	1.33	0.55

<u>Notes</u>: * = p-value <.05; ** = p-value <.01; *** = p-value <.001; VR= Virtual Reality; GI= Guided Imagery; HR= Heart rate frequency.

Figure 15. (A) Graphical description of the differences between and within subjects considering the HR administered before and after the relaxation session at T1. (B). Graphical description of the differences between and within subjects considering the HR administered before and after the relaxation session at T2.



Pairwise post-hoc comparisons of the Session*Group interaction showed no differences in heart rate between T1 and T2 for the "Audio-track & VR" group (β =-0.99; SE=1.08; t_(1,525)=-1.75; *p-value*>.05), and the "Zoom & GI" group (β =-0.02; SE=1.08; t_(1,525)=-0.04; *p*-

value>.05). Instead, a global increased in heart rate emerged comparing the assessment at T1 and T2 only for the "Zoom & VR" group (β =-3.26; SE=1.08; t_(1,525)=-5.73; *p-value*<.001) (Figure 16). Considering the between-subjects effect, the "Zoom & GI" group obtained higher state anxiety scores than the other two groups at T1 (vs. "Audio-track & VR": β =6.71; SE=1.08; t_(1,141)=6.21; *p-value*>.001; vs. "Zoom & VR": β =3.26; SE=1.08; t_(1,141)=3.01; *p-value*>.001). Moreover, the "Audio-track & VR" group had a lower heart rate level recorded before the relaxation session than the "Zoom & VR" group (β =-3.45; SE=1.08; t_(1,141)=-3.20; *p-value*>.001). At T2, the "Audio-track & VR" obtained a lower HR level than both the "Zoom & GI" (β =5.73; SE=1.08; t_(1,141)=5.31; *p-value*<.001) and the "Zoom & VR" group (β =5.72; SE=1.08; t_(1,141)=5.30; *p-value*<.001) (Figure 16).

Notably, the greater increase in heart rate obtained by the "Zoom & VR" group from T1 to T2 probably explains the global decrease highlighted, considering "session" as a main effect. Even if the increase is recorded for the "Audio-track & VR" group too, the "Zoom & VR" group seems to be less incisive in promoting relaxation using the strategies learned during the training sessions (s1-s4), and T1, compared to what emerged by data of the "Audio & VR" group. This outcome seems not to be coherent with what is highlighted for STAI-Y1, considering the increase in self-reported state anxiety increase from T1 and T2 but with a lower effect than what is highlighted by the Heart Rate.




Note: Heart Rate Session*Group interaction results. The black asterisk represents *between-subjects* differences, while the same color as the lines represent *within-group* differences. ***: *p-value*<.001. Bars represent the Standard Error. Before=Heart Rate assessment before the relaxation exposure; During=Heart Rate assessment during the relaxation exposure; After=Heart Rate assessment after the relaxation exposure.

4.4.4 Aim n.2: Investigate the effect of the entire relaxation protocol on state anxiety

To investigate the differences in the STAI-Y1 assessed before and after each of the six PMR training sessions for the "Zoom & VR", "Zoom & GI" and the "Audio-track & VR" groups, a linear mixed-effect model on the scores has been performed, using "group", "block (pre/post assessment based on the STAI-Y1)", and "session" as predictors and a random intercept for each subject to account for repeated measures. For a graphical representation of the STAI-Y1 assessment's phases, see Figure 17. In Table 5 are the descriptive analysis of STAI-Y1 filled out before and after the training sessions (descriptive statistics of the STAI-Y1 assessed at T1 and T2 are in Table 3).

Figure 17. A flowchart about relaxation sessions in which the participants of the three groups filled out the STAI-Y1.



<u>Notes</u>: VR= Virtual Reality; GI= Guided Imagery; STAI-Y1= State-Trait Anxiety Inventory (Y1-State Anxiety); s1-4= Sessions from 1 to 4; T1= Session 6; T2 (Session 7); PMR= Progressive Muscle Relaxation Technique.

Table 5. Descriptive analysis (Mean and Standard Deviation) related to the STAI-Y1 score assessed for ea	ich
group before and after each training session (s1-s4), Session 6 (T1), and Session 7 (T2).	

Dependent Variables	Time	Group	M ^a (SD) ^b
	before	Zoom & VR	48.19 (5.32)
STAI-Y1 (s1)		Zoom & GI	48.06 (4.14)
		Audio-track & VR	45.61 (9.91)
	after	Zoom & VR	42.49 (4.23)

		Zoom & GI	42.81 (3.82)
		Audio-track & VR	34.50 (7.17)
	before	Zoom & VR	48.57 (4.09)
		Zoom & GI	45.89 (4.30)
STAI-Y1 (s2)		Audio-track & VR	45.17 (6.42)
	oftor	Zoom & VR	41.23 (3.04)
		Zoom & GI	41.31 (3.76)
		Audio-track & VR	33.53 (7.88)
		Zoom & VR	44.77 (3.40)
	before	Zoom & GI	43.69 (4.92)
STAI-Y1 (\$3)		Audio-track & VR	42.17 (7.93)
		Zoom & VR	42.57 (3.10)

af		Zoom & GI	40.97 (3.89)
		Audio-track & VR	33.58 (6.39)
	before	Zoom & VR	45.74 (4.46)
STAI-Y1 (s4)		Zoom & GI	43.92 (5.27)
		Audio-track & VR	42.31 (9.18)
	after	Zoom & VR	44.14 (4.07)
		Zoom & GI	41.33 (4.62)
		Audio-track & VR	34.19 (7.95)

<u>Notes</u>: ^aM=Mean; ^bSD= Standard Deviation; VR= Virtual Reality; GI= Guided Imagery; STAI-Y1= State-Trait Anxiety Inventory (Y1-State Anxiety); s1-4= sessions from 1 to 4.

Results showed significant "group" ($F_{(2,105)}=27.55$; *p-value*<.001), "block" ($F_{(1,1154)}=1383.46$; *p-value*<.001), and "session" main effects ($F_{(5,1154)}=14.21$; *p-value*<.001) (Figure 18), as well as significant Block*Session ($F_{(5,1154)}=65.86$; *p-value*<.001), Block*Group ($F_{(2,1154)}=54.49$; *p-value*<.001), Session*Group ($F_{(10,1154)}=13.21$; *p-value*<.001) (Figure 19), and Session*Block*Group interactions ($F_{(10,1154)}=3.55$; *p-value*<.001) (Figure 20).

Pairwise post-hoc contrast of "group" as main effect showed that the "Audio-track & VR" group had lower scores than both the "Zoom & VR" (β =-4.51; SE=0.80; t_(1,105)=-5.65; *p*-*value*<.001), and the "Zoom & GI" group (β =-5.58; SE=0.80; t_(1,105)=-6.99; *p*-*value*<.001).

Pairwise post-hoc contrast of "session" as main effect showed differences between Session 1 and 2 (β =0.98; SE=0.41; t_(1,1154)=2.38; *p-value*<.05), Session 2 and 3 (β =1.34; SE=0.41; t_(1,1154)=3.25; *p-value*<.01), highlighting a decrease in state anxiety which reaches its peak at Session 3, and remain stable until the end of Session 4 that correspond to the last PMR training session (β =-0.62; SE=0.41; t_(1,1154)=-1.51; *p-value*>.05). A further noteworthy decrease in anxiety is between Session 4 and T1 (β =3.18; SE=0.41; t_(1,1154)=7.73; *p-value*<.001), shed in light an additional decrease in anxiety once participants complete the Relaxation session complementary to the exposure in the VR or GI scenario (T1). Once participants are exposed to the self-directed relaxation session at T2, they globally obtain state anxiety levels comparable to what they reported after the second training session (Session 2) (*p-value*>.05).

Moreover, as stated before, the statistically significant difference related to "block" revealed a general decrease in state anxiety from the STAI-Y1 score administered before and after each session. This pattern characterized each group separately, based on the significance of the block*group interaction (Pre/Post "Audio-track & VR": β =12.09; SE=0.41; t_(1,1154)=29.39; *p*-*value*<.001; Pre/Post "Zoom & VR": β =8.35; SE=0.41; t_(1,1154)=20.26; *p*-*value*<.001; Pre/Post "Zoom & GI": β =6.08; SE=0.41; t_(1,1154)=14.77; *p*-*value*<.001).

Figure 18. Plot about Session (s1-s4 and T1-T2) as a main effect.



Notes: STAI state Session*Group interaction results. The black asterisks represent the main *within-sessions* differences. ***: p-value <.001. Bars represent the Standard Error. s1-s4= the four PMR training sessions; T1= Session 6; T2= Session 7.

The Session*Group interaction is more interesting than what can be seen from the Session main effect since it shows that the maximum decrease between s4 and T1 that emerged from the main effect is due to the decrease in anxiety scores obtained by the "Zoom & VR" group (β =5.25; SE=0.71; t_(1,1154)=7.37; *p-value*<.001), and, partly, by the "Audio-track & VR" group (β =1.67; SE=0.71; t_(1,1154)=2.34; *p-value*<.05). It is also notable that the "Zoom & GI" group has an opposite trend as anxiety at T1 had already started to decrease compared to the training sessions (β =-2.46; SE=0.71; t_(1,1154)=-3.45; *p-value*<.01). The increase at T2 seems to be pushed by the "Audio-track & VR" group (β =-3.22; SE=0.71; t_(1,1154)=-4.52; *p-value*<.001), while in the groups that were previously exposed to the Zoom relaxation training the level of anxiety is almost equal (*p-value*>.05) (Figure 19).

Figure 19. Plot about Session*Group interaction.



<u>Notes</u>: The asterisks of the same colour as the lines represent *within-group* differences. *:*p-value*<.05; **:*p-value*<.01; ***:*p-value*<.001. Bars represent the Standard Error. STAI state=State anxiety is assessed based on the State-Trait Anxiety Inventory (Y1-State Anxiety); s1-s4= the four training sessions by Zoom or Audio-track; T1=Session 6; T2=Session 7.

Considering the Block*Session*Group interaction, interesting is that the state anxiety felt by the "Zoom & GI" group, recorded before and after every single session (from s1 to T2), shows a constant decrease but remains relatively stable once participants are exposed to the Guided Imagery scenario (*p-value*>0.05). This does not happen in other groups: in the "Audio-track & VR" group, state anxiety undergoes a large reduction at T1 (β =6.36; SE=1.02; t_(1,1154)=6.31; *pvalue*<.001), and then increases at T2 (β =-4.17; SE=1.01; t_(1,1154)=4.13; *p*-value<.01). At the same time, in the "Zoom & VR" group, anxiety decreases from s4 and T1 (β =12.92; SE=1.01; t_(1,1154)=12.82; *p-value*<.001), and remains down without changing from T1 and T2 (*p*value<.05), showing how the latter seems to be the best condition to promote a decrease in anxiety. As was the case for all groups, the GI group greatly increased state anxiety before the in-presence sessions if comparing anxiety recorded before at s4 and T1 (β =-4.14; SE=1.25; t_(1,533)=3.31; *p-value*<.01). But, unlike evidenced by the groups exposed to virtual reality, the "Zoom & GI" group has not reached a subsequent sharp decrease. The increase in anxiety experienced before the in-presence sessions (at T1 and T2) and the reduced decrease recorded immediately after helps to explain the overall increase in anxiety experienced by the "Zoom & GI" group, if compared to the general level obtained during the training sessions. Instead, the decrease in anxiety after the exposure to VR was so significant that it was evident, even if, as for the group "Zoom & GI", subjects exposed to the VR conditions had a much higher level of state anxiety before the in-presence sessions than the anxiety recorded before the training sessions done by Zoom (Figure 20).

Moreover, it is interesting to note how, from the third training session, individuals of all groups experienced lower anxiety levels before the training than what they experienced before s2 (*p*-*value* >.01), (Figure 20); it could be outlined that already after two web-based sessions, the participants began to get used and be more confident with the relaxation procedure and it is related with the above-mentioned decrease in state anxiety which reaches its peak at Session 3 revealed by considering "session" as the main effect (Figure 18).

Figure 20. The plot about Block*Session*Group interaction.



<u>Notes</u>: The asterisks of the same color as the lines represent *within-group* differences. *:*p-value*<.05; **:*p-value*<.01; ***:*p-value*<.001. Bars represent the Standard Error. STAI state=State anxiety is assessed based on the State-Trait Anxiety Inventory (Y1-State Anxiety); s1-s4= the four training sessions by Zoom or Audio-track; T1=Session 6; T2=Session 7.

4.4.5 Aim n. 3: Understand if VR promotes a better sense of presence and engagement in the scenario compared to GI after Session 6 (T1) and if it helps to recall the image and be immersed in the relaxing scenario in Session 7 (T2)

To investigate the second aim of the current study, a GLM model is used for each of the ITC-SOPI subscales. In Table 6 are the descriptive statistics.

Table 6. Descriptive analysis (Mean and Standard Deviation) related to the ITC-SOPI subscales scores assessed for each group after Session 6 (T1), and Session 7 (T2).



ITC-SOPI Sense of Physical space	Zoom & GI	57.00 (8.99)
(presence) (T1)	Audio-track & VR	62.69 (7.68)
ITC-SOPI	Zoom & VR	46.61 (4.66)
Engagement (T1)	Zoom & GI	42.42 (5.71)
	Audio-track & VR	50.81 (3.65)
ITC-SOPI	Zoom & VR	19.92 (2.12)
Ecological Validity (T1)	Zoom & GI	17.67 (2.75)
	Audio-track & VR	19.31 (2.58)
ITC-SOPI Negative Effects	Zoom & VR	13.14 (5.17)
(T1)	Zoom & GI	12.08 (1.86)
	Audio-track & VR	9.86 (4.47)
ITC-SOPI	Zoom & VR	50.06 (11.88)
Sense of Presence (T2)	Zoom & GI	41.22 (19.81)
	Audio-track & VR	53.50 (7.88)
ITC-SOPI	Zoom & VR	40.50 (6.88)
Engagement (T2)	Zoom & GI	40.36 (6.16)
	Audio-track & VR	42.47 (5.46)
ITC-SOPI	Zoom & VR	17.44 (3.85)
Ecological Validity (T2)	Zoom & GI	14.17 (4.06)
	Audio-track & VR	16.89 (3.16)
ITC-SOPI Negative Effects	Zoom & VR	10.89 (2.42)
(T2)	Zoom & GI	10.89 (2.82)

Notes: M=Mean; SD= Standard Deviation; VR= Virtual Reality; GI= Guided Imagery.

ITC-SOPI "Sense of Physical Space"

Considering the ITC-SOPI "Sense of Physical Space" subscale, significant main effects between and within subjects were found considering "group" ($F_{(2,210)}=11.14$; *p-value*<.001), and "session" ($F_{(2,210)}=54.37$; *p-value*<.001). In general, scores related to the perception of "being immersed" in the scenario decreased independently of the group once participants experienced the self-directed relaxation session at T2.

About "group" as main effect, the "Zoom & GI" group experienced a lower sense of presence in the relaxing scenario rather than the other two groups in which individuals were exposed in a virtual environment ("Zoom & GI" vs. "Zoom & VR": β =-5.94; SE=1.94; t_(1,105)=-3.07; *pvalue*<.01; "Zoom & GI" vs. "Audio-track & VR": β =-8.99; SE=1.94; t_(1,105)=-4.64; *pvalue*<.001).

ITC-SOPI "Engagement"

Considering the ITC-SOPI "Engagement" subscale, significant main effects between and within subjects were found considering "group" ($F_{(2,210)}=16.07$; *p-value*<.001), and "session" ($F_{(1,210)}=55.04$; *p-value*<.001). Moreover, the Session*Group interaction was revealed ($F_{(2,210)}=5.69$; *p-value*<.001) (Figure 21).

The "Zoom & GI" group experienced lower engagement in the relaxing scenario rather than groups exposed to a virtual environment ("Zoom & GI" vs. "Zoom & VR": β =-2.08; SE=0.92; $t_{(1,105)}$ =-2.27; *p-value*<.05; "Zoom & GI" vs. "Audio-track & VR": β =-5.17; SE=0.92; $t_{(1,105)}$ =-5.63; *p-value*<.001). Moreover, the "Audio-track & VR" group showed higher engagement in the VR context than the "Zoom & VR" group (β =3.08; SE=0.92; $t_{(1,105)}$ =3.36; *p-value*<.01). As for the "Physical Space" subscale, engagement in the scenario decreased once participants experienced the self-directed relaxation session at T2.

Pairwise post-hoc comparisons of the Session*Group interaction showed that the two VR groups had higher engagement in the experience at T1 rather than T2 ("Zoom & VR": β =6.11; SE=1.30; t_(1,105)=4.71; *p-value*<.01; "Audio-track & VR": β =8.33; SE=1.30; t_(1,105)=6.42; *p-value*<.001). No differences emerged for the "Zoom & GI" group (β =2.22; SE=1.30; t_(1,105)=1.71; *p-value*<.001). Looking at between-subjects effect, "Audio-track & VR" group

was more engaged in the experience than the other two groups at T1 ("Audio-track & VR" vs. "Zoom & VR": β =4.19; SE=1.30; t_(1,210)=3.23; *p-value*<.01; "Audio-track & VR" vs. "Zoom & GI": β =8.22; SE=1.30; t_(1,210)=6.34; *p-value*<.001), but not at T2, were no differences have been highlighted if comparing groups with each other (*p-value*>.05). The "Zoom & VR" group had a higher score in the engagement scale than the "Zoom & GI" group at T1 (β =4.03; SE=1.30; t_(1,210)=3.11; *p-value*<.01). An interesting information from the outcome is that no differences in engagement emerged from T1 and T2 for the "Zoom & GI" group, further highlighting the importance of having made the experience in a more immersive context in VR on the sense of engagement felt during the exposure.





Note: ITC-SOPI Engagement Session*Group interaction results. The black asterisk represents *between-subjects* differences, while the same color as the lines represent *within-group* differences. **; *p-value*<.01; ***: *p-value*<.001. Bars represent the Standard Error. T1=Session 6; T2= Session 7.

ITC-SOPI "Ecological Validity"

Considering the ITC-SOPI "Ecological Validity" subscale, significant main effects between and within subjects were found considering "group" ($F_{(2,105)}=11.69$; *p-value*<.001), and "session" ($F_{(1,105)}=60.95$; *p-value*<.001).

The "Zoom & GI" group perceived their imagined environment less credible when they have to refer to the level of realism of the scenario's content rather than groups exposed in a virtual environment ("Zoom & GI" vs. "Zoom & VR": β =-2.76; SE=0.60; t_(1,105)=-4.59; *p*-value<.001; "Zoom & GI" vs. "Audio-track & VR": β =-2.18; SE=0.60; t_(1,105)=-3.62; *p*-value<.01). Switching the attention on the "session" main effect, it is revealed that the perception of realism about the scenario in which individuals were exposed, decreased once participants experienced the self-directed relaxation session at T2.

ITC-SOPI "Negative Effects"

Considering the ITC-SOPI "Negative Effects" subscale, significant main effects between and within subjects were found considering "group" ($F_{(2,105)}=8.00$; *p-value*<.001) and "session" ($F_{(1,105)}=7.33$; *p-value*<.001).

As expected, considering that the participants exposed to the imagined scenarios had not to wear the headset, it was found that the "GI & VR" group experienced fewer symptoms (e.g., eye discomfort) than the other two groups ("Zoom & GI" vs. "Zoom & VR": β =-2.26; SE=0.59; $t_{(1,105)}$ =-3.82; *p*-value<.01; "Zoom & GI" vs. "Audio-track & VR": β =-2.69; SE=1.30; $t_{(1,105)}$ =-3.97; *p*-value<.01). In relation to the significance of the "session" as a main effect, the perception of discomfort experienced when individuals had to wear the headset was higher at T1 and decreased at T2, when subjects have just to recall the virtual scenarios that they experienced in the previous session.

4.4.6 Aim n.4: Can the sense of presence, engagement, and the perception that the scenario is realistic to predict a more significant reduction in state anxiety after the virtual reality experience?

A linear regression analysis was conducted on the "Zoom & VR" and the "Zoom & GI" groups data to investigate whether the sense of presence, engagement, and the perception of a realistic scenario predicted a more significant reduction in state anxiety after the relaxation experiences. We considered three ITC-SOPI scales (Sense of Presence, Engagement, and Ecological Validity) as predictors and the STAI-Y1 scores recorded after the VR experience or the GI experience as dependent variables. For the Virtual Reality group, data showed that the factors considered in the model are related to the state anxiety level after the VR experience (Adjusted R Square=0.17; $F_{(3,35)}$ =3.32; *p-value*<.05). The data indicates that engagement (ITC-SOPI Engagement) is predictive of state anxiety levels (STAI-Y1) experienced after VR (see Table

7). In the GI group, there are no effects observed after the experience in relation to state anxiety level (Adjusted R Square=0.008; $F_{(3,35)}=1.10$; *p-value*>.05) (see Table 8).

The same investigation has been executed at T2. As for T1, considering the VR group, data showed that the factors considered in the model are related to the state anxiety level after the VR experience (Adjusted R Square= 0.173; $F_{(3,35)}=3.45$; *p-value*<.05). Data evidenced that ecological validity (ITC-SOPI Ecological Validity) is predictive of the state anxiety levels (STAI-Y1) experienced after the relaxation session seven at T2 (see Table 7). No significant effects were observed for the GI group (Adjusted R Square= -0.05; $F_{(3,35)}=0.42$; *p-value* >.05) (see Table 8).

State anxiety level after the VR experience (T1)	State anxiety level after the session 7 (T2)
β(t)	β(t)
0.04 (.34)	0.16 (1.34)
-0.52 (-2.67*)	-0.19 (-1.31)
-0.05 (0.15)	-0.77 (-2.54*)
	State anxiety level after the VR experience (T1) β(t) 0.04 (.34) -0.52 (-2.67*) -0.05 (0.15)

Table 7. Linear regression models related to the Virtual Reality group.

Table 8. Linear regression models related to the Guided Imagery group.

Predictors	State anxiety level after the VR experience (T1)	State anxiety level after the session 7 (T2)
	β (t)	β (t)
ITC-SOPI Sense of Presence	-0.32 (-1.31)	-0.14 (-1.06)

ITC-SOPI Engagement	0.05 (0.22)	0.10 (0.54)
ITC-SOPI Ecological	0.14 (0.83)	0.30 (0.98)
Validity		

4.4.7 Aim n.5: Investigate group differences regarding trait anxiety, stress, depression, and coping strategies.

STAI-Y2-Trait Anxiety

To investigate the differences in trait anxiety using the STAI-Y2 between the three groups at baseline (T0) and after Session 7 (T2), we applied a GLM model using random intercepts. Significant main effects of "group" ($F_{(2,105)}=11.01$; *p-value*<.001) and "session" ($F_{(1,105)}=43.83$; *p-value*<.001) have been revealed, as well as a significant interaction between the variables "time of assessment" and "group" factors ($F_{(2,105)}=11.62$; *p-value*<.001) emerged. In Table 9 are the descriptive statistics of STAI-Y2 filled out at T0 and T2.

Table 9. Descriptive analysis (Mean and Standard Deviation) related to the STAI-Y2 score assessed for each group at T0 and after Session 7 (T2).

Dependent Variables	Time	Group type	M ^a (SD) ^b
Γ	T0	Zoom & VR	48.06 (3.78)
		Zoom & GI	47.56 (3.56)
STAI-Y2		Audio-track & VR	45.33 (7.50)
	T2	Zoom & VR	44.25 (4.07)
		Zoom & GI	47.36 (4.29)
		Audio-track & VR	39.08 (8.46)

Notes: ^aM=Mean; ^bSD= Standard Deviation; VR= Virtual Reality; GI= Guided Imagery.

In general, the "Audio-track & VR" group had lower scores in trait anxiety than the other groups, with a greater difference if compared with the "Zoom & GI" group ("Audio-track & VR" vs. "Zoom & GI": β =-5.25; SE=1.17; t_(1,105)=-4.51; *p-value*<.001; "Audio-track & VR" vs "Zoom & VR": β =-3.94; SE=1.17; t_(1,105)=-3.39; *p-value*<.01). Turning the focus on within-subjects effect, trait anxiety had a greater decrease when assessed at the end of the entire relaxation protocol (T2) than at T0.

It was interesting to observe that once the Session*Group interaction is analyzed, no significant changes in anxiety were recorded from T0 and T2 for the group that had the relaxation experience in the Guided Imagery condition (*p-value*>.05) (Figure 22). Since both the VR groups obtained a reduction in anxiety over time, a positive impact of the use of a more realistic, customized, and immersive virtual environment could have promoted a stronger reduction of anxiety ("Zoom & VR": β =3.81; SE=0.89; t_(1,105)=4.26; *p-value*<.001; "Audio-track & VR": β =6.25; SE=0.89; t_(1,105)=6.99; *p-value*<.001). Otherwise, the greater differences between the "Audio-track & VR" and the "Zoom & VR" groups at T2 (β =3.11; SE=1.33; t_(1,162)=2.35; *p-value*<.05) lead to consider also the empowered impact of the relaxation training (sessions t1-t4) by Audio-track in the anxiety reduction.



Figure 22. Plot about Session*Group interaction.

Note: STAI trait Session*Group interaction results. The black asterisk represents *between-subjects* differences, while the same color as the lines represent *within-group* differences. *: *p-value*<.05; **; *p-value*<.01; ***: *p-value*<.001. Bars represent the Standard Error. T0= Assessment at Baseline; T2= Session 7.

DASS-21-Depressive, Anxiety, Stress Symptoms Scale-21

We investigated differences in depressive, anxiety, and stress symptomatology using the DASS-21 between and within the "Zoom & VR", the "Zoom & GI" and the "Audio-track & VR" groups before and after the complementary relaxation experience in Session 6 (T1) and Session 7 (T2). As for the previous comparisons, a GLM model has been deployed with "group", "block (pre/post assessment based on DASS-21 subscales score)", and "session" as predictors and a random intercept for each subject to account for repeated measures. In Table 10 are the descriptive statistics of the DASS-21 subscales filled out by participants of all groups at T0 and T2.

Dependent Variables	Time	Group	M ^a (SD) ^b
DASS-21 - Depressive	TO	Zoom & VR	8.89 (8.08)
		Zoom & GI	10.89 (7.33)
		Audio-track & VR	10.22 (9.71)
	T2	Zoom & VR	8.39 (3.94)
		Zoom & GI	10.56 (6.23)
		Audio-track & VR	6.89 (6.97)

 Table 10. Descriptive analysis (Mean and Standard Deviation) related to the DASS-21 score assessed for each group at T0 and after Session 7 (T2).

		Zoom & VR	9.06 (3.15)
DASS-21 - Anxiety	ТО	Zoom & GI	9.33 (3.61)
		Audio-track & VR	9.06 (5.77)
		Zoom & VR	6.83 (3.95)
	T2	Zoom & GI	5.61 (3.85)
		Audio-track & VR	4.39 (4.68)
	ТО	Zoom & VR	17.39 (7.42)
		Zoom & GI	15.56 (6.01)
DASS-21 - Stress		Audio-track & VR	15.94 (8.01)
	T2	Zoom & VR	13.56 (4.14)
		Zoom & GI	12.67 (4.99)
		Audio-track & VR	12.72 (4.93)

<u>Notes</u>: ^aM=Mean; ^bSD= Standard Deviation; VR= Virtual Reality; GI= Guided Imagery; T0=Baseline Session; T2=Session 7.

Considering the DASS-21 - Depressive Scale, "session" as a main effect was significant ($F_{(1,105)}=5.84$; *p-value*<.01), which outlined a decrease of depressive symptomatology comparing T0 and T2. Coherently with what emerged from the STAI-Y2, the DASS-21 "Anxiety" scores decreased from T0 and T1 ($F_{(1,105)}=93.86$; *p-value*<.001). Moreover, a significant interaction Session*Group is highlighted ($F_{(2,105)}=3.80$; *p-value*<.05) (Figure 23). Precisely, for all three groups a decrease in anxiety was revealed ("Zoom & VR": $\beta=2.22$; SE=0.63; $t_{(1,105)}=3.51$; *p-value*<.001; "Audio-track & VR": $\beta=4.67$; SE=1.00; $t_{(1,154)}=4.65$; *p-value*<.001; "Zoom & GI": $\beta=3.44$; SE=1.00; $t_{(1,154)}=3.44$; *p-value*<.001). About the betweensubjects comparisons, no differences emerged at T0 (Baseline assessment) (*p-value*>.05). Differently, at T2, the "Audio-track & VR" group obtained a larger decrease in anxiety symptoms than the "Zoom & VR" group ($\beta=2.44$; SE=1.00; $t_{(1,154)}=2.44$; *p-value*<.05). The DASS-21 - Stress Scale was significantly only for "session" as a main effect ($F_{(1,105)}=48.36$; *p-value*<.001), highlighting a decrease in stress symptomatology from T0 to T2.





<u>Note</u>: DASS-21 Anxiety Session*Group interaction results. The black asterisk represents *between-subjects* differences, while the same color as the lines represent *within-group* differences. *: p < .05; **; p < .01; ***: p < .001. Bars represent the Standard Error. T0= Assessment at Baseline; T2= Session 7.

PGWBI-Psychological General Well-being

To investigate the differences in psychological well-being within and between the "Zoom & VR", the "Zoom & GI" and the "Audio-track & VR" groups at T0 (Baseline assessment), and Session 7 (T2), a GLM model has been deployed using "group", "block (pre/post assessment based on PGWBI total score)", and "session" as predictors and a random intercept for each subject to account for repeated measures. In Table 10 are the descriptive statistics of the PGWBI filled out at T0 and T2.

Table 10. Descriptive analysis (Mean and Standard Deviation) related to the PGWBI total score assessed for each group at T0 and after Session 7 (T2).

Dependent Variables	Time	Group type	M ^a (SD) ^b
Γ	T0	Zoom & VR	60.94 (4.20)
		Zoom & GI	63.22 (3.27)
PGWBI-Total		Audio-track & VR	61.75 (4.49)
	T2	Zoom & VR	60.86 (6.97)
		Zoom & GI	63.53 (9.48)
		Audio-track & VR	62.61 (3.89)

<u>Notes</u>: ^aM=Mean; ^bSD= Standard Deviation.

A significant "session" as a main effect ($F_{(1,105)}$ =43.83; *p-value*<.001) has been revealed, outlining an increase in the global psychological well-being condition for all the groups (see Figure 24).

Figure 24. Plot about Session (T0/T2) as a main effect.



<u>Note</u>: PGWBI Session as a main effect results. The black asterisk represents *between-session* differences. ***: *p-value* <.001. Bars represent the Standard Error. T0= Assessment at Baseline; T2= Session 7.

COPE-NVI - Coping Orientation to Problems Experienced-Nuova Versione Italiana

We investigated differences in strategies people used to cope with problems or stress based on the COPE-NVI between and within the "Zoom & VR", the "Zoom & GI" and the "Audio-track & VR" groups at T0 (Baseline assessment) and Session 7 (T2). A GLM model has been deployed with "group", "block (pre/post assessment based on the COPE-NVI subscales score)", and "session" as predictors and a random intercept for each subject to account for repeated measures. In Table 11 are the descriptive statistics of the COPE-NVI subscales filled out by participants of all groups at T0 and T2.

For the COPE-NVI "Positive Attitude" subscale, a significant "Session" effect emerged $(F_{(1,105)}=37.17; p-value<.001)$ (Figure 25). As it is possible to observe in Figure 21, a general increase in the attitude of acceptance and restructuring beliefs, turning them into a more positive valence, is observed at T2 (T0: Mean= 30.35, Standard Deviation=5.09; T2: Mean= 32.63, Standard Deviation=4.80).

Figure 25. Plot about Session (T0/T2)*Group as a main effect.



Differences between groups emerged based on the significance of the "group" factor as a main effect ($F_{(2,105)}=3.64$; *p-value*<.05). The trend observed in the "Zoom & VR" group was to use more social support to deal with problems than what is revealed from the usual coping strategies of the other groups ("Zoom & VR" vs. "Zoom & GI": $\beta=3.39$; SE=1.42; $t_{(1,105)}=2.39$; *p-value*<.05; "Zoom & VR" vs. "Audio-track & VR": $\beta=3.24$; SE=1.42; $t_{(1,105)}=2.28$; *p-value*<.05). An interesting data emerged for the "Problem-Solving" subscale; indeed, individuals tended to use more active, focused to the problem, and planning strategies at the end of the entire relaxation protocol assessed at T2, rather than what participants reported during the T0 assessment (Baseline), ($F_{(1,105)}=5.84$; *p-value*<.05) (Figure 26).

Figure 26. Plot about Session (T0/T2) as a main effect.



Note: COPE Problem-Solving Session as a main effect results. The black asterisk represents *between-session* differences. *: *p-value*<.05; **; *p-value*<.01; ***: *p-value*<.001. Bars represent the Standard Error. T0= Assessment at Baseline; T2= Session 7.

Significant differences were found between "session" ($F_{(1,105)}=4.87$; *p-value*<.05) (Figure 27), and "group" ($F_{(2,105)}=20.32$; *p-value*<.001) as main effects for the COPE-NVI "Transcendental Orientation" subscale, showing that the "Audio-track & VR" group make less use of a transcendental approach, using religion than the other two groups (vs. "Zoom & VR": $\beta=4.21$; SE=0.80; $t_{(1,105)}=5.29$; *p-value*<.001; vs. "Zoom & GI": $\beta=4.56$; SE=0.80; $t_{(1,105)}=5.73$; *p-value*<.001). Moreover, independently of what kind of training participants experienced, a reduction from T0 to T2 in using religion to cope with problems has been revealed.

Figure 27. Plot about Session (T0/T2) as a main effect.



Note: COPE Turning to Religion session as a main effect results. The black asterisk represents *between-session* differences. *: *p-value*<.05. Bars represent the Standard Error. T0= Assessment at Baseline; T2= Session 7.

4.4.8 Satisfaction feedback by participants

No differences in terms of utility ($F_{(2,105)}=1.21$; *p-value*>.05) and positive feedback emerged from groups ($F_{(2,105)}=0.56$; *p-value*>.05). On a scale investigating the valence of the general experience considering the entire procedure (from 1=the experience was generally negative for me; to 5=the experience was generally positive for me), individuals expressed positive

feedback with a mean score of 4.44 (0.78). Moreover, on a scale from 1 (useless experience) to 5 (helpful experience), participants reported that the relaxation protocol is applicable based on a mean score of 4.35 (0.77). Interesting qualitative feedback, to be considered in the implementation of further studies, emerged from participants and are summarized in Table 11. Four main themes are derived from the qualitative feedback. They are, for the most part, about 1) the advantages of using VR, 2) the importance of adding more sessions and time to learn the relaxation techniques deeply, 3) Considerations of the opportunity to deploy the procedure in settings more similar to daily life, and 4) Suggestions and considerations to improve on the following studies.

Main themes from qualitative feedback	"Zoom & VR"	"Audio-track & VR"	"Zoom & GI"
Advantages of using VR	"I think it's an excellent implementation of relaxation techniques. With the headset, I tried a more intense and, therefore, more relaxing experience. I think I've reached a deeper relaxation in that condition"	"Having done all the training on the bed, changing in the last sessions was not immediate. In my opinion, the relaxation worked better when I was inside Oculus with the guide"	N/A
	"During the follow-up meeting, imagining the environment I had chosen in VR was very helpful to immerse myself in the situation and facilitate relaxation. Focusing on the most detailed re-enactment of the place had helped me free myself from intrusive and unnecessary thoughts"	"The session with Oculus was where I seemed to relax better" "I was excited about the VR experience. Choosing and customizing the environment to my liking was essential because I felt more comfortable and involved in the experiment."	
		"One evening, I couldn't sleep. I spontaneously thought of the VR image that had been proposed to me during the	

Table 11. Qualitative feedback by participants about the entire relaxation experience.

		penultimate meeting, which helped me to relax"	
More sessions and time	"More relaxation sessions to acquire better the PMR technique"	"More sessions would be useful, especially with virtual reality" "Maybe more meetings with Oculus could be planned. As for me, the one with the oculus was the most relaxing experience. It would also be useful to use the image better when you are not wearing the headset."	"Sessions I would have preferred to last longer, especially those mid-way, as it is the time when you start to acquire more awareness" "The training via Zoom should last at least two to three more sessions to familiarize yourself with the technique better and use it in a more proper way during meetings in the presence and also in everyday life" "I would like to have the opportunity to do more Zoom sessions"
Relaxation procedures in a more ecological environment	"I would love to try the virtual reality session at my home, on the bed. I think it would help even more"	"It would be interesting to use personal environments and better investigate the impact of the sense of familiarity on immersivity"	"I suggest allowing people to sit or lie down as they wish during lab sessions. This would also help to make the setting more ecological (in addition to facilitating relaxation)"
Suggestions and considerations to improve on the following studies	"For boring audio tracks, better to vary the contents. Beneficial immersion in the chosen virtual scenario" "I recommend providing a bed in the laboratory session because the lying position facilitates relaxation. The experience with Oculus was positive and immersive, but I don't think I had more benefits than the meetings on Zoom. Connecting on Zoom helped my willingness to	"As I had done the relaxation sessions on the bed, I found it difficult to apply relaxation to the lab chair"	"During the self- directed session, I used the therapist's suggestions in the Zoom training" "During the last session it was difficult to use the imagined scenario. Otherwise, I think the Zoom meetings helped me a lot in obtaining relaxation" "I appreciated both

p h n tu e	participate because it helped me relax. With those meetings, I have learned a technique that I try to use even in everyday life"		the Zoom and the GI experiences"
------------------------	--	--	-------------------------------------

4.5 Discussion

4.5.1 Principal Findings

The research on deploying customized VR scenarios for promoting relaxation is growing. Still, few definitive conclusions on its effectiveness have been achieved. Even less is known about the usefulness of integrating standardized and evidence-based relaxation techniques with new technologies that could provide an advantage in promoting autonomous adoption and self-empowerment by users and potential savings of treatment delivery costs (e.g., 215). As stated above, the possibility to customize audio and visual stimuli in the VR environment is a promising approach for meeting users' preferences and needs [e.g., 3], and it has shown encouraging results regarding both feasibility and potential impact on psychological well-being [e.g., 1].

The primary aim of this study was to evaluate the impact of alternative, complementary relaxation training composed of web-based PMR training and exposure to a customized, relaxing scenario in VR aimed to reduce anxiety and promote relaxation considering a sample of university students.

Before starting with the core results discussion, it is essential to note that participants in the VR groups reported no serious motion sickness effects. The most common symptoms were eye-related (e.g., discomfort or slight burning in the eyes once Oculus was removed). Even when experienced, symptoms were defined as not so impactful to ending the experience or not getting pleasure and involvement during the exposure to the virtual environment. As will be outlined in Chapter 5, our outcomes align with previous work that reported that static scenes and a constant speed during movement could prevent cybersickness symptoms (e.g., 312, 313). It is possible that our technological design, and the aim to promote relaxation, reduced the likelihood of occurrence of VR-related motion-sickness symptoms; indeed, the VR design and apparatus used in this study are labelled as static scenarios with only slight visual oscillations

(e.g., seawater and oscillation of plants due to the action of the wind), but without any rapid movement.

The first objective of our research was to investigate differences in state anxiety between and within groups. The former being when participants were directly exposed to a virtual or imaginative relaxation session (T1) and the latter using a self-directed relaxation session based on what they learned during the entire procedure (T2).

Although a reduction in state anxiety was observed in all groups at T1 and T2, our results highlighted how being in a customized virtual scenario promoted a greater reduction of the self-perceived state anxiety than in the imaginative condition, when individuals were directly exposed to the VR or GI experience (at T1), and when they have to recall the strategies learned during the relaxation protocol (T2). Indeed, when groups were asked to self-direct the relaxation session, which consisted of recalling the customized image they experienced in VR or GI at T1, those previously exposed to the customized VR scenario obtained lower state anxiety scores than those in the other group after Session 7 at T2. These results are in line with the literature and our first hypothesis since, independently of the type of web-based relaxation training administered before, VR is more effective than in-imagination exposure in allowing a subjective reduction of the state of tension and worry felt (measured based on the STAI-Y1) (e.g., 277-279, 292). Our outcomes are coherent with studies showing how VR could be a valuable tool in reducing stress and promoting relaxation, resulting from an assessment based on self-reported measures [314]. Moreover, although Hoag et al. study [315] target users' sample was composed of children and young adults with acute and chronic illness, our outcomes align with it, showing a significant decline in state anxiety from pre- to post-VR exposure. Our data support the role of VR in facilitating a powerful distraction effect on users reducing their focus on thoughts and external events that would elicit anxiety responses [198, 316]. Regarding the collected data on heart rate activity, findings showed a generally lower heart rate level during and after the relaxation sessions than before for all groups. This outcome is coherent with the self-reported data on state anxiety. But it is important to consider that this pattern may also be related to the fact that the baseline heart rate was much higher and potentially associated with the white coat effect due to the agitation experienced when a person knows that a professional is detecting their heart rate. Considering T1, data are coherent with what has been observed for self-reported anxiety. Indeed, a greater decrease in heart rate frequency was revealed for the two VR groups rather than for the GI group. Our data supporting previous studies' outcomes highlighted the impact of VR relaxing scenarios in maintaining lower levels of heart rate frequency (e.g., 314). Still, additional research is needed to deeply investigate whether and how naturalistic and relaxing VR scenarios induce relaxation and stress reduction by providing feedback on detected changes, such as in heart rate frequency and variability, respiration rate, or skin conductance.

Differently, at T2, no difference emerged between groups previously exposed to the Zoom web-based training, with an increase in heart rate from T1 to T2 for the "Zoom & VR" group. Notably, only the "Audio-track & VR" group obtained a decrease in heart rate levels at T2 comparable to those they achieved in the VR session at T1.

Results tell us that the experience in VR was more impacting on heart rate reduction maintenance than in the condition in which individuals had to recall the immersive image without the HMD support. It is possible that the fact of having learned relaxation based on the audio-track during all the sessions, without the presence of the experimenter, allowed them to generalize the relaxation learned in a more autonomous condition, without the experimenter's presence.

On the other hand, different patterns emerged for self-reported state anxiety. Indeed, once asked participants to report their subjective perception of state anxiety, the "Zoom & VR" group was the only one that, at T2, showed maintenance of the decrease in anxiety previously obtained at T1. Otherwise, the "Audio-track & VR" group had an increase in subjective anxiety levels comparing assessments at T1 and T2.

If we consider the overall performance during all the relaxation sessions, we can say that all groups have shown a reduction in state anxiety. Still, the group exposed to the audio-track condition obtained the most significant decrease during the training phase. Moreover, while the "Zoom & GI" group maintained the level of anxiety reached during the training sessions in the following sessions (T1 and T2), the two VR groups reached a further anxiety reduction compared to the state-anxiety scores in the training phases. The audio-track group has been shown to have achieved the most significant relaxation during training and exposure to VR. This group has most likely benefited from more factors: the fact of having carried out the sessions independently, in a family context without the presence, albeit remotely, of the therapist, and considering the possibility of being exposed to the audio track (which provided for the same modality of the training sessions) when they were also exposed to virtual reality. The fact that there has not been the maintenance of anxiety at T2 shows a possible difficulty of these participants in generalizing relaxation if the self-reported information about the anxiety level is considered. At the same time, the "Zoom & VR" group seems to generalize more based on the STAI-Y1 scores. This leads us to hypothesize that the therapist's presence during the sessions, and the booster effect of VR, seem to have helped more in the learning process of the

relaxation procedure. In addition, the environmental situation experienced by this group during the session in the presence was perceived as more like that expected by the therapist during the training phase. This may have affected the greater ability to generalize relaxation in the "Zoom & VR" group during the self-managed laboratory experience. It is possible that the group "Audio-track" perceived the session in the laboratory as more complex than during the training phase, in which the participants of the "Audio-track" group could choose more flexibly and autonomously the places and times to relax, and it is possible that this affected their perceived anxiety level. In general, discrepancies between subjective and psychophysiological measures emerged. This may be due to the fact that cognitive processes were more stressed during the training in the condition with the experimenter (Zoom & VR group). Otherwise, data show that the group trained based on the "Audio-track" gave more attention to the psychophysiological measures collection would provide us with more accurate information about the impact of these two web-based training interventions on anxiety.

The study's objective to understand the impact of VR on the sense of presence, realism, and engagement if compared with the standard guided imagery strategy, outcomes revealed that all groups experience a greater sense of being immersed in the scenarios, higher level of engagement, and sense of realism once they were in the VR or GI conditions than in the self-directed condition (T2). It is also revealed that the Guided Imagery condition allowed participants to experience a lower sense of presence and realism in the scenario than what individuals felt in the VR environment. This result is coherent with studies that showed how the properties of VR in facilitating interaction and converging people's attention allow individuals to experience a greater sense of presence than can be achieved in a context to be imagined, which also requires more cognitive resources from the person (e.g., 257).

Data also showed how virtual reality in the current study promoted a higher level of engagement in the relaxing scenario, especially for those who had completed the training phase in the audio-track condition. As stated before, it is possible that participants exposed to the audio-track condition were more familiar with the audio-track listened to during the VR exposure, and there is a chance that this facilitated further the engagement in the VR scenario. Indeed, it is plausible that the participants who received the entire training sessions guided by the therapist in the "Zoom & VR" group had more difficulty feeling involved as they were not confident with the audio instruction as the users who had done the whole training in the audio-track supported condition.

Our data highlighted how engagement with VR scenarios contributes to reducing perceived anxiety after the VR experience. Indeed, it seems that exposure to the VR scenario, instead of the imaginative one, facilitates the reduction of anxiety levels after the relaxation session. Even if our results need to be further investigated with larger user samples, they are aligned with previous studies, which underlined the potential benefit of using pleasant, relaxing, and immersive VR scenarios for facilitating relaxation and engagement in individuals from the general population [184, 233, 242]. Regarding the sense of presence, engagement, and sense of realism assessment, a point of strength of the current experimental design is the preliminary evaluation of the impact that the ability to control and vividly imagine mental images may have. Since we have not found differences between the two groups on these abilities before the exposure to the relaxation session at T1, we can support the hypothesis of a positive effect that exposure to a VR context can have on reducing anxiety.

The VR group's participants perceived the recalled scenario at T2 as more realistic than individuals in the GI group. Our data showed that the realism that characterized the imagined scenarios, derived from the immersive virtual scenario experienced in the previous T1 session, partially explains the state of anxiety levels felt after the T2 session by the participants in the VR condition. Moreover, participants in the VR group obtained additional benefits in terms of relaxation and state anxiety reduction related to a more realistic sensorial experience that played a substantial role in facilitating the visualization of the scenario and enabled users to focalize their attention on the relaxation activities. Coherently with other studies [184, 198, 242], this core outcome supports the role of immersive VR in promoting relaxation through visualization, engagement, and immersion processes. The prominent impact of being exposed to realistic scenarios in VR on the enhanced visualization at T2 is a key contribution of our study, shedding light on the impact that exposure to VR may have in more ecological everyday settings, such as when people are not wearing the Head-Mounted Display (HMD) but have the chance to transfer the relaxation skills learned in VR to real-world situations.

Our data are coherent with studies that show that a 3D representation of a scenario is more effective in the retention and recall processes than an imaginative representation [317, 318]. Even if the aim was to compare VR scenarios versus a scene on a PC desktop, our outcome aligns with Krokos et al. study [319], highlighting the prominent impact of VR scenarios on memory recall ability. The hypothesis of the current and the Krokos et al. studies [319] are anchored on classical studies in cognitive psychology based on the method of loci [320], and the context-dependent memory theory [321]. These theories imply the essential role of learning and mnemonic processes in creating an association between the mnemonic content and a

mental frame of scenarios and then recalling contents by mentally visualizing the scenarios in which the learning and memorization processes occurred [319]. Since that presence, immersion, and engagement in the VR scenarios implies sensorimotor contingencies similar to the real world [322], and the way we create and recall mental constructs is influenced by perception and action in the environment [323, 324], our data coherently confirm the possibility of immersive, virtual environments in enhancing learning and recalling for the intervention of vestibular and sensorimotor inputs [325].

An interesting result is also related to differences within and between groups on trait psychopathological constructs. In general, an increase in the global psychological well-being condition for all the groups has been evidenced, other than a reduction in trait anxiety, depression, and stress symptomatology once assessed at the end of the entire relaxation protocol (T2) than at T0. It was interesting to observe that the significant changes in anxiety were due to the groups exposed to VR since no differences from T0 and T2 were revealed for the group that had the relaxation experience in the GI condition. Specifically, our data showed how participants in the "Audio-track & VR" group obtained lower scores in trait anxiety than the other groups, with a greater difference if compared with the "Zoom & GI" group. It emerged that the audio-track relaxation training, with the booster contribution of VR, significantly reduced anxiety over time. Future studies should investigate this aspect more deeply, introducing an experimental design composed of more than one exposure in VR. Indeed, in this case, we are talking of a more stable manifestation of anxiety, and it is possible that other uncontrolled variables contributed to the final outcome. In any case, these results are promising and support the claim that VR plays a key role in amplifying the effectiveness of already validated interventions, maintaining their effects over time.

Based on the COPE-NVI subscales, the coping strategies, too, seemed to be modified during the relaxation training protocol. Independently on the type of relaxation strategy experienced, participants have started to use a more strategy focused on accepting and restructuring their beliefs when facing stressful or anxious situations.

Moreover, at the end of the entire relaxation protocol, individuals tended to use more strategies to focus on the resolution of the problem and reduced the habits of using religion to cope with problems.

4.5.2 Strengths, limitations, and future perspectives

The current study is the first to examine whether customized virtual reality scenarios work better than guided imagery in improving relaxation and decreasing anxiety when integrated with PMR as a complementary relaxing method. Moreover, this contribution adds new knowledge on the importance of customizing and customizing digital interventions according to the user's perspective, needs, and preferences [3]. Another point of strength of the current experimental design is the assessment of the impact that the ability to control and vividly imagine mental images may have. The outcomes are gathered from a well-designed pilot randomized controlled trial involving three selected groups of 108 university students whose socio-demographic and psychological characteristics have been controlled to balance their effect on the investigated variables. The experimental procedure adopted supports the reliability and validity of the results and the conclusions presented.

Our study also has some limitations that could be overcome by future studies. Our results cannot be generalized to the whole non-clinical population, as our sample consisted of university students carried out in a single university. To further generalize and validate our results' effectiveness, the involvement of participants belonging to clinical and non-clinical populations, stratified according to different socio-demographic characteristics, should be considered [262]. The following studies should consider structuring the relaxation protocol with more VR sessions to better understand the impact of a virtual scenario on relaxation. It should also be useful to introduce other follow-up sessions to investigate the impact of a webbased intervention with VR over time. Another aspect to be considered in further experimental design is introducing a control group that received the training session with the physical presence of the therapist. Given the promising results in studies characterized by partially different experimental designs, it would be helpful to introduce the structuring of virtual webbased contexts in which an avatar guides the learning of relaxation and visually shows how to perform PMR exercises remotely without the therapist. This may potentially facilitate the learning of the training as well as the transfer of the learned techniques in other more ecological contexts where it is not planned to wear the headset (e.g., before an exam). Considering the impact of customization on anxiety and for engaging users in the virtual scenarios, another important aspect that also needs to be considered is the opportunity to introduce in the virtual environment customized stimuli based on the personal life experience of each participant. As an example, introducing olfactory stimuli could be another essential aspect to enhance the sense of realism, immersiveness and presence in the virtual scenarios (e.g., 326-328).

In our study, the prominent data are derived from self-reported questionnaires. Considering the importance of psychophysiological assessment in evaluating tension and other anxiety and arousal measures (e.g., 329), future studies should include in the assessment procedure other more objective measures such as a photoplethysmography (PPG) to record the pulse rate

changes comparing data collected before and after the relaxation sessions (Lee et al., 2020: Leong et al., 2022), and electroencephalogram (EEG) to assess the alpha and theta waves related with anxiety [330].

Finally, it is important to consider that seven participants in the "Audio-track & VR" condition dropped the intervention during the training sessions. This informed us about the importance of introducing, in future studies, rewarding strategies to maintain high levels of motivation of participants, especially in individuals that have to administer the training in a more autonomous manner. In future studies, it should also be interesting to obtain more information about the dispositional and personality traits of participants who drop out of the training to predispose the most optimal condition to promote compliance with the interventions.

To conclude, the main objective of psychological interventions, in general, is to offer people the possibility to learn strategies to manage their daily life more constructively and in a selfdirected way. The opportunity to deploy VR for promoting self-management of state anxiety also in real-world situations constitutes an interesting line of investigation to be further explored by future studies involving larger non-clinical and clinical populations (e.g., chronic pain patients) to validate and standardize relaxation protocols integrated with new VR tools.

CHAPTER 5.

EXPLORING THE ROLE OF CUSTOMIZED VIRTUAL REALITY SCENARIOS FOR RELAXATION AND ANXIETY MANAGEMENT IN PATIENTS WITH MEDICAL CONDITIONS

The present chapter provides an overview of the third study of the Ph.D. program. The current version is under review by the *Nature - Scientific Reports* (Collection "Virtual Reality in Psychological Research"). The current research is registered as a clinical trial in a publically accessible primary register that participates in the WHO International Clinical Trial Registry Platform (National Institute of Health (NIH) U.S. National Library of Medicine, ClinicalTrials.gov NCT05863065; 17/05/2023).

The research has been conducted in collaboration with Prof. Lora Appel (York University, Toronto, Canada; University Heath Network, Toronto, Canada).

Being immersed in a natural context has a beneficial and pervasive impact on well-being, it is strongly related to the self-perceived and the physiological reduction of stress, relaxation, and the sense of restoration, and it is helpful in regulating mood [331]. In long-term care settings, allowing older adults to enjoy recreational experiences in natural environments can be beneficial, especially if they have cognitive and physical impairments. Despite this, it is essential to reflect on how obstacles (e.g., reduced motor skills and fear of injury during movement) may hinder and limit participation in organized outdoor activities. Virtual Reality (VR) can help expose people to naturalistic scenarios virtually, overcoming obstacles that prevent them from visiting real natural environments. VR could also increase engagement and relaxation in older adults with and without cognitive impairment. Growing evidence supports the claim that immersive virtual reality scenarios: 1) provide innovative strategies for overcoming these obstacles, 2) improve the quality of life in older adults' healthcare facilities, 3) have a positive impact on behavioral and psychological symptoms, and 4) promote engagement and relaxation. Considering the promising outcomes of recent empirical investigations and the need for more depth and breadth studies to have VR interventions adopted by mainstream healthcare, the current presentation aims to describe the results of a Proof-of-Concept and Mixed-Methods study. The research is mainly focused on investigating the feasibility of a customized naturalistic VR scenario by assessing motion-sickness effects, engagement, pleasantness, and emotions related to exposure to a VR context of 23 individuals living in a long-term care home diagnosed with cognitive impairment. The measures administered were self-reported and observational to obtain information from users and seven

healthcare staff professionals who attended the VR sessions. At the end of the experimental phase, a focus group session was conducted with the healthcare staff to acquire additional information, mainly on strengths, weaknesses, future perspectives, and risks that are associated with using VR in a long-term care setting.

Feasibility and acceptability were proved to be satisfactory since users' VR experience was well-tolerated, and no adverse side effects were reported or observed. Moreover, preliminary outcomes showed that more than half of the participants were engaged, pleased, and calm most of the time during the virtual activity, showing decreased state anxiety during and after the experience. The current study's key leading innovation is its contribution to advancing knowledge on the role of customized VR scenarios and their impact on the acceptability and potential clinical efficacy of their usage with frail older adults.

STUDY 3. CUSTOMIZED VIRTUAL REALITY NATURALISTIC SCENARIOS PROMOTING ENGAGEMENT AND RELAXATION IN PATIENTS WITH COGNITIVE IMPAIRMENT: A PROOF-OF-CONCEPT MIXED-METHODS STUDY

5.1 Theoretical background

There is growing evidence of the positive impact of being immersed in natural environments on physical and mental health [27-29]. Natural contexts increase the perception of positive emotions, reduce fear, anxiety, anger, and sadness, and play an essential role in regulating heart rate, blood pressure, muscle tension, and brain electrical activity [87].

In healthcare facilities, allowing older adults to have recreational and pleasant experiences in natural environments, especially if they have a cognitive and physical impairment, although challenging, can be beneficial [30]. Participating in organized outdoor activities may be limited by reduced motor ability, cognitive impairment of the patients, the fear of injury during movement, as a side effect of medication, and weather conditions. In addition, outdoor spaces could not always be available. Overall, evidence shows that even if unable to engage directly in a natural environment, visible exposure to nature scenes can be beneficial, reducing hospital length-of-stay and drug consumption [30-33]. Based on these findings, it is desirable to identify alternative strategies for providing older adults with exposure to natural scenes while accounting for barriers to mobility, autonomy, and safety concerns.

A potential innovative method helpful in overcoming these obstacles could be the deployment of Virtual Reality (VR). Indeed, VR technologies have been demonstrated to reduce isolation

and increase engagement across several populations [34]. VR systems consist of technologies that provide the user with multiple sensory inputs through, for example, visual and auditory displays and can have varying degrees of immersion generally conveyed through displays, where the interface allows exposure to content that can be explored in 360-degrees, as with head-mounted displays (HMD).

VR-based interventions have been successfully deployed for addressing a myriad of clinical conditions such as specific phobia, social anxiety, panic disorder, post-traumatic stress disorder, and depressive disorders [332, 333].

Evidence suggests that immersive virtual scenarios can improve the quality of life in older adults' healthcare facilities, highlighting a positive impact on behavioral and psychological symptoms. VR was also found to overcome some barriers that can hinder participation in recreational activities, such as the difficulty of moving in the environment [37, 219, 334]. A study by Brimelow et al. [219] demonstrated how immersive scenarios can positively impact users with heterogeneous cognitive impairment levels living in a long-term care home with minimal negative effects.

Patients with cognitive decline may benefit from improving cognitive, emotional, and physical fitness functions if exposed to multisensory VR contexts characterized by quasi-naturalistic and realistic stimuli [335-339]. To our knowledge, few studies have investigated the feasibility and effectiveness of using immersive virtual environments for managing anxiety and promoting a state of relaxation in populations of elderly individuals with impaired physical and cognitive functions [219, 336, 340, 341]. Moreover, there is a lack of research on customizing virtual scenarios for therapeutic purposes.

Some published research has highlighted how using VR HMD promotes the management of depressive and anxious symptoms in older people [3, 221]. Appel et al. [36] investigated the relationship between relaxing, positive environments and both positive and negative emotions based on a sample of older adults with varying (mild to severe) cognitive and physical impairment levels. The participants were exposed through an HMD for an average variable period of 8 minutes to a relaxing environment in which realistic natural scenarios were depicted. Qualitatively, positive feedback is associated with the experience and the perception of a higher state of relaxation. Additionally, there has been an increase in the intensity and frequency of positive emotions experienced and a reduction in the levels of some negative emotions [36]. The authors concluded that exposure of people with cognitive and physical impairments to realistic and natural immersive scenarios in virtual reality through an HMD is

safe and feasible. These findings encourage future studies to investigate the role of virtual reality scenarios' customization.

Additional research by Moyle et al. [221] exposed individuals with dementia living in care facilities to a visual and auditory interactive virtual experience with the aim of improving the quality of life by obtaining information about users' moods, apathy, and engagement. The exposure involved observing a screen, without using an HMD, in which a natural scenario was represented. During the exposure, participants were able to select seasons (spring and autumn) and different types of virtual objects through hand and arm movements. Residents, family members, and staff have received positive feedback globally. Participants experienced higher levels of engagement, pleasure, activation, and fear during the exposure to the virtual forest than the control sample. Of note, the authors stated that differences in users' preferences should be considered as a main factor limiting the versatility of the VR Forest.

Research on the effect of customizing VR-scenarios has grown in recent years [3, 233], uncovering benefits such as an increased sense of presence and engagement in the virtual environment. A relaxing and customizable VR environment implies an *a priori* identification of conditions that can be pleasant and relaxing based on the user's needs, allowing the management of any interfering environmental factors that might also be present in the real context [3, 5, 228]. This person-centered approach is precious for those living with cognitive impairment as it also provides a sense of familiarity and security within the VR context [233], but additional investigations are still needed to obtain more consistent data on its feasibility and effectiveness in this population.

5.2 Aims

Considering the promising outcomes of recent studies (e.g., 36, 37, 221) and the recognition that more depth (through rigorous research), and more breadth (encompassing new elements of design, e.g., customization) are needed for VR-interventions to be adopted by mainstream healthcare, we designed a proof-of-concept study, with the following goals:

- The primary objective is to evaluate the impact of VR on self-reported and observational levels of motion-sickness, engagement, and pleasantness in older adults living with cognitive impairment and residing in long term care.
- Based on our knowledge, distinct from most studies in this field, the current study intended to provide customization of visual and audio stimuli over and above, offering a select number of outdoor settings. For this reason, the secondary aim is to investigate if personalized, relaxing virtual environments can be acceptable to
the target group, as well as positively impact feelings and state anxiety, at least in the framework of a proof-of-concept study. Indeed, the current study is part of a continuum of research on the requirements and effects of VR customization for different target populations. The study leveraged on the positive results of VR customization collected in a previous study we conducted [221] involving non clinical population and it intended to further explore the feasibility of different customization options.

- A third goal is to investigate the usability of the VR apparatus from the perspective of health staff working in long-term care to better understand the acceptance and potential adoption of the proposed solution in these clinical settings.

5.3 Methodology

The current proof-of-concept and feasibility study is a one-session single-centre trial conducted at the *Azienda Pubblica di Servizi alla Persona (APSP) "Margherita Grazioli*", a long-term care home in Trento (Italy) in collaboration with the Department of General Psychology – University of Padova (Italy) and the Centre for Health and Wellbeing-Fondazione Bruno Kessler (Italy). The proposed study protocol was approved by the Ethical Committee for Clinical Experimentation - *Azienda Provinciale per i Servizi Sanitari - Provincia Autonoma di Trento* (Italy), (Nr: A827, 10/2022, Rep. Int. 20298 11/22/2022). The research complies with the relevant ethical regulations of the Declaration of Helsinki.

The study design is based on a mixed-methods approach inspired by the Obesity-Related Behavioral Intervention Trials (ORBIT) framework [241] for the design (Phase Ib) of digital interventions and their preliminary testing (Phase IIa). At the current stage, a sample is usually selected from accessible individuals to understand if the intervention deserves an increased depth of analysis, improvement, and future testing.

5.3.1 Participants

An APSP healthcare professional (*Educatrice socio-assistenziale*) identified individuals eligible for the study based on an ad hoc schedule and obtained informed consent. Written informed consent was collected from all participants or their Substitute Decision Makers (SDMs). In addition, authorization by the primary care physician was also requested. Patients over 18, Italian mother tongue, and with mild, moderate, or severe cognitive impairment were

eligible for participation. Exclusion criteria were: 1) palliative care; 2) diagnosis of psychosis; 3) severe neurological damage; 4) a positive diagnosis of epilepsy or having first-degree relatives diagnosed with epilepsy; 5) cardiac pacemaker or other metal devices; 6) infectious or gastrointestinal disorders; 7) the presence of open wounds at the level of the face; 8) motor or visual dysfunctions and neuromuscular pain that prevent the use of Oculus. Participants did not receive remuneration upon completion of the study. Overall, 47 users were considered for participation in the study. Of these, 24/47 did not participate for different reasons: 1) 17/47 did not meet all the criteria of inclusion/exclusion; 2) 3/47 did not agree to make the preliminary assessment phase; 3) 4/47 did not want to participate because they did not want to try a new experience that needed to be further studied to better understand the effects. The final sample was composed of 23/47 users, one of whom was not exposed to Oculus for fear of wearing the HMD (see Figure 1).



CONSORT 2010 Flow Diagram



5.3.2 Intervention

5.3.2.1 Hardware and software equipment

A commercially available VR HMD (Oculus Quest 2, Meta Quest) [342], Alienware m15 Ryzen Edition R5" workstation, and a link cable were used. The virtual scenarios were developed in the Unity framework by using the C# programming language (source of assets: https://freesound.org/; Unity; Polyhaven; HDRIhaven). The code was versioned via GitLab. The protocol for the virtual environment design in the current study has been outlined by

Pardini et al. [233]. The individual sessions were video-recorded to allow for triangulation, including nonverbal behavior. Most of the experimental sessions were conducted in the dedicated music-therapy room at the APSP. All of the sessions were administered and evaluated by the same assessor. Health workers accompanied and passively assisted the procedure as observers to get information on the feasibility and usefulness of the procedure and to inform subsequent studies and system design with their requirements. Observations from health professionals were obtained based on self-reported questionnaires and by a Focus Group session. To manage problems related to physical movements and fear of going into an unfamiliar context, two individual sessions were administered in a more commonplace setting, and for one participant, the procedure was administered in their private room. Each session lasted approximately 50 minutes. For safety reasons, VR was always administered when participants were seated safely in a chair or wheelchair. To allow individuals to explore the virtual context, the experimenter and the health care professional who assisted the patient during the procedure helped the participants to rotate their heads and move their chair or wheelchair.

Figure 2. The PC's interface with icons related to the different customizable variables (e.g., music, wind, weather conditions, time of day, and presence of people) [233].



A brief baseline assessment (T0) investigating the ability to move head and body, the use of visual and auditory aids, and the general emotional state were conducted based on an ad hoc measure and a modified version of the *State-Trait Anxiety Inventory- Y1* (STAI-Y1) inspired to the version developed by Appel et al. [36, 37]. At least 2 minutes of VR exposure were conducted before the trial to familiarize patients with the virtual context. Users were invited to choose one of three realistic VR scenarios: a mountain, marine, or countryside environment.

With technical operator support, participants could select and customize all the audio and video stimuli (see Figure 2). A specific range of customization options was chosen as appropriate for the target group of the study, also relying on results of previous studies [40]. Participants had the possibility to visualise the different options and to ask the researchers to modify specific characteristics of the environment. The technical operator and the experimenter could customize virtual scenarios based on a PC's dashboard interface connected to Oculus Quest 2 via a cable. Specifically, the interface presented a series of icons related to the different customizable variables (Figure 2) such as music, wind, weather conditions, time of day, and presence of people). The connection between the PC and the viewer allowed the operator to see what the participant was observing during the experience. Participants had the possibility to visualise the different options and to ask the researchers to modify specific characteristics of the environment. The technical operator and the experimenter could customize virtual scenarios based on a PC's dashboard interface connected to Oculus Quest 2 via a cable. Specifically, the interface presented a series of icons related to the different customizable variables (Figure 2) (such as music, wind, weather conditions, time of day, and presence of people). The connection between the PC and the viewer allowed the operator to monitor in real time what the participant observed during the experience. At any time during the session, participants had the possibility to visualise the different options and to ask the researchers to modify specific characteristics of the environment. Then, the participant was told they could remove the HMD when desired. Throughout the VR session, the experimenter took notes of the participants' verbal and non-verbal communication based on the Observed Emotion Rating Scale (OERS), and an Ad Hoc Observation Form. Post the intervention phase (T2), the following measures were collected: emotional state, motion-sickness symptoms, usability, engagement, and pleasantness experienced during the activity. Considering the difficulties participants with cognitive impairment could have in peculiar situations, and moments of the day (e.g., negative symptoms relating to the sundowning syndrome), each VR session was arranged based on the participants' and health staff's needs and preferences-Figures 3a and 3b present pictures of two users during their VR experience.

At the end of the entire experimental phase an assessment based on self-report questionnaires of all participants was made and a focus group was conducted with the health staff professionals that observed and assisted the older adult participants during the VR experience.

Figure 3. A participant during the experimental phase with Virtual Reality. Written, informed consent was obtained from the individual or their SDMs to publish these images. Fondazione Bruno Kessler and TrentinoSalute4.0 have the copyright, and permission must be obtained for use in other sources.



5.3.3 Outcome measures

Data collection. Measures are administered in the form of interviews or as observation administered before, during, and after the VR experience. Questionnaires were administered verbally, and participants were asked to respond when they could. Information about the participant's current mental and physical conditions were extrapolated from their medical record.

A *Socio-demographic questionnaire* was administered by the health care professional to obtain information about gender, age, marital status, past occupation, degree level, proneness to motion sickness symptoms, hearing and vision impairments, mobility limitations or use of physical aids, previous experience with VR, and years of stay at the APSP facility. Moreover, the health care professional obtained information about the cognitive impairment level and psychological symptoms from the medical record (e.g., *Mini-Mental State Examination score-*MMSE [343, 344], and the *U.C.L.A. Neuropsychiatric Inventory-NPI* [345]). The inclusion and exclusion criteria evaluation were performed with the demographic schedule.

The modified version of the *State-Trait Anxiety Inventory-Y1 (STAI-Y1)*, inspired by Appel et al. [36, 37], was administered in this study to obtain information on the state-anxiety level and other types of emotions individuals experienced before and after the VR session. The current study specifically investigates relaxation, worry/anxiety, sadness, tightness, annoyance, and

anger. The scale was administered as an interview based on a 5-point Likert scale from 1 (not at all) to 5 (a lot). Cronbach's α displayed an adequate consistency for the total score recorded at T0 (Cronbach's $\alpha = 0.75$) and at T1 (Cronbach's $\alpha = 0.71$).

The experimenter used the *Observed Emotion Rating Scale (OERS)* during the VR session. It was adapted from the original version of Lawton et al. [346] and used as an observation tool to assess the presence and frequency of negative emotions (fear, anxiety, anger, and sadness) and positive (pleasure) feelings experienced during the session based on a scale from 1 to 5 (1: "undetected emotion"; 2: "emotion observed for less than 16 seconds"; 3: "emotion observed for 16-59 seconds"; 4: "emotion observed for 1-5 minutes"; 5: "emotion observed for more than 5 minutes").

During the VR session, the *Ad Hoc Observation Form* was also completed to record the type and frequency of vocalizations, phrases, facial expressions, body movements, reminiscences, and pleasant memories elicited during the VR exposure.

The *Virtual Reality Symptom Questionnaire (VRSQ)* [249, 292] assesses the general and eyerelated physical symptoms of exposure to a virtual reality environment. The score assigned to each item ranges from 0 to 6, with a maximum total score of 84 (48 for general symptoms and 36 for eye symptoms). Higher scores represent worse symptoms, with 0 corresponding to no adverse effects, and 84 to serious adverse effects.

The VR experience tolerability, acceptance and wearability was primarily estimated based on the frequency of time spent in the VR context, on positive and negative emotion expressed (such as boredom, joy) and the weight of the HMD. Finally, the usability, engagement, pleasantness, and satisfaction experienced in using the VR apparatus were inspired by a measure developed and described by Appel et al. [36, 37], composed of both self-reported questions and other queries that the experimenter answered by observing the participant's behaviors during the experience and characterized by a series of items based on a 5 points Likert scale (Questions about Level of interest, awareness, engagement, and enjoyment observed: 1="very much", 5="not at all"; questions on other information in relation to the VR experience: 1= "strongly disagree", 5= "strongly agree"), and six open-ended questions focalized in obtaining additional information, where possible, about: 1) what participants liked best and least; 2) what participants would like to see; 3) if participants would like to repeat the experience; 4) if participants would recommend the experience to a friend. To obtain information from health professionals working in the APSP about usability and the perceived quality rating of the VR set-up deployed, the *Adapting-Mobile App Rating Scale (A-MARS)-Subjective Quality Scale* [347], and the *System Usability Scale (SUS)* [348-350] were filled in by operators who participated during the VR sessions. Questionnaires' administration was performed after the administration of the experimental procedure. At the end of the experimental phase, a focus group was conducted with the health professionals that assisted users during the VR experience. Issues discussed during the focus group were the strengths and weaknesses associated with using virtual reality, the future perspectives, and the risks associated with using virtual reality with users affected by cognitive impairment.

5.3.4 Sample size estimation

Sample size calculations were performed to have more rigorous information for interpreting in a more conservative way quantitative data. This permits to estimate the sample size needed to detect a significant difference based on the administration of the STAI. An a priori power analysis has been executed determining an expected medium standardized effect size (effect size = .65), with a power of 80%. Considering a non-normal distribution of the STAI, sample size was calculated with G-Power for non-parametric tests (Kruskal-Wallis Test, Wilcoxon test). Results suggest that a sample of at least 22 patients have to be enrolled. Sample size and Power for non-parametric tests (Kruskal-Wallis Test, Wilcoxon test) were calculated with G-Power. Results suggest that, with a non-normal distribution, 22 patients have to be enrolled. If parametric statistics are conducted (assuming the normal distribution), the sample should instead consist of n = 21 users with Bonferroni correction (*alpha* = 0.05), an effect size equal to .65, and a power of 80%. Considering the current sample was not normally distributed based on the STAI, a sample of at least 22 patients is enrolled.

5.3.5 Data analysis

Continuous data are presented as mean \pm standard deviation. Categorical data are presented as numbers with percentages. The statements made by participants during the VR experience were analyzed by SP and SO using thematic analysis and reported as frequencies. Data were analyzed thematically following an inductive, data-driven approach [351]. Data codes were generated systematically, collated into themes, and applied to the data set to generate frequencies. For a comparison of non-normally distributed continuous variables before and

after the VR intervention, the Wilcoxon signed-rank test is used for each intervention separately. A two-sided p-value <.05 is considered significant. Statistical analyses were undertaken using SPSS® version 29 [293]. Moreover, qualitative responses were transcribed before, during, and after the VR exposure. In order to analyze data obtained by the audio-recorded focus group, the micro interlocutor analysis method has been applied [352]. The present method is helpful in obtaining information on participants' attitudes, points of view on the use of VR, and permits to have quantitative data on participant grouping. Data were analyzed based on descriptive statistics.

5.4 Results

5.4.1 Demographic analysis

Twenty-three older adults were recruited from the APSP "Margherita Grazioli". Demographics and clinical features of the study sample are described in Tables 1 and 2. It is important to specify that twenty individuals (87%) could move their heads without difficulty, and only 5 (34.8%) were able to move autonomously in the environment, but with supervision. Eleven participants were in a wheelchair during the VR experience (47.8%), and twelve (52.2%) used the chair provided by researchers.

Demographic features	Sample Group N=23
Gender	N (%)
Women	19 (82.6%)
Age	Mean (SD)
	86.6 (5.12)
Marital Status	N (%)
Single	1 (4.3%)
Married	5 (21.7%)
Divorced	2 (8.4%)
Widowed	15 (65.2%)

 Table 1. Demographic features of the entire sample.

Education	N (%)
Primary school	13 (56.5%)
Middle school	3 (13%)
High school	3 (13%)
Employability training certificate	4 (17.4%)
Past employment	N (%)
Housekeeper	11 (47.8%)
Worker	7 (30.4%)
Clerical worker	3 (13%)
Teacher	2 (8.6%)
Aids	N (%)
Wearing glasses	12 (52.2%)
Hearing problems	6 (26.1%)
Wheelchair user	11 (47.8%)
Walking aid	6 (26.1%)
Multifunctional highchair	1 (4.3%)
Walking stick	1 (4.3%)
No aids	1 (4.3%)
Dizziness experienced during	N (%)
movement of transportation	N
Yes	None
Sickness experienced while traveling	N (%)
Yes	0
Sometimes	6 (26.1%)
No	17 (73.9%)
Feel involved in watching television	N (%)
Yes	13 (56.5%)
A little	5 (21.7%)
No	5 (21.7%)
Aware of what VR is	N (%)
Yes	2 (8.7%)
Yes No	2 (8.7%) 21 (91.3%)
Yes No Past experiences with VR	2 (8.7%) 21 (91.3%)

Favourite actual pleasant activities	N (%)	
Listen to music	2 (8.7%)	
Reading	2 (8.7%)	
Walking	6 (26.1%)	
Dancing	1 (4.3%)	
Talking with friends	2 (8.7%)	
Painting	1 (4.3%)	
Playing card games	1 (4.3%)	
Doing crossword puzzles	1 (4.3%)	
Watching TV	6 (26.1%)	
Outdoor activities	1 (4.3%)	
Activities to cope with anxiety (in the	N (%)	
past)	1 (4.3%)	
Painting	14 (60.9%)	
Outdoor activities	2 (8.7%)	
Knitting	1 (4.3%)	
Reading	1 (4.3%)	
Listen to music	2 (8.7%)	
Spending time with family and friends	1 (4.3%)	
Singing	1 (4.3%)	
To pray		

Table 2. Clinical features of the entire sample (N=23).

Mini-Mental State Examination (MMSE)

Cognitive Impairment Levels (range)	N (%)
Mild (18-26) (n, %)	11 (47.8%)
Moderate (10-17) (n, %)	9 (39.2%)
Severe (<10) (n, %)	3 (13%)

Neuropsychiatric Inventory (NPI)

Clinical features	N (%)	(Frequency x Severity) Mean (SD)
Delusions	0	0

Hallucinations	0	0
Agitation/Aggression	2 (8.7%)	0.43 (1.47)
Depression/Dysphoria	7 (30.4%)	1.52 (3.12)
Anxiety	9 (39.2%)	2.57 (4.09)
Elation/Euphoria	0	0
Apathy/Indifference	4 (17.4%)	1.22 (3.45)
Disinhibition	3 (13%)	0.91 (2.94)
Irritability/Lability	7 (30.4%)	1.30 (2.98)
Aberrant motor behavior	5 (21.7%)	1.61 (3.81)
Sleep and Nighttime Behavior Disorders	5 (21.7%)	1.30 (3.11)
Appetite and Eating Disorders	2 (8.7%)	0.87 (2.94)

In order to control the potential relationship between the presence of neuropsychiatric symptoms and the cognitive impairment level, Kendall's tau rank and Spearman's rho correlations between the MMSE and the NPI total score were executed considering the entire sample. No relationship between neuropsychiatric symptoms and the cognitive impairment level emerged (Kendall's tau rank = -0.23; p-value > .05; Spearman's rho = -0.29; p-value > .05).

5.4.2 1[^] Objective: Investigation of tolerability, motion-sickness effects, engagement, and pleasantness associated with the VR experience.

VR tolerability was operationalized as the frequency of time spent in the VR context that is also related to the emotions felt during the experience and the wearability of the HMD. On average, participants spent 9.91 minutes exposed to the VR scenario (SD=4.78; minimum =1 minute; maximum = 17 minutes). Fifteen users (68.19%) spent at least 9 minutes of time in the VR context (SD=4.78; minimum =1 minute; maximum = 17 minutes). None of the users referred to experiencing a general condition of discomfort, drowsiness, headache, sweating,

claustrophobia or disorientation, and nausea during the VR experience. Three participants 3/22 (13.1%) felt a slight fatigue state, 1/22 (4.3%) experienced boredom, 10/22 (43.5%) reported that their eyes were slightly tired and irritated during the VR exposure, 2/22 (8.7%) experienced a slightly blurred vision, and 3/22 (13%) reported having had slight difficulty focusing. Ten (43.5%) participants experienced discomfort due to wearing the HMD (two participants said: *"these special glasses are heavy"*), and one participant completed only the T0 assessment but decided not to try the VR since he was worried about wearing the HMD.

Table 3 summarizes the responses assessed on a 5-point Likert scale, from 1 (very much) to 5 (not at all), related to the interest, awareness, engagement, and enjoyment shown during the VR experience highlighting how the VR activity, on average, was appreciated by participants.

Interest and awareness	Mean (SD)	
General interest in the visual and auditory stimuli of the VR scenario	1.85 (0.75)	
Awareness during the experience (based on posture and facial expression)	1.85 (0.67)	
Engagement	Mean (SD)	
Engagement (based on verbal feedback)	1.95 (0.83)	
Reminiscences	2.20 (1.20)	
Enjoyment	Mean (SD)	
Enjoyment (based on posture and facial expression)	1.20 (0.41)	

Table 3. Level of interest, awareness, engagement, and enjoyment observed.

Some of the verbal feedback (direct quotes) about engagement in VR were: "beautiful", "I would like to add other flowers", "I am feeling good here", and "It is marvellous". Two users

said, "the experience was good, but it should be useful for younger people", and "not bad (...), but I prefer to do other things".

During and after the experience, 11/22 participants (50%) reported that the VR scenario triggered memories by saying, for example: "When I was younger, I went to Sicily with my babies", "It is different (but) I am thinking about the picnic at the mountain with friends". Moreover, post-intervention, a participant said that the experience allowed her not to think about negative things for a while. In general, 16/22 users (72.7%) expressed their desire to be exposed to VR scenarios in the future and would recommend the experience to their loved ones and friends.

Overall, seven main themes were identified based on what participants verbally shared during the experience (see Table 4): (1) reminiscence (11/22 participants; 47.8%); (2) aesthetic appreciation of the virtual context (8/22 participants; 34.7%); (3) realism (6/22 participants; 26.1%); (4) sense of safeness and protection in the virtual environment (4/22 participants; 17.4%); (5) suggestions for improving the VR scenarios (2/22 participants; 8.4%); (6) lack of interest (2/22 participants; 8.4%); (7) appreciation of the new experience (1/22 participants; 4.3%).

Table 4. Themes and quotes by participants.

Main Theme	User Quotes
"Reminiscence"	 "() there is also my mom, she is old now () we went together to the see ()" (Participant 2). "I was with my dad at the sea - I went to the sea" (Participant 2). "() my mountain ()" (Participant 3). "I walked a lot in my mountains" (Participant 4). "We went there with the children" (Participant 6). "We used to play on a field like this" (Participant 6). "When I could walk in the mountains () melancholy" (Participant 5) (then she cocked up by remembering what she is no longer able to do). "I went to the sea in Jesolo, good memories. Now the children go alone" (Participant 9). "() in Sardinia, this sea reminds me of my happy time" (Participant 10). "I took care of my garden" (Participant 20). "How I liked to walk in the mountains" (Participant 14). "() I remember lots of things ()" (Participant 22).
"Aesthetic appreciation of the virtual context"	"We went to the mountains with blankets () and everyone sang the mountain songs" (Participant 16). " () beautiful here" (Participant 3).

	 "Suddenly a beautiful snowfall () the trees are beautiful () there are little animals and butterflies that go around () is beautiful () beautiful trees and the river that moves split in two - the effect is more positive than negative" (Participant 4). "Beautiful" (Participant 12). "Beautiful place" (Participant 13). "Wow() how beautiful () what beautiful streams" (Participant 14). "Beautiful, beautiful () that wonder" (Participant 15). "Beautiful, the sea is immense, what a wonder, I see clouds" (Participant 22). "Beautiful that stuff () how did you do, beautiful, but is there such a place?" (Participant 22).
"Realism"	 "There's a wad there that's gone because I touched it" (Participant 7). "Touch the water of the sea (laughs) () oh maria looks at the beautiful trees () just bathe the water" (Participant 10). "I can't see well what it looks like, it seems cloudy but it is clear they are not" (Participant 18). "They seem like the <i>Cascata delle Marmore</i>" (Participant 15). "Does not increase the wind that takes me away" (Participant 15). "() seem like scarecrows that make blue flowers" (Participant 15). "Now comes the water, () it makes me feel good - but is it true? - look how much water that comes" (Participant 16). "Some things don't seem true () plant shadows don't move ()" (Participant 13).
"Sense of safeness and protection in the virtual environment"	 "I feel calm here" (Participant 2). "I feel good" (Participant 5). "I feel safe" (Participant 8). "The sea is the place that makes me feel good and safe even in reality" (Participant 16).
"Lack of interest"	"I'd rather go back to my room () I saw that birds pass by and that's it, I like all the tools (annoyed), I'm fine everything it's nice but nothing more, I want to go back to my room" (Participant 1). "I don't care so much today, I'm old, a little heavy (referred to the HMD)" (Participant 19).
"Suggestions"	"Beautiful, but I would like more people here" (Participant 7). "Plants must be better arranged" (Participant 9).
"Appreciation of the new experience"	"Nice to see new things" (Participant 8).

5.4.3 2[^] Objective: Preliminary investigation of the customized VR scenarios on emotional states investigated before, during, and after the virtual exposure.

Based on the modified version of the STAI-Y1, a Wilcoxon Signed Ranks Test revealed that scores related to a state of relaxation were significantly higher after the VR experience (Md = 3, n = 22) than before (Md = 2, n = 22), z = 2.29, *p*-value = .022. Moreover, levels of manifested anxiety and worry significantly decreased after the experience (Md = 1, n = 22) if compared to levels found before the virtual exposure (Md = 2, n=22), z = -3.13, *p*-value = .002. No differences were found for sadness, tightness, annoyance, and anger (-1.73 < z < -0.88; *p*-value > .05) (see Table 5).

Table 5. State emotions comparisons	3.
-------------------------------------	----

	T0 (before VR)			T1 (after VR)			
	25th percentile	50th percentile (Median)	75th percentile	25th percentile	50th percentile (Median)	75th percentile	
Relaxed	2	2	3	2	3	4	
Sad	1	1	1.25	1	1	2	
Uptight	1	2	3	1	1	2.25	
Annoyed or upset	1	1	2	1	1	1.25	
Worried/Anxiety state	1	2	3	1	1	2	

More than half of participants (12/22; 52.2%) expressed being engaged and involved in the virtual environment. During the VR experience, it was observed that 10/22 (43.5%) participants felt a state of pleasantness for more than 5 minutes, 5/22 (21.7%) seemed to enjoy the activity for a period between 1 and 5 minutes, 1/22 (4.3%) for at least a minute, 3/22 (13%) less than 16 seconds and a minority, 3/22 (13%), reported no enjoyable manifestations (see Figure 4). Only two participants (8.7%) showed a state of annoyance for at least a minute, six users (26.1%) showed to be worried during the virtual activity for at least a minute, and 1/22 (4.3%)

for at least two minutes. Four participants (17.4%) expressed sadness for no more than a minute (see Figure 4).



Figure 4. Pie charts related to the frequency of emotional states recorded with the OERS.





5.4.4 3[^] objective: Preliminary investigation of the usability of the VR procedure and apparatus as perceived by the health staff.

Seven individuals from the health staff provided assistance during the VR sessions (at least two VR sessions each). At the end of the entire VR experimental procedure with participants, 5/7 health staff operators took part in a dedicated assessment phase focused on assessing the VR procedure's usability from their individual perspectives and in a Focus Group.

Based on the SUS questionnaire, all participating staff rated the technology is equal to or above 67.5 with an average score of 69 (Standard Deviation=2.24; min.=67.5; max.=72.5), which corresponds to a level of "marginal" acceptability and usability, defined as "OK" [355, 356]. The score is associated with the fact that the health staff participants stressed that the support of a technical person would be needed to ensure the successful implementation of the VR-scenarios, especially due to the role the PC plays in customizing the environments. In general, the low SUS score regards the technical difficulties expected by operators in administering it independently.

The A-MARS - Section E scale was used to assess the subjective quality rating of the VR apparatus. All the health staff individuals reported that they would recommend the VR apparatus to several people that could obtain benefit from its use. Most of the staff respondents (3/5) were willing to use the VR solution at least 10-50 times in a year, and participants were willing to pay for such a product. When requested to evaluate the overall experience they had with the VR solution, healthcare staff participants expressed an average score of 3.5 points on a scale from 1 ("*One of the worst e-tools I've used*") to 5 ("*One of the best e-tools I've used*"). Finally, all healthcare staff said that they thought the use of VR was useful in facilitating relaxation, especially in patients with mild cognitive impairment, since people with more severe impairment may find additional obstacles in experiencing VR (e.g., related to the difficulty in understanding what objects represented).

The focus group was aimed at investigating five main themes related to the VR apparatus usage. For each theme, at least two questions were asked, and the following themes were derived:

• Theme 1: Strengths and advantages of VR use

(1.1: "According to your experience, are there strengths and advantages in using virtual reality as a pleasant activity to promote relaxation in people with cognitive impairment?"; 1.2: "What are they all about?").

All the participants agreed in judging the VR apparatus and procedure as a simple activity to do with users since it does not require particular cognitive or motor skills. Health staff individuals thought it was a valuable activity for promoting pleasant and potentially relaxing experiences in older people with various levels of cognitive impairment. Another advantage observed was the capacity VR can allow hospitalized people to see places they can no longer experience. Another strength from the health staff's point of view is the possibility of exploring a different kind of virtual context and choosing the preferred scenarios according to the users' preferences and needs.

• Theme 2: Weaknesses and disadvantages of using VR

(2.1: "Based on your experience, are there disadvantages and weaknesses in using virtual reality as a pleasant activity for promoting relaxation in people with cognitive impairment?";
2.2: "What are they all about?").

All participants agreed that the heaviness and wearability of the Oculus quest-2 HMD, or the need to wear it, was the main obstacle in the VR procedure and the primary reason why some

people refused to participate in the activity. According to 3/5 health staff professionals, the structured device was seen as unsafe, especially for people with higher cognitive impairment or concerned about their safety.

• Theme 3: Future perspectives and what to change for continuing the use of VR with the health facility's users

(3.1: "Based on your experience, should the virtual reality procedure be modified to continue to be used with people with cognitive impairment?"; 3.2: "How?").

All health staff individuals agreed in suggesting to change the type of HMD, going for models that are lighter and more wearable. For people with higher cognitive impairment, it would be difficult to understand the reason for wearing the HMD, and this may also elicit some anxiety. A less invasive HMD may likely help to overcome these difficulties. A suggestion was to consider exposing users that perceive higher anxiety levels in wearing the HMD, and greater difficulty understanding the activity, in a virtual setup that does not require to wear the HMD (e.g., a Cave Automatic Virtual Environment).

All participants agreed to actively involve the health staff in the VR deployment with users to make the context even more familiar.

Another element to consider for the future is to expand the number of virtual reality experiences to familiarize the person with the equipment and the virtual context.

Three health staff suggested creating other VR environments and making natural contexts more realistic. Another suggestion for future investigation was to deploy the procedure without using the PC with the aim of facilitating the intervention's administration.

• Theme 4: Facilitating factors of the VR procedure deployment

(4.1: "Based on your experience, are there any factors that have facilitated the use of virtual reality?; 4.2: "What are they all about?").

All participants said that a facilitating factor could be the deployment of the VR procedure in a familiar setting to facilitate users' perception of safety.

One health staff member added that, regardless of the severity of the cognitive impairment, having peculiar personality characteristics (e.g., being predisposed to new experiences) could be another facilitating factor. From a general perspective, health staff participants agreed in considering low levels of cognitive decline as a condition that might decrease the benefit provided by the VR intervention (e.g., for issues related to understanding the intervention's aim, and for the difficulty in expressing and understanding if the activity is really appreciated).

However, they reported that the most important human factor to consider when inviting to a VR session is related to the individual personality characteristics, since they observed how two participants - despite a critical level of cognitive impairment - were calm and positive during the VR exposure.

• Theme 5: Risks in using virtual reality

(5.1: "Based on your experience, are there risks in applying this technology the way it was done in this research?"; 5.2: "What are they all about?").

The most significant risk encountered is eliciting intense emotions, not only positive, that must be contained even after exposure to virtual reality. They also considered this a potential point of strength of the solution since it allows people to be exposed to their feelings and memories. Based on the experience gained during the VR sessions, they did not believe there were any further risks as the procedure was assisted.

From the focus group transcript, information on the frequency of agreement and disagreement expressed directly or through examples from each participant was collected for questions 1.1, 2.1, 3.1, 4.1, and 5.1 and was systematically described in Table 6 following the example of previous studies [352, 353].

Question		Type of Response				
	A	SE	NR	D	SD	AR
1.1 Are there strengths and advantages in using VR as a pleasant activity for promoting relaxation in people with cognitive impairment?	4	1	0	0	0	0
2.1 Are there disadvantages and weaknesses in using VR as a pleasant activity for promoting relaxation in people with cognitive impairment?	5	5	0	0	0	0
3.1 Can the VR procedure be modified to continue to be used with people with cognitive impairment?	3	2	0	0	0	0
4.1 Are there any factors that have facilitated the use of VR?	1	3	1	0	0	0

Table 6. Frequencies related to the agreement/disagreement expressed by the health care staff participants to the main themes of the focus group.

5.1 Are there risks in applying this technology the way it was done in this $3 \quad 1 \quad 1 \quad 0 \quad 0 \quad 0$ research?

Notes: A: Explicit agreement; **SE**: Provided example suggesting agreement; **NR**: Did not indicate agreement or dissent (e.g., nonresponse or did not know); **D**: Indicated dissent; **SD**: Provided significant example suggesting dissent; **AR**: Ambivalent response.

5.5 Discussion

The main aim of the current study was to assess the feasibility of and potential benefit of VR exposure in a target of older adults living in long-term care. Feasibility and acceptance proved to be satisfactory, in line with previous research (e.g., 36, 37): the exposure to the VR apparatus was well-tolerated, and no severe or notable adverse side effects (such as nausea, dizziness, or confusion) were reported. The most frequently reported symptoms by only those who wore the VR apparatus for more than five minutes were a slight sensation of fatigue and eye discomfort limited to the period of exposure in the HMD. It is relevant to note that the VR environments used in this study are defined as static scenarios with only slight visual oscillations (e.g., seawater and oscillation of plants due to the action of the wind), but without any rapid movement. This was an intentional design consideration, given that previous work reported that static scenes and a constant speed during movement could prevent cybersickness symptoms (e.g., 354-357).

The main difficulty referred to by nearly half of the participants (both patients and healthcare staff) was the weight and wearability of the HMD. This issue has been previously reported, for instance, in Kalantari et al. study [218] but differs slightly from older studies (e.g., 36) where the model of HMD used (Samsung Gear VR) only weighed 318 grams, while Oculus quest 2, used here, weighs 503 grams. Even if a larger sample is necessary to consider these results representative of the target population, our sample was entirely composed of older adults, individuals with a slender physique and weak muscle mass, which undoubtedly contributes to the perception of heaviness. Despite this limitation, the perceived weight of the device by the participants did not significantly affect the overall experience of the virtual context or withdrawal from the study.

Similar to other studies (e.g., 36, 37, 218), participants expressed verbally or manifested with nonverbal behavior a general state of interest, awareness, engagement, and enjoyment during the VR exposure showing appreciation of the experience. Reports collected during the intervention showed that the exposure to a realistic, natural, and above all, customized virtual

environment adapted to users' preferences and needs, prompted reactions, such as reminiscence of melancholic memories, the aesthetic appreciation of what they were seeing, the sense of safeness and protection in an appreciated context. Interestingly given the research context, participants also expressed feelings of self-realization and "being useful" by providing opinions and suggestions to the experimenters on how to improve the virtual scenarios.

Moreover, more than 70% of the participants stated that they would recommend VR to a friend, confirming the potential benefits of the VR experience in users with cognitive impairment.

The general increased relaxation state and the reduction in worry/anxiety post-VR, both self-reported and observed, are consistent with prior findings (e.g., 36, 37, 358), supporting the hypothesis that VR can serve as a restorative or stress-reducing experience.

A set of core constructs emerged from the focus groups conducted with healthcare professionals. In line with data reported by older adult participants involved in the study, one healthcare staff perceived the HMD to be too difficult and heavy for older adults. Even if corrective measures could be taken to improve comfort when wearing the HMD, novel and lightweight HMD might further facilitate acceptance and ease of use, especially for those in a more fragile state.

A second constraint is related to the difficulty of properly assessing a patient's VR experience in case of severe cognitive impairment. In this case, the collection of bio-physiological parameters could be added as part of the experimental procedure, to better assess the experience of the patient and triangulate different sources of information (e.g., heart rate, skin conductance, etc.).

Finally, the collection of informed consent was reported as a problematic - albeit necessary - task: signing (pen-and-paper) a form was *per se* not well accepted by older adults, whilst other approaches of collecting informed consent might be more suitable and easy-to-deploy (e.g., verbal consent). It is, however, important to add that this step is essential for the deployment of a research project, but it is expected it would not impact the implementation of a therapy program.

Considering the major (perceived) advantages of using VR for patients, focus group participants highlighted the possibilities to deploy environments patients are not able to experience anymore (e.g., seascapes, mountainscapes), because of clinical and/or logistical issues.

Given the feedback provided by staff during the focus group and the results from patient participants using VR, we provide several recommendations to inform future developments

and improvements of the solutions, both from a methodological viewpoint and an operational perspective.

From an operational perspective, the HMD proved to be generally acceptable but a lighterweight HMD might improve usability, particularly for this target group. In addition, to mitigate discomfort and potential anxiety, identifying customized time and space could be beneficial to meet patients' needs. For instance, certain times of day (e.g., sundown syndrome) and locations (unfamiliar rooms) may trigger anxiety and behaviours, which should be avoided when necessary. Finally, preparatory procedures should include a targeted and careful assessment of physical condition, including hearing aids, glasses, and mobility devices, although, in the current research, no major problems were reported.

From a methodological perspective, enriching the VR experience with additional audio and video details could further improve the impact of the VR intervention, considering that patients might be more familiar with some environments (e.g., mountain scenery) over others, and thus paying more attention and wanting to explore specific details. In terms of customization, this option appeared to be appropriate and viable for the target group. In fact, the core objective was to explore the feasibility of the entire procedure (including the availability of certain degrees of customization), as this option could potentially be neither viable nor acceptable for this context.

Adding psycho-physiological measures (e.g., heart rate variability, skin conductance, etc.) could greatly improve the assessment of anxiety/relaxation, particularly in patients with severe cognitive impairment, where there may be challenges in communicating. It is important to acknowledge that participant anxiety levels in this study may have been higher than expected because patients were exposed to VR for the first time and in an unfamiliar room (which was dedicated to this study). An additional methodological issue is related to participants' sample inclusion criteria. In the large majority of studies, measures that evaluate the cognitive decline level (e.g., the MMSE) are used as reference (e.g., 36). In our study we demonstrate that individuals of all ranges of cognition are able to benefit from VR. Preliminary data from the Focus group suggests that personality traits might play a role in reinforcing efficacy of these interventions. Indeed, some specific traits (like willingness to try new experiences) could be a reliable proxy of potential success, therefore these characteristics should be considered (perhaps weighted even more) than cognitive abilities in future studies to properly assess their impact and role in the process.

Another important outcome of this study highlights the potential of VR customization. In terms of acceptance, personal-adaptation refers to the ability to tailor environments and features to

users' preferences (e.g., removing personal triggers for anxiety related to peculiar visual and auditory stimuli that could characterize the VR scenarios). For efficacy, adaptation implies the opportunity to facilitate a stronger connection between the virtual context and a personal history or narrative. In other words, linking the virtual environment with personal constructs/meanings could facilitate the emergence of viable and sustainable "therapeutic" experiences, where the patient is involved somewhere in between the area of elaboration, even considering the reminiscence, that can emerge during the exposure and may contribute to decreasing anxiety (e.g., 359).

5.5.1 Limitations

While promising, it is important to keep in mind several potential limitations that may have introduced bias in the present study. Given the small sample size and recruitment from a single long-term care facility, generalizability of the results should be taken with caution. Moreover, evidence on gender differences in VR's user experience, presence, and usability are still debated in the literature [e.g., 308, 360]. For this reason, future studies should stratify the samples to control the effect of gender other than various socio-demographic variables.

Additional bias could be introduced through subjective researchers' observations, and the lack of triangulation of self-reports with more objective measures, such as biophysiological data.

Authors cannot exclude that an increased level of interaction (e.g., combining metaverse and game mechanics) - not included in this study - could have an impact on the potential acceptance of the proposed experience. In addition, a limitation could be represented by the lack of neural networks to further analyze the different sources of data: this would be possible only with large datasets, and this limitation is directly linked to the purpose of the study and the related sample size.

An additional limitation is linked to the relatively limited range of customization options made available: they were limited as this range was deemed appropriate for the target group of the study, but could be appropriate to suggest more focused stimuli very close to the personal life, interest, values and needs of each user.

Finally, because we used an adapted version of the STAY questionnaire (to suit our population of interest) [36, 37], we cannot ensure the validity of the tool.

5.5.2 Future Directions

Future research should compare different HMDs (e.g., PICO 4 HMD) [361] and weights (among other features - more easily applied to vision glasses, etc.) to see if this has a noticeable

impact on tolerability by older adults. VR-content should also be enriched with added audio and video details.

In the future, the focus could shift towards expanding the range of customization options in virtual scenarios and measuring their impact while also extending the contribution and knowledge provided by our study.

Studies should assess the impact of multiple exposures, both in terms of acceptance and the clinical effect on anxiety. With this approach, healthcare professionals could administer VR sessions several times to the same individuals, which may decrease feelings of fear, anxiety, and prejudice related to the unfamiliar technology [358]. It also allows researchers to evaluate the potential impact of the intervention at different moments of the day [40].

Considering the emotional impact VR sessions could evoke, future interventions should design pre- and post-strategies to help mitigate or manage emotional reactions potentially triggered by the VR. For example, the introduction of listening spaces or dedicated space to contain emotional states. A more focused analysis of the specific diseases would be useful to better understand if, for example, some but not other visual and auditory problems prevent ease-ofuse of the HMD. Stratifying the sample according to the severity of cognitive deterioration to better assess if the VR apparatus deployed could be more effective for some groups over others, as well as to investigate the possible changes to the virtual context in relation to users' preferences and needs. Moreover, following studies should recruit a larger sample size and include objective psycho-physiological assessments.

5.5 Conclusions

This study builds on the growing body of literature empirically evaluating the acceptance and potential impact of immersive VR in long-term care settings, with specific considerations for older adults with varying degrees of cognitive, sensory, and mobility impairments. Based on our knowledge, unique to other work, we took into account the effect of customizing the VR-experience. This represents a valuable step-forward compared to the standard "one-size-fits-all" approach adopted in the majority of the studies in the field of VR [359]. At the same time, while customization could improve acceptance by accommodating and tailoring the VR environment in light of patients' preferences (e.g., 3, 5, 233) it might lead to potential challenges in its deployment. This factor may significantly impact either enhancing or hindering the experience. To our knowledge, this is one of the few studies that have explored customization as an option for this type of population.

Since user-centred approaches need more effort by those responsible for administering the therapy, it is also critical that we seek their input on the design of interventions if we want to encourage their adoption and sustainability (e.g., 359).

The current research took into account the viewpoints of the healthcare staff which brought valuable insight into the patient's previous history and usual emotional patterns and reactions. In addition, feedback from health staff participants gave us the opportunity to consider alternative way in administering the procedure. Indeed, the following study that we will carry out operators will administer the procedure directly to patients and will involve the use of an easy-to-use smartphone interface for customizing the VR environments without the support of a technician.

Moreover, we were able to correlate feedback between patient and staff groups, and across the multiple data collection methods.

Despite limitations, this research contributes to advancing knowledge related to the customization of VR experiences and their related impact on acceptance and potential clinical efficacy in older adults. Given the recognition that current interventions are not satisfactory and that the number of older adults with cognitive decline will only increase, we need to dedicate resources to the co-design and evaluation of novel therapies for this population. VR presents a relatively cost-effective and customizable solution that can be deployed in long term care contexts, offering new hope for personalized-care.

PART 3 ⊣⊢ GENERAL DISCUSSION AND FUTURE PERSPECTIVES

CHAPTER 6

GENERAL CONCLUSIONS AND FUTURE PERSPECTIVES

6.1 Summary of results

Virtual Reality (VR) is extensively deployed in the clinical psychology and psychotherapy fields. It is used to manage different mental health problems, such as anxiety disorders. The application of VR is also growing to improve the effectiveness and flexibility of various relaxation techniques or is used alone to promote a state of calm and anxiety reduction. Indeed, literature shows that VR scenarios representing, for instance, visual and auditory elements of natural relaxing environments can facilitate the learning of relaxation techniques, such as Body Scan and Progressive Muscle Relaxation Training (PMR). Moreover, the opportunity for customization of the VR scenarios could be an element that further facilitates relaxation, allowing participants to experience more realistic emotional conditions, increasing relaxation, sense of presence, and perception of security in the virtual reality. The current Ph.D. project aimed to investigate customized, relaxing VR scenarios, used alone or in association with specific relaxation training protocols, to support the management of anxiety and other related psychological constructs. Three studies characterized by mixed methodologies have been conducted to reach this goal:

- STUDY 1: The first study (Chapter 3) investigated anxiety symptoms, user experience, and individual preferences regarding exposure to customized and non-customized relaxing VR environments combined with a Body Scan audio guiding track. The study was based on an exploratory crossover experimental design mixed-method approach on a non-clinical population. Findings revealed that customized scenarios could enhance the subjective perception of relaxation, immersivity, and pleasure compared to non-customized virtual settings. These findings expand on the impact of customized VR environments, suggesting the role of customization as a promising component in facilitating engagement and promoting a higher perceived state of relaxation. The study was in the preliminary phases based on the Obesity-Related Behavioural Intervention Trial (ORBIT) model [241], specifically the design (Phase Ib) of digital interventions and their preliminary testing (Phase IIa). The results of this study allowed the subsequent works to be conducted in which the next phases of the ORBIT model were implemented to evaluate the preliminary efficacy of the customized relaxation

intervention with a pilot study (Chapter 4) and the feasibility of the one-shot VR procedure on a clinical sample (Chapter 5).

_

STUDY 2: The knowledge about the importance of validating innovative and more flexible interventions and the results from the first study informed and led us to deploy a second pilot study, reported in Chapter 4, which investigated the impact that exposure to a customized context in VR can have on anxiety, stress, depression, coping strategies, quality of life, and other variables related to engagement and the sense of presence, compared to a standard treatment that involves the application of guided imagination. In this case, the virtual or imagined scenarios were administered with an abbreviated PMR, and the main aim was to understand if VR can have the potential to overcome possible limitations of the "as usual" Guided Imagery treatment. The focus was on assessing the impact on state anxiety of PMR associated with a customized-relaxing scenario in VR and the role of virtual scenarios in facilitating the recall of relaxing images and the sense of presence. Moreover, this research was interested in understanding the impact of two web-based interventions used to administer the PMR. To investigate the research's aims, the sample was composed of 108 university students randomly assigned to one of three different experimental conditions: 1) PMR Training sessions by Zoom and VR exposure; 2) PMR Training sessions by Audio-track and VR exposure; 3) PMR Training sessions by Zoom and Guided Imagery exposure. The study is novel and imbued with innovative aspects based on the new challenges in validating psychological treatments for relaxation that allow them to be administered online and with the help of customized, immersive techniques supported by virtual reality. The results highlighted how being in a customized virtual scenario promoted a greater reduction of the self-perceived state anxiety than in the imaginative condition, when individuals were directly exposed to the VR or GI experience and when they had to recall the strategies learned during the relaxation protocol. Independently of the type of web-based relaxation training administered before, VR is more effective than inimagination exposure in allowing a subjective reduction of the state of tension and worry felt (e.g., 277-279, 292). The audio-track group achieved the most significant relaxation during training and exposure to VR. At the same time, it was revealed that the therapist's presence during the training sessions, and the booster effect of VR, helped more in learning the relaxation procedure.

STUDY 3: Considering the importance of assessing customized VR environments to _ reduce arousal and anxiety in clinical groups, a third proof-of-concept study, reported in Chapter 5, was conducted involving a sample of older adults diagnosed with dementia. This research topic is part of a new line of research investigating usability and the possible impact on reducing activation, anxiety, and negative emotions that can be experienced frequently in the ordinary day of people living in healthcare facilities. To add useful and innovative components starting from what already exists and driven by the indication of Appel et al. [361] and Moyle et al. [221], the customization aspect was added to the current experimental design as well as the healthcare professionals' point of view was investigated to obtain useful feedback to inform future studies that our research group is going to implement in the next months. Previous studies showed how VR could increase engagement and relaxation in older adults with and without cognitive impairment. The main aim of this study was to investigate the feasibility of customized naturalistic VR scenarios by assessing motion-sickness effects, engagement, pleasantness, and emotions related to exposure in the virtual context. Feasibility and acceptability have proved satisfactory, considering that the VR experience was well-tolerated and no adverse side effects have been reported. Regarding the output from the focus group and overall research analysis, different recommendations were listed in Chapter 5 to inform future developments and improvements of the solution, both from a methodological and operational perspective. From an operational perspective, HMD has proved to be generally acceptable. At the same time, lightweight HMD might improve usability, particularly for this target group. In addition, to mitigate discomfort and potential anxiety, identifying personalized time slots and dedicated familiar rooms could be beneficial to meet patients' needs. For instance, researchers should offer different settings for VR exposure (e.g., patient's room rather than less familiar locations). From a methodological perspective, enriching the VR experience with additional audio and video elements could further improve the efficacy of a VR intervention, considering patients familiarity with some environments and therefore paying more attention to specific details. As previously mentioned, adding psycho-physiological measures (e.g., heart rate variability, skin conductance, etc.) could largely improve the assessment of anxiety/relaxation patterns, particularly in patients with high levels of cognitive impairment. From this perspective, it cannot be excluded that the reported levels in terms of anxiety are somehow linked to the fact that patients were exposed to VR for the first time, within a unique timeslot, and in a

dedicated (albeit not familiar to patients) room. Future research could assess the impact of multi-session interventions in terms of acceptability and effect on anxiety patterns and the impact of multi-sessions embedded in therapeutic intervention. Considering the emotional impact and the personal experiences a VR session could evoke, it might be worth identifying strategies to create pre- and post-VR dedicated sessions. This is in light of managing cognitive/emotional reactions that could emerge from the VR experience and supporting patients. An additional methodological issue is related to patients' selection. In most studies, measures that evaluate the cognitive decline level (e.g., the MMSE) are used as reference (e.g., 36). The present study suggests that regardless of their cognitive decline scores or clinical issues, patients with specific personality profiles could benefit from VR, particularly those more prone to experiment with novel approaches.

6.2 Take-home messages and future perspectives

The current studies highlighted the importance of personalizing and customizing VR environments from feasibility and efficacy perspectives.

The current Ph.D. project introduced a novel therapeutic approach to promote relaxation, structuring evidence from the literature on the features of relaxing natural scenarios and using standardized relaxation training administered with alternative tools (e.g., Zoom platform, audio-track, immersive VR tool).

Indeed, inspired by Recchia et al. [362], we used elements of evidence-based therapies described in the literature and introduced new technological tools to administer the interventions tested. We started by obtaining qualitative and quantitative information to build the software and procedure apparatus based on the personal experience of convenience samples. This pre-clinical development phase provided as an output a combination of digital active ingredients and digital platforms suitable for the best use.

The acceptance and preliminary efficacy of the interventions were tested with promising results. In other words, customizing the virtual environments could promote an experience closer to users' needs and preferences, potentially providing greater efficacy than non-customized ones. Moreover, exposure to virtual scenarios was found as a better way to reduce anxiety than the standard modality consisting in the deployment of the guided imagery technique.

In general, these preliminary results assist in planning future studies on the effect of relaxation training in customized VR environments on large-scale clinical and non-clinical samples. They examined the feasibility and preliminary effectiveness that is indicated in the ORBIT model framework as a key objective of the pre-clinical assessment of interventions. Based on subsequent randomized clinical trials that will have to investigate the digitalized relaxation protocol introduced with the present thesis on large clinical samples, our solutions could represent future digital therapies to be delivered in the ecological situations of the real-world context, in which health professionals, family members, or users by themselves can use virtual reality as a treatment excipient for delivering evidence-based interventions for mental health and well-being based on cognitive and behavioral therapy (e.g., [362]).

This speculation is also based on one of the crucial take-home messages of the presented studies about the prominent impact of being exposed to realistic and customized scenarios in VR. Specifically, the effects that exposure to VR may have in more ecological everyday settings are supported by the results of the study presented in Chapter 4, showing the benefit for individuals to transfer the relaxation skills learned in VR to real-world situations. Considering the positive outcomes found in the above-reported studies, future research may explore using VR during the training sessions with customizable avatars that guide users in learning relaxation techniques more interactively, based on Artificial Intelligence (AI) empowered solutions fitting the users' needs. One planned study of our research group will go in this direction, considering both user samples from the non-clinical population and individuals that suffer from chronic pain and oncological diseases. Indeed, the aim will be to use complemented relaxation strategies guided by a virtual therapist to learn how to manage chronic pain based on individuals with heterogeneous diseases and to reduce state-anxiety before and after surgical procedures with breast cancer patients.

The fact that evaluating web-based training without the support of virtual reality has allowed us to achieve promising results also suggests the opportunity of studying the delivery of standardized relaxation training by using alternative technological solutions, including the use of Chatbots, a computing program that uses AI and Natural Language Processing (NLP) to understand customer questions and automate their answers, simulating human conversation [363]. Chatbots may be deployed as virtual assistants, interacting with users, and providing specific support in case of problems during training. Using chatbots to support the delivery of relaxation interventions may help to overcome the criticalities of remote access and interaction with the therapist, providing interactive support without invading the individual's personal space during relaxation. Chatbots may also overcome the possible limitations of using the audio-track guided relaxation as they are more interactive and potentially engaging solutions that could reduce the risk of drop-out and allow the person to resolve any doubts associated with certain activities during the training phase.

From the perspective of the Stepped Care Model [364], we can hypothesize that the different investigated relaxation strategies and means for delivering a relaxation treatment could be administered in the future as stand-alone or complementary strategies according to the goals, the target problems and the population considered. Indeed, for example:

- A web-based delivery mode deployed autonomously without the presence of a therapist could be used to reach the needs of the entire population to acquire a technique useful to prevent or manage bearable and not invalidated anxiety and stress problems;

- A chatbot-based approach could be a more tailored mode of delivering mental well-being interventions that also provide specific information based on the type of problem affecting the target user;

- An approach in which the remote therapist is guiding the relaxation training could be a suitable modality for people who are doing treatment with a professional but do not need a strictly personalized support during the training phase;

- The use of VR could be more appropriate for interventions to be delivered in clinical and healthcare facilities where it is necessary to provide healthcare services to several people at the same time, but where limitations in personnel resources may complicate the service delivery.

Each intervention can be adequate in different contexts, and it is essential to be investigated with rigorous studies on the most effective protocols best fitting the different target populations.

Another aspect that future studies should better address is the deployment of more ecological experimental designs. For example, healthcare professionals can administer the experimental procedure several times to stem the potential obstacle of a "one shot" intervention with VR. This would potentially increase the positive effect on anxiety and the generalization of what has been learned from the virtual experience to other real-world contexts (e.g., facing an exam at the university) and also decrease feelings of fear and anxiety related to the non-familiarity with the VR apparatus, as is the case of people with a diagnosis of dementia.

Moreover, the thesis's studies informed on the possibility of introducing other improvements to overcome limitations during the administration procedures:

- The use of proper software to personalize and customize virtual scenarios by smartphone. Future applications of VR should be easily administered autonomously by users or healthcare professionals without peculiar expertise in the technology field.

Considering this, one of the difficulties of the current studies was using the PC to customize scenarios. Indeed, sometimes software updating operations occur, and we noticed that this would make it more difficult the procedures' administration by therapists or users not able to cope with unexpected events, mainly due to technical issues. For this reason, in our next studies a dedicated software by Smartphone will be deployed to customize virtual scenarios more easily by non-expert administrators and users, connected to the HMD without a link cable.

- Introducing more scenarios and stimuli that users can select to customize their relaxing virtual experience.
- The integration of the virtual reality software with a virtual avatar, AI powered chatbots, web-based audio-track, and assessment measures in public, embraceable, and familiar Personal Health Record PHR platform (e.g., the Trentino Citizens Clinical Record-TReC platform), accessible to the population of communities living in specific region or country to promote an Integrated Care Model for patients' empowerment and data repository [365], and to facilitate improved communication and quality of the services delivered with advantages for both health care staff and citizens/patients [366].
- The deployment of customization of virtual reality scenarios more dynamically during exposure to respond to users' needs and preferences based on their emotional responses, monitored continuously with psycho-physiological cues, self-reported information, body movements, facial expressions, eye tracking, and brain signals. AI techniques (e.g., machine learning, deep learning, and natural language processing) provide computers with reasoning and analytical capabilities, and the possibility of merging AI and customized VR experiences has yet to be fully realized and studied (e.g., 3, 5).
- Despite the advantages of personalized VR in offering an intervention based on user's needs and preferences, future research is also needed to overcome possible obstacles, such as the difficulty of integrating complex technological and personalized systems in healthcare facilities [3, 367, 368].

To conclude, the introduction of VR and web-based training could positively impact the reduction of therapy sessions and their cost, as it allows partial self-management of the treatment. If the combined administration of a personalized and customized VR relaxation scenario and the deployment of a standardized and validated relaxation protocol effectively obtain a better subjective perception of relaxation than the standard procedure, it could offer a valuable opportunity for users to do relaxation with greater autonomy. For this reason,
assessing the efficacy of a validated relaxation intervention in additional VR-based settings could extend the treatment administration options, creating interesting opportunities for introducing innovative interventions in situations where the standard procedure is more challenging to be effectively deployed.

The results and limitations of these studies will be the basis for the design and implementation of future research, currently in progress at the Bruno Kessler Foundation Digital Health research unit, that will involve the validation of VR-based relaxation solutions with different clinical populations, such as individuals with cognitive deterioration, cancer, chronic pain, and palliative care.

REFERENCES

- [1] Pizzoli et al. (2019). Comparison of relaxation techniques in virtual reality for breast cancer patients. 5th Experiment@ International Conference (exp.at'19), June 12th-14th, 2019, University of Madeira, Funchal, Madeira, Portugal.
- [2] Beverly, E. et al. (2022). A tranquil virtual reality experience to reduce subjective stress among COVID-19 frontline healthcare workers. *PloS one*, *17*(2), e0262703. https://doi.org/10.1371/journal.pone.0262703.
- [3] Pizzoli, S.F.M., Mazzocco, K., Triberti, S., Monzani, D., Raya, M.L.A., & Pravettoni, G. (2019). User-Centered Virtual Reality for Promoting Relaxation: An Innovative Approach. *Frontiers in Psychology*, 10(479). doi: 10.3389/fpsyg.2019.00479.
- [4] Holland, A. C., and Kensinger, E. A. (2010). Emotion and autobiographical memory. *Phys. Life Rev.* 7, 88–131. doi: 10.1016/j.plrev.2010.01.006.
- [5] Heyse, J., Torres Vega, M., De Jonge, T., De Backere, F., & De Turck, F. (2020). A Personalized Emotion-Based Model for Relaxation in Virtual Reality. *Applied Sciences*, 10, 6124. Doi: 10.3390/app10176124.
- [6] Mazgelytė, E., Rekienė, V., Dereškevičiūtė, E., Petrėnas, T., Songailienė, J., Utkus, A., Chomentauskas, G., & Karčiauskaitė, D. (2021). Effects of Virtual Reality-Based Relaxation Techniques on Psychological, Physiological, and Biochemical Stress Indicators. *Healthcare*, 9, 1729. https://doi.org/10.3390/healthcare9121729.
- [7] Umanodan, R., et al. (2009). Effects of a Worksite Stress Management Training Program with Six Short-hour Sessions: A Controlled Trial among Japanese Employees. *Journal of Occupational Health*, 51, 294-302.
- [8] Bernstein, E. A. and Borkovec, T. D. (1973). *Progressive relaxation training: A manual for the helping professions,* Champaign, IL: Research Press.
- [9] Bernstein, D. A., Carlson, C. R., & Schmidt, J. E. (2007). Progressive relaxation: Abbreviated methods. In P. M. Lehrer, R. L. Woolfolk, & W. E. Sime (Eds.), *Principles* and practice of stress management (pp. 88–122). The Guilford Press.
- [10] Toussaint, L. et al. (2021). Effectiveness of Progressive Muscle Relaxation, Deep Breathing, and Guided Imagery in Promoting Psychological and Physiological States of Relaxation. *Hindawi- Evidence-Based Complementary and Alternative Medicine, 2*, 5924040. doi: 10.1155/2021/5924040. PMID: 34306146; PMCID: PMC8272667.
- [11] Keptner K. M., Fitzgibbon C., & O'Sullivan J. (2020). Effectiveness of anxiety reduction interventions on test anxiety: a comparison of four techniques incorporating sensory modulation. *British Journal of Occupational Therapy*, *70*(6), 1-9.

- [12] Taylor, A.G., Goehler, L.E., Galper, D.I., et al. (2010). Top-down and bottomup mechanisms in mind body medicine: Development of an integrative framework for psychophysiological research. *The Journal of Science and Healing 6*(1), 29–41.
- [13] Pv, J., & Lobo, S. M. (2020). Effectiveness of relaxation technique in reducing stress among nursing students. *International Journal of Nursing and Health Research*, 2(1):54-56.
- [14] Chaudhuri, A., Manna, M., Mandal, K., & Pattanayak, K. (2020). Is there any effect of progressive muscle relaxation exercise on anxiety and depression of the patient with coronary artery disease? *International Journal of Pharma Research and Health Sciences*, 8(5), 3231–3236. doi: 10.21276/ijprhs.2020.05.03.
- [15] Pradhan, J., Pradhan, R., Samantaray, K., & Pahantasingh, S. (2020). Progressive muscle relaxation therapy on anxiety among hospitalized cancer patients. *European Journal of Molecular & Clinical Medicine*, 7(8), 1485-1488.
- [16] Blanchard, E.B., & Andrasik, F. (1985). *Management of chronic headache: A psychological approach*. Elmsford, New York: Pergamon Press.
- [17] Baird, C.L., & Sands, L. (2004). A pilot study of the effectiveness of guided imagery with progressive muscle relaxation to reduce chronic pain and mobility difficulties of osteoarthritis. *Pain Management Nursing*, 5(3), 97-104, https://doi.org/10.1016/j.pmn.2004.01.003.
- [18] Charalambous, A., Giannakopoulou, M., Bozas, E., & Paikousis, L. (2016). A Randomized Controlled Trial for the Effectiveness of Progressive Muscle Relaxation and Guided Imagery as Anxiety Reducing Interventions in Breast and Prostate Cancer Patients Undergoing Chemotherapy. *Hindawi-Evidence-Based Complementary and Alternative Medicine*, ttp://dx.doi.org/10.1155/2015/270876.
- [19] Nasiri, S., Akbari, H., Tagharrobi, L., & Tabatabaee, A.S. (2018). The effect of progressive muscle relaxation and guided imagery on stress, anxiety, and depression of pregnant women referred to health centers. *Journal of Education and Health Promotion*, 1(7), 41. doi: 10.4103/jehp.jehp_158_16.
- [20] Tinga, A.M., De Back, T.T., & Louwerse, M.M. (2019). Non-invasive neurophysiological measures of learning: A meta-analysis. *Neuroscience & Biobehavioral Reviews*, 99, 59–89.
- [21] De Gauquier, L., Brengman, M., Willems, K., Van Kerrebroeck, H. (2019). Leveraging advertising to a higher dimension: Experimental research on the impact of virtual reality on brand personality impressions. *Virtual Real., 23*, 235–253.
- [22] Lau, K.W., & Lee, P.Y. (2019). Shopping in virtual reality: A study on consumers' shopping experience in a stereoscopic virtual reality. *Virtual Real, 23*, 255– 268.

- [23] Borkovec, T.D. e Sides, J.K. (1979). The contribution of relaxation and expectancy to fear reduction via graded imaginal exposure to feared stimuli. *Behaviour Research and Therapy*, *17*, 529-540.
- [24] Carlson, C. R., & Hoyle, R. H. (1993). Efficacy of abbreviated progressive muscle relaxation training: A quantitative review of behavioral medicine research. *Journal of Consulting and Clinical Psychology*, 61(6), 1059–1067. https://doi.org/10.1037/0022-006X.61.6.1059.
- [25] Chung, J., Mundy, M.E., Hunt, I., Coxon, A., Dyer, K.R., & McKenzie, S. (2021). An Evaluation of an Online Brief Mindfulness-Based Intervention in Higher Education: A Pilot Conducted at an Australian University and a British University. *Frontiers in Psychology*, 12, 752060, doi: 10.3389/fpsyg.2021.752060.
- [26] Pickett, S.M., et al. (2022). The comparison of brief, online mindfulness and relaxation interventions to reduce stress and improve sleep-related outcomes in college students. *Journal of American College Health*, DOI: 10.1080/07448481.2022.2066979.
- [27] Lim, P.Y., Dillon, D. & Chew, P.K.H. (2020). A Guide to Nature Immersion: Psychological and Physiological Benefits. *Int. J. Environ. Res. Public Health* 17(16), 5989. doi: 10.3390/ijerph17165989.
- [28] Park, S.H. & Mattson, R.H. (2009). Ornamental indoor plants in hospital rooms enhanced health outcomes of patients recovering from surgery. *J Altern. Complement. Med.* 15(9), 975-80. doi: 10.1089/acm.2009.0075.
- [29] Jo, H., Song, C. & Miyazaki, Y. (2019). Physiological Benefits of Viewing Nature: A Systematic Review of Indoor Experiments. *Int J Environ Res Public Health*. 16(23), 4739. doi: 10.3390/ijerph16234739.
- [30] Benjamin, K., Edwards, N., Ploeg, J. & Legault, F. (2014). Barriers to physical activity and restorative care for residents in long-term care: a review of the literature. *J Aging Phys Act.* 22, 154–65. doi: 10.1123/japa.2012-0139.
- [31] Franco, L.S., Shanahan, D.F. & Fuller, R.A. (2017). A Review of the Benefits of Nature Experiences: More Than Meets the Eye. *Int J Environ Res Public Health*. 14(8), 864. doi: 10.3390/ijerph14080864.
- [32] Park, S.H. & Mattson, R.H. (2009). Ornamental indoor plants in hospital rooms enhanced health outcomes of patients recovering from surgery. *J Altern Complement Med.* 15, 975–80. doi: 10.1089/acm.2009.0075.
- [33] Park, B.J., Tsunetsugu, Y., Kasetani, T., Kagawa, T. & Miyazaki, Y. (2010). The physiological effects of Shinrin-yoku (taking in the forest atmosphere or forest bathing): evidence from field experiments in 24 forests across Japan. *Environ Health Prev Med.* 15, 18–26. doi: 10.1007/s12199-009-0086-9.
- [34] Lin, C.X., Lee, C., Lally, D. & Coughlin, J.F. Impact of Virtual Reality (VR) Experience on Older Adults' Well-Being. In: Zhou, J., Salvendy, G. (eds) *Human Aspects of IT for the Aged Population. Applications in Health, Assistance, and*

Entertainment. ITAP 2018. Lecture Notes in Computer Science, vol 10927. (Springer, Cham., 2018). https://doi.org/10.1007/978-3-319-92037-5_8.

- [35] Riva G. (2022). Virtual Reality in Clinical Psychology. *Comprehensive Clinical Psychology* 91–105. doi: 10.1016/B978-0-12-818697-8.00006-6.
- [36] Appel, L. et al. (2020). Older adults with cognitive and/or physical impairments can benefit from immersive virtual reality experiences: A feasibility study. *Frontiers in Medicine* 6, 329. doi: 10.3389/FMED.2019.00329.
- [37] Appel, L. et al. (2021). Administering Virtual Reality Therapy to Manage Behavioral and Psychological Symptoms in Patients With Dementia Admitted to an Acute Care Hospital: Results of a Pilot Study. *JMIR Form. Res.* 5(2), e22406. doi: 10.2196/22406.
- [38] Ugliotti, F. M. Virtual Representations for Cybertherapy: A Relaxation Experience for Dementia Patients. In Handbook of Research on Implementing Digital Reality and Interactive Technologies to Achieve Society 5.0 (pp. 1-23). (IGI Global, 2022).
- [39] Syed-Abdul, S. et al. (2019). Virtual reality among the elderly: a usefulness and acceptance study from Taiwan. *BMC Geriatr. 19*, 223. https://doi.org/10.1186/s12877-019-1218-8.
- [40] Niki, K. et al. (2020). Immersive Virtual Reality Reminiscence Reduces Anxiety in the Oldest-Old Without Causing Serious Side Effects: A Single-Center, Pilot, and Randomized Crossover Study. *Front Hum Neurosci.* 14, 598161. doi: 10.3389/fnhum.2020.598161.
- [41] Zhai, K. et al. (2021). Virtual Reality Therapy for Depression and Mood in Long-Term Care Facilities. *Geriatrics*, 6(2), 58. https://doi.org/10.3390/geriatrics6020058.
- [42] https://dictionary.apa.org/relaxation
- [43] Borkovec, T.D., Johnson, M.C., & Block, D.L. (1984). Evaluating experimental designs in relaxation training. In R. L. Woolfolk & P. M. Lehrer (Eds.), *Principles and practice of stress management*. New York: Guilford Press.
- [44] Shapiro, D.H. (1984). Overview: Clinical and physiological comparison of meditation with other self-control strategies. *Meditation: Classic and contemporary perspectives*, 5-12.
- [45] West, M.A. (Ed.). (1987). *The psychology of meditation*. Clarendon Press/Oxford University Press.
- [46] Lazarus, R., & Folkman, S. (1984). *Stress, Appraisal, and Coping.* New York: Springer.

- [47] Torales, J., O'Higgins, M., Barrios, I., González, I., & Almirón, M. (2020). An overview of Jacobson's progressive muscle relaxation in managing anxiety. *Revista* Argentina de clinica psicologica, 29(3), 17.
- [48] Jensen, C., Forlini, C., Partridge, B., & Hall, W. (2016). Australian University Students' Coping Strategies and Use of Pharmaceutical Stimulants as Cognitive Enhancers. *Front Psychol.* 1(7), 277. doi: 10.3389/fpsyg.2016.00277. PMID: 26973573; PMCID: PMC4771940.
- [49] Smith, J.C. (1988). Steps toward a cognitive-behavioral model of relaxation. Biofeedback and Self-Regulation 13, 307–329. https://doi.org/10.1007/BF00999087.
- [50] Smith, J. C. (1989).Relaxation dynamics: *A cognitive-behavioral approach to relaxation*. Champaign, IL: Research Press.
- [51] Smith, J. C. (1989). *Cognitive-behavioral relaxation training: Theory and practice*. New York: Springer.
- [52] Bernstein, D. A., & Borkovec, T. D. (1973). *Progressive relaxation training: A manual for the helping professions*. Research Press.
- [53] Jacobson, E. (1938). *Progressive relaxation* (2nd ed.). Univ. Chicago Press.
- [54] https://dictionary.apa.org/progressive-relaxation.
- [55] Bernstein, D., Carlson, C., & Schmidt, J. (2007) Progressive Relaxation, Abbreviated methods. In P. Lehrer (Ed.), *Principles and practice of stress management* (pp. 88-125). New York: Guilford Press.
- [56] Bernstein, D. A., Borkovec, T. D., & Hazlett-Stevens, H. (2000). New directions in progressive relaxation training: A guidebook for helping professionals. Praeger Publishers/Greenwood Publishing Group.
- [57] Borkovec, T. D., & Sides, J. K. (1979). Critical procedural variables related to the physiological effects of progressive relaxation: A review. *Behaviour Research and Therapy*, *17*(2), 119–125. https://doi.org/10.1016/0005-7967(79)90020-2.
- [58] Lehrer, P. M., Woolfolk, R. L., Rooney, A. J., McCann, B., & Carrington, P. (1983). Progressive relaxation and meditation: A study of psychophysiological and therapeutic differences between two techniques. *Behaviour Research and Therapy*, 21(6), 651–662. https://doi.org/10.1016/0005-7967(83)90083-9.
- [59] Luebbert, K., Dahme, B., & Hasenbring, M. (2001). The effectiveness of relaxation training in reducing treatment-related symptoms and improving emotional adjustment in acute non-surgical cancer treatment: A meta-analytical review. *Psycho-Oncology*, 10(6), 490–502. https://doi.org/10.1002/pon.537.
- [60] Keptner KM, Fitzgibbon C, O'Sullivan J. Effectiveness of anxiety reduction interventions on test anxiety: A comparison of four techniques incorporating sensory

modulation. *British Journal of Occupational Therapy*. 2021;84(5):289-297. doi:10.1177/0308022620935061.

- [61] Singh, N., Moneghetti, K.J., Christle, J.W., Hadley, D., Plews, D., Froelicher, V. (2018). Heart Rate Variability: An Old Metric with New Meaning in the Era of using mHealth Technologies for Health and Exercise Training Guidance. Part One: Physiology and Methods, *Arrhythmia & Electrophysiology Review*;7(3):193–8. https://doi.org/10.15420/aer.2018.27.2.
- [62] Conrad, A., & Roth, W. T. (2007). Muscle relaxation therapy for anxiety disorders: It works but how? *Journal of Anxiety Disorders*, 21(3), 243–264. https://doi.org/10.1016/j.janxdis.2006.08.001.
- [63] Pifarré, P., Simó, M., Gispert, J.D., Plaza, P., Fernández, A., & Pujol, J. (2015). Diazepam and Jacobson's progressive relaxation show similar attenuating short-term effects on stress-related brain glucose consumption. *Eur Psychiatry*, 30(2), 187-92. doi: 10.1016/j.eurpsy.2014.03.002. Epub 2014 Jun 5. PMID: 24908148.
- [64] Pawlow, L.A., & Jones, G.E. (2005). The impact of abbreviated progressive muscle relaxation on salivary cortisol and salivary immunoglobulin A (sIgA). *Appl Psychophysiol Biofeedback*, 30(4), 375-87. doi: 10.1007/s10484-005-8423-2. PMID: 16385425.
- [65] Wolpe, J. (1968). Psychotherapy by reciprocal inhibition. *Conditional Reflex*, 3(4), 234–240. https://doi.org/10.1007/BF03000093.
- [66] Paul, G. L. (1966). *Insight versus desensitization in psychotherapy*. Stanford, CA: Stanford University Press.
- [67] Carlson, C. R., & Hoyle, R. H. (1993). Efficacy of abbreviated progressive muscle relaxation training: A quantitative review of behavioral medicine research. *Journal of Consulting and Clinical Psychology*, 61(6), 1059– 1067. https://doi.org/10.1037/0022-006X.61.6.1059.
- [68] Toussaint, L., Nguyen, Q.A., Roettger, C., Dixon, K., Offenbächer, M., Kohls, N., Hirsch, J., & Sirois, F. (2021). Effectiveness of Progressive Muscle Relaxation, Deep Breathing, and Guided Imagery in Promoting Psychological and Physiological States of Relaxation. Evid Based *Complement Alternat Med.*, 2, 5924040. doi: 10.1155/2021/5924040. PMID: 34306146; PMCID: PMC8272667.
- [69] Zargarzadeh, M., & Shirazi, M. (2014). The effect of progressive muscle relaxation method on test anxiety in nursing students. *Iran J Nurs Midwifery Res.*, 19(6), 607-12. PMID: 25558258; PMCID: PMC4280725.
- [70] Chaudhuri, A., Manna, M., Mandal, K., Pattanayak, K. (2020). Is there any effect of progressive muscle relaxation exercise on anxiety and depression of the patient

with coronary artery disease? *International Journal of Pharma Research and Health Sciences*. 8(5), 3231–3236. doi: 10.21276/ijprhs.2020.05.03.

- Zigmond, A.S., Snaith, R.P. (1983). The hospital anxiety and depression scale.
 Acta Psychiatrica Scandinavica. 67(6), 361–370. doi: 10.1111/j.1600-0447.1983.tb09716.x.
- [72] Pradhan, J., Pradhan, R., Samantaray, K., Pahantasingh, S. (2020). Progressive muscle relaxation therapy on anxiety among hospitalized cancer patients. *European Journal of Molecular & Clinical Medicine*. 7(8), 1485–1488.
- [73] Grover, N., Kumaraiah, V., Prasadrao, P.S., D'souza, G. (2002). Cognitive behavioural intervention in bronchial asthma. *J Assoc Physicians India*. 50, 896-900. PMID: 12126343.
- [74] Storb, S.H., Strahl, H.M. (2006). Gruppentherapie bei Tinnitus aurium Eine retrospektive Betrachtung der Behandlungseffizienz [Cognitive group therapy for tinnitus--a retrospecive study of their efficacy]. *Laryngorhinootologie*. 85(7), 506-11. German. doi: 10.1055/s-2006-925231.
- [75] Conrad, A., Roth, W. T. (2007). Muscle relaxation therapy for anxiety disorders: It works but how? *Journal of Anxiety Disorders*, 21(3), 243–264. https://doi.org/10.1016/j.janxdis.2006.08.001
- [76] https://www.apa.org/topics/anxiety
- [77] Rosen, J. B., & Schulkin, J. (1998). From normal fear to pathological anxiety. *Psychological Review*, 105(2), 325–350. https://doi.org/10.1037/0033-295X.105.2.325
- [78] Torales, J., Barrios, I., Almirón, M., De la Cueva, R. (2017). Physiotherapy in the treatment of anxiety disorders. *International Journal of Culture and Mental Health*, 10(3), 298–299. https://doi. org/10.1080/17542863.2017.1303075
- [79] Conrad, A., Roth, W. T. (2007). Muscle relaxation therapy for anxiety disorders: It works but how? *Journal of Anxiety Disorders*, 21(3), 243–264. https://doi.org/10.1016/j.janxdis.2006.08.001
- [80] Kumar, S., Nayak, R., Kumari, S. (2015). Effectiveness Jacobson's progressive muscle relaxation technique (PMRT) to relieve anxiety among alcoholic patients MHI, SCB, Cuttack, Odisha. *Journal of Nursing and Health Science*, *4*(4), 1–6.
- [81] Manzoni, G.M., Pagnini, F., Castelnuovo, G. et al. (2008). Relaxation training for anxiety: a ten-years systematic review with meta-analysis. *BMC Psychiatry*, *8*, 41 https://doi.org/10.1186/1471-244X-8-41.
- [82] Ramasamy, S., Panneerselvam, S., Govindharaj, P., Kumar, A., Nayak, R. (2018). Progressive muscle relaxation technique on anxiety and depression among persons affected by leprosy. *Journal of Exercise Rehabilitation*, 14(3), 375–381. https://doi.org/10.12965/jer.1836158.079.

- [83] Redd, W.H., Montgomery, G.H., DuHamel, K.N. (2001). Behavioral intervention for cancer treatment side effects. *Journal of the National Cancer Institute*, 93(11), 810–823. https://doi.org/10.1093/jnci/93.11.810.
- [84] Kaplan, S. (1995). The restorative benefits of nature: Toward an integrative framework. *Journal of Environmental Psychology*, *15*(3), 169–182. https://doi.org/10.1016/0272-4944(95)90001-2.
- [85] Kaplan, R., & Kaplan, S. (1989). *The experience of nature: A psychological perspective*. Cambridge University Press.
- [86] Ulrich, R.S. (1984). View through a window may influence recovery from surgery. *Science*. 27, 224(4647):420-1. doi: 10.1126/science.6143402.
- [87] Ulrich, R.S., Simons, R.F., Losito, B.D., Fiorito, E., Miles, M.A., Zelson, M. (1991). Stress recovery during exposure to natural and urban environments. *Journal of Environmental Psychology*, *11*(3), 201–230. https://doi.org/10.1016/S0272-4944(05)80184-7.
- [88] Joye, Y., & Van den Berg, A. (2011). Is love for green in our genes? A critical analysis of evolutionary assumptions in restorative environments research. *Urban Forestry & Urban Greening*, *10*(4), 261-268.
- [89] Dosen, A.S., Ostwald, M.J. (2013). Prospect and Refuge Theory: Constructing a Critical Definition for Architecture and Design. *The International Journal of Design in Society*, 6 (1), 9-24. doi:10.18848/2325-1328/CGP/v06i01/38559.
- [90] Appleton, Jay (1975/1996 rev.ed.). *The Experience of Landscape*. John Wiley & Sons, London.
- [91] Gatersleben, B., Andrews, M. (2013). When walking in nature is not restorativethe role of prospect and refuge. *Health Place*. 20, 91-101. doi: 10.1016/j.healthplace.2013.01.001.
- [92] Wheeler, B.W., White, M., Stahl-Timmins, W., Depledge, M.H. (2012). Does living by the coast improve health and wellbeing?. *Health & place*, *18*(5), 1198-1201.
- [93] Schweitzer, R., Glab, H., Brymer, E. (2018). The human-nature relationship: a phenomenological-relational perspective. *Frontiers in psychology*, 9.
- [94] Nguyen, J., & Brymer, E. (2018). Nature-based guided imagery as an intervention for state anxiety. *Frontiers in Psychology*, *9*, 1858. doi: 10.3389/fpsyg.2018.01858.
- [95] Koivisto, M., & Grassini, S. (2022). Mental imagery of nature induces positive psychological effects. *Current Psychology*, 1-16.
- [96] Sanadgol, S., Firouzkouhi, M., Badakhsh, M., Abdollahimohammad, A., & Shahraki-vahed, A. (2020). Effect of guided imagery training on death anxiety of nurses at COVID-19 intensive care units: a quasi-experimental study. *Neuropsychiatria i*

Neuropsychologia/Neuropsychiatry and Neuropsychology, 15(3), 83-88. https://doi.org/10.5114/nan.2020.101290.

- [97] Arbuthnott, K.D., Arbuthnott, D.W., & Rossiter, L. (2001). Guided imagery and memory: Implications for psychotherapists. *Journal of Counseling Psychology*, *48*(2), 123–132. https://doi.org/10.1037/0022-0167.48.2.123.
- [98] Ji, J.L., Heyes, S.B., MacLeod, C., & Holmes, E.A. (2016). Emotional mental imagery as simulation of reality: Fear and beyond—A tribute to Peter Lang. *Behavior Therapy*, 47(5), 702-719.
- [99] Nguyen, J., Brymer, E. (2018). Nature-based guided imagery as an intervention for state anxiety. *Frontiers in Psychology*, *9*, 1858. doi: 10.3389/fpsyg.2018.01858.
- [100] Martyn, P., & Brymer, E. (2014). The relationship between nature relatedness and anxiety. J. Health Psychol. 21, 1436–1445. doi: 10.1177/1359105314555169.
- [101] Holmes, E.A., & Mathews, A. (2010). Mental imagery in emotion and emotional disorders. *Clin. Psychol. Rev.* 30, 349–362. doi: 10.1016/j.cpr.2010.01.001.
- [102] Holmes, P.S., & Collins, D.J. (2001). The PETTLEP approach to motor imagery: a functional equivalence model for sport psychologists. J. Appl. Sport Psychol. 13, 60–83. doi: 10.1080/10413200109339004.
- [103] Apóstolo, J., & Kolcaba, K. (2009). The effects of guided imagery on comfort, depression, anxiety, and stress of psychiatric inpatients with depressive disorders. Arch. Psychiatr. Nurs. 23, 403–411. doi: 10.1016/j.apnu.2008.12.003.
- [104] Kealy, K.L.K., & Arbuthnott, K.D. (2003). Phenomenal characteristics of cocreated guided imagery and autobiographical memories. *Applied Cognitive Psychology*, 17(7), 801–818. https://doi.org/10.1002/acp.910.
- [105] Borst, G., & Kosslyn, S. M. (2008). Visual mental imagery and visual perception: Structural equivalence revealed by scanning processes. *Memory & Cognition*, 36(4), 849–862. https://doi.org/10.3758/MC.36.4.849.
- [106] Arbuthnott, K.D., Geelen, C.B., & Kealy, K.L. (2002). Phenomenal characteristics of guided imagery, natural imagery, and autobiographical memories. *Memory & Cognition*, 30(4), 519-528.
- [107] Segal, S.J., & Fusella, V. (1969). Effects of imaging on signal-to-noise ratio, with varying signal conditions. *British Journal of Psychology*, 60(4), 459– 464. https://doi.org/10.1111/j.2044-8295.1969.tb01219.x
- [108] Baddeley, A.D., & Andrade, J. (2000). Working memory and the vividness of imagery. *Journal of Experimental Psychology: General*, 129(1), 126– 145. https://doi.org/10.1037/0096-3445.129.1.126.

- [109] Ganis, G., Thompson, W.L., & Kosslyn, S.M. (2004). Brain areas underlying visual mental imagery and visual perception: an fMRI study. *Brain Res Cogn Brain Res. 20*(2), 226-41. doi: 10.1016/j.cogbrainres.2004.02.012.
- [110] Ayres, J., & Hopf, T.S. (1985). Visualization: A means of reducing speech anxiety, *Communication Education*, 34(4) 318-323, doi: 10.1080/03634528509378623
- [111] Speck, B.J. (1990). The effect of guided imagery upon first semester nursing students performing their first injections. J Nurs Educ. 29(8), 346-50. doi: 10.3928/0148-4834-19901001-06.
- [112] Stephens, R.L. (1992). Imagery: a treatment for nursing student anxiety. *J Nurs Educ.* 31(7), 314-20. doi: 10.3928/0148-4834-19920901-08.
- [113] Rees, B.L. (1995). Effect of relaxation with guided imagery on anxiety, depression, and self-esteem in primiparas. *J Holist Nurs.* 13(3), 255-67. doi: 10.1177/089801019501300307.
- [114] Casida, J., & Lemanski, S.A. (2010). An evidence-based review on guided imagery utilization in adult cardiac surgery. *Journal of Doctoral Nursing Practice*, *3*(1), 22.
- [115] Thomas, K.M., & Sethares, K.A. (2010). Is guided imagery effective in reducing pain and anxiety in the postoperative total joint arthroplasty patient? *Orthop Nurs.* 29(6), 393-9. doi: 10.1097/NOR.0b013e3181f837f0.
- [116] Vineeta G.K., Bray M.A., & Kehle T.J. (2010). School-based intervention: relaxation and guided imagery for students with asthma and anxiety disorder. *Can. J. Sch. Psychol.* 25, 311–327. 10.1177/0829573510375551.
- [117] Serra, D., Parris, C.R., Carper, E., Fleishman, S., Harrison, L., & Chadha, M. (2012). Outcomes of guided imagery in patients receiving radiation therapy for breast cancer. *Clin. J. Oncol. Nurs.* 16, 617–623. 10.1188/12.CJON.617-623.
- [118] Prabhaa, R.M.R., & Joseph, A. (2020). Intraoperative guided imagery on anxiety. *International Journal of Innovative Science and Research Technolog.* 5(10), 663–668. doi: 10.35940/ijitee.e2641.039520.
- [119] Felix, M.M.D.S., Ferreira, M.B.G., Oliveira, L.F., Barichello, E., Pires S., & Barbosa M.H. (2018). Guided imagery relaxation therapy on preoperative anxiety: a randomized clinical trial. *Revista Latino-Americana De Enfermagem. 6*, e3101. doi: 10.1590/1518-8345.2850.3101.
- [120] Sanadgol, S., Firouzkouhi, M., Badakhsh, M., Abdollahimohammad, A., & Shahraki-vahed, A. (2020). Effect of guided imagery training on death anxiety of nurses at COVID-19 intensive care units: a quasi-experimental study. *Neuropsychiatria I Neuropsychologia/Neuropsychiatry and Neuropsychology*, 15(3), 83–88. doi: 10.5114/nan.2020.101290.

- [121] Eaton, K.W., & Ferrari, T.M. (2020). Heart rate variability during an internal family systems approach to self-forgiveness. *International Journal of Clinical and Experimental Physiology*, 7(2), 52–57. doi: 10.5530/ijcep.2020.7.2.14.
- [122] Baider, L., Uziely, B., & De-Nour, A.K. (1994). Progressive muscle relaxation and guided imagery in cancer patients. *Gen Hosp Psychiatry*, 16(5), 340-7. doi: 10.1016/0163-8343(94)90021-3.
- [123] Nasiri, S., Akbari, H., Tagharrobi, L., & Tabatabaee, A.S. (2018). The effect of progressive muscle relaxation and guided imagery on stress, anxiety, and depression of pregnant women referred to health centers. *J Educ Health Promot.* 7(41). doi: 10.4103/jehp.jehp_158_16.
- [124] Kapogiannis, A., Tsoli, S., & Chrousos, G. (2018). Investigating the Effects of the Progressive Muscle Relaxation-Guided Imagery Combination on Patients with Cancer Receiving Chemotherapy Treatment: A Systematic Review of Randomized Controlled Trials. *Explore (NY), 14*(2), 137-143. doi: 10.1016/j.explore.2017.10.008.
- [125] Tee, V., Kuan, G., Kueh, Y.C., Abdullah, N., Sabran, K., Tagiling, N., Sahran, N.F., Alang, T.A.I.T., Lee, Y.Y. (2022). Development and validation of audio-based guided imagery and progressive muscle relaxation tools for functional bloating. *PLoS One*, 17(9), e0268491. doi: 10.1371/journal.pone.0268491.
- [126] Sinha, M.K., Barman, A., Goyal, M., & Patra, S. (2021). Progressive Muscle Relaxation and Guided Imagery in Breast Cancer: A Systematic Review and Metaanalysis of Randomised Controlled Trials. *Indian J Palliat Care*, 27(2), 336-344. doi: 10.25259/IJPC_136_21.
- [127] Fitzgerald, M., Langevin, M. Imagery. In: Lindquist R, Snyder M, Tracy MF.
 (Ed). Complementary & alternative therapies in nursing. Part II: Mind-body-spirittherapies (pp. 73-98). 7 ed. New York: Springer Publishing; 2014. E-book ISBN: 978-0-8261-9634-7.
- [128] Felix, M.M.D.S., Ferreira, M.B.G., Oliveira, L.F., Barichello, E., Pires, P.D.S., & Barbosa, M.H. (2018). Guided imagery relaxation therapy on preoperative anxiety: a randomized clinical trial. *Rev Lat Am Enfermagem. 29*(26), e3101. doi: 10.1590/1518-8345.2850.3101.
- [129] Alhawatmeh, H.N., Rababa, M., Alfaqih, M., Albataineh, R., Hweidi, I., & Abu Awwad, A. (2022). The Benefits of Mindfulness Meditation on Trait Mindfulness, Perceived Stress, Cortisol, and C-Reactive Protein in Nursing Students: A Randomized Controlled Trial. Adv Med Educ Pract. 13(13), 47-58. doi: 10.2147/AMEP.S348062.
- [130] Polomano, R.C., Fillman, M., Giordano, N.A., Vallerand, A.H., Nicely, K.L.W., & Jungquist, C.R. (2017). Multimodal analgesia for acute postoperative and trauma-related pain. *The American journal of nursing*, 117(3), S12-S26.
- [131] Williams, A.M., Davies, A., & Griffiths, G. (2009). Facilitating comfort for hospitalized patients using non-pharmacological measures: Preliminary development of clinical practice guidelines. *International Journal of Nursing Practice*, 15(3), 145-155.

- [132] Alam, M., Roongpisuthipong, W., Kim, N.A., Goyal, A., Swary, J.H., Brindise, R.T., Iyengar, S., Pace, N., West, D.P., Polavarapu, M., & Yoo, S. (2016). Utility of recorded guided imagery and relaxing music in reducing patient pain and anxiety, and surgeon anxiety, during cutaneous surgical procedures: A single-blinded randomized controlled trial. *J Am Acad Dermatol.* 75(3), 585-589. doi: 10.1016/j.jaad.2016.02.1143.
- [133] Glickman-Simon, R., Tessier, J. (2014). Guided imagery for postoperative pain, energy healing for quality of life, probiotics for acute diarrhea in children, acupuncture for postoperative nausea and vomiting, and animal-assisted therapy for mental disorders. *Explore (NY). 10*(5), 326-9. doi: 10.1016/j.explore.2014.06.012.
- [134] Luberto, C.M., Shinday, N., Song, R., Philpotts, L.L., Park, E.R., Fricchione, G.L., Yeh, G.Y. (2018). A Systematic Review and Meta-analysis of the Effects of Meditation on Empathy, Compassion, and Prosocial Behaviors. *Mindfulness (NY). 9(3)*, 708-724. doi: 10.1007/s12671-017-0841-8.
- [135] Luberto, C.M., Goodman, J.H., Halvorson, B., Wang, A., & Haramati, A. (2020). Stress and Coping Among Health Professions Students During COVID-19: A Perspective on the Benefits of Mindfulness. *Glob Adv Health Med.* 13(9), 2164956120977827. doi: 10.1177/2164956120977827.
- [136] Dolbier, C.L., & Rush, T.E. (2012). Efficacy of abbreviated progressive muscle relaxation in a high-stress college sample. *International Journal of Stress Management*, *19*(1), 48–68. https://doi.org/10.1037/a0027326
- [137] Vancampfort, D., De Hert, M., Knapen, J., Wampers, M., Demunter, H., Deckx, S., Maurissen, K., & Probst, M. (2011). State anxiety, psychological stress and positive well-being responses to yoga and aerobic exercise in people with schizophrenia: a pilot study. *Disabil Rehabil.* 33(8), 684-9. doi: 10.3109/09638288.2010.509458.
- [138] Shapiro, J.R., Pisetsky, E.M., Crenshaw, W., Spainhour, S., Hamer, R.M., Dymek-Valentine, M., & Bulik, C.M. (2008). Exploratory study to decrease postpradial anxiety: Just relax! *International Journal of Eating Disorders*, 41, 728 –733. doi:10.1002/eat.20552.
- [139] McCallie, M.S., Claire, B.S.W., Blum, M.R.N., & Hood, C.J. (2006). Progressive Muscle Relaxation, *Journal of Human Behavior in the Social Environment*, 13(3), 51-66, DOI: 10.1300/J137v13n03_04
- [140] Ponce, A.N., Lorber, W., Paul, J.J., Esterlis, I., Barzvi, A., Allen, G.J., & Pescatello, L.S. (2008). Comparisons of varying dosages of relaxation in a corporate setting: Effects on stress reduction. *International Journal of Stress Management*, 15(4), 396.
- [141] Rausch, S.M., Gramling, S.E., & Auerbach, S.M. (2006). Effects of a single session of large-group meditation and progressive muscle relaxation training on stress reduction, reactivity, and recovery. *International Journal of Stress Management*, 13(3), 273–290. https://doi.org/10.1037/1072-5245.13.3.273

- [142] Morone, N.E., & Greco, C.M. (2007). Mind-body interventions for chronic pain in older adults: a structured review. *Pain Med.* 8(4), 359-75. doi: 10.1111/j.1526-4637.2007.00312.x.
- [143] Nickel, C., Lahmann, C., Tritt, K., Loew, T.H., Rother, W.K., & Nickel, M.K. (2005). Stressed aggressive adolescents benefit from progressive muscle relaxation: A random, prospective, controlled trial. *Stress and Health*, 21(3), 169-175.
- [144] Krajewski, J., Sauerland, M., & Wieland, R. (2011). Relaxation-induced cortisol changes within lunch breaks-an experimental longitudinal worksite field study. *Journal of Occupational and Organizational Psychology*, *84*(2), 382-394.
- [145] Kwekkeboom, K.L., Wanta, B., & Bumpus, M. (2008). Individual difference variables and the effects of progressive muscle relaxation and analgesic imagery interventions on cancer pain. *J Pain Symptom Manage.* 36(6), 604-15. doi: 10.1016/j.jpainsymman.2007.12.011.
- [146] Glantz, K., Durlach, N.I., Barnett, R.C., & Aviles, W.A. (1997). Virtual reality (VR) and psychotherapy: Opportunities and challenges. *Presence: Teleoperators & Virtual Environments*, 6(1), 87-105.
- [147] Riva, G. (Ed.). (1997). Virtual reality in neuro-psycho-physiology: Cognitive, clinical and methodological issues in assessment and rehabilitation.
- [148] Vincelli, F., & Molinari, E. (1998). Virtual reality and imaginative techniques in clinical psychology. *Stud Health Technol Inform.* 58, 67-72. PMID: 10350930.
- [149] https://cbt4panic.org/3-types-of-graduated-exposure/
- [150] Beverly, E., Hommema, L., Coates, K., Duncan, G., Gable, B., Gutman, T., Love, M., Love, C., Pershing, M., & Stevens, N. (2022). A tranquil virtual reality experience to reduce subjective stress among COVID-19 frontline healthcare workers. *PLoS One*, 17(2), e0262703. doi: 10.1371/journal.pone.0262703.
- [151] Riva, G., Botella, C., Baños, R., Mantovani, F., García-Palacios, A., Quero, S., Serino, S., Triberti, S., Repetto, C., Dakanalis, A., Villani, D., Gaggioli, A. (2015). In: *Immersed in Media*. Lombard M., Biocca F., Freeman J., Ijsselsteijn W., Schaevitz R.J., editors. Springer International Publishing; New York. Presence-inducing media for mental health applications.
- [152] Parsons, T.D., Gaggioli, A., & Riva, G. (2017). Virtual reality for research in social neuroscience. *Brain sciences*, 7(4), 42.
- [153] Cruz-Neira, C., Sandin, D.J., DeFanti, T.A., Kenyon, R.V., & Hart, J.C. (1992). The CAVE: audio visual experience automatic virtual environment. *Communications* of the ACM, 35(6), 64-73.

- [154] Steuer, J.S. (1992). Defining virtual reality: Dimensions determining telepresence. *Journal of Communication*, 42(4), 73–93. https://doi.org/10.1111/j.1460-2466.1992.tb00812.x
- [155] Wilkinson, M., Brantley, S., & Feng, J. (2021). A Mini Review of Presence and Immersion in Virtual Reality. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 65(1), 1099– 1103. https://doi.org/10.1177/1071181321651148
- [156] Gaggioli, A. (Ed.). (2009). Advanced technologies in rehabilitation: empowering cognitive, physical, social and communicative skills through virtual reality, robots, wearable systems and brain-computer interfaces (Vol. 145). IOS Press.
- [157] Slater, M. (1999) Measuring Presence: A Response to the Witmer and Singer Presence Questionnaire. *Presence*, 8, 560-565. http://dx.doi.org/10.1162/105474699566477
- [158] Witmer, B.G., & Singer, M.J. (1998). Measuring presence in virtual environments: A presence questionnaire. *Presence: Teleoperators and Virtual Environments*, 7(3), 225–240. https://doi.org/10.1162/105474698565686
- Biocca, F., & Delaney, B. (1995). Immersive virtual reality technology. In F.
 Biocca & M. R. Levy (Eds.), *Communication in the age of virtual reality* (pp. 57–124).
 Lawrence Erlbaum Associates, Inc. (Reprinted in modified form from "Journal of Communication," Aut 1992)
- [160] Lombard, M., Ditton, T. B., Crane, D., Davis, B., Gil-Egui, G., Horvath, K., (...) & Park, S. (2000, March). Measuring presence: A literature-based approach to the development of a standardized paper-and-pencil instrument. In *Third international* workshop on presence, delft, the netherlands (Vol. 240, pp. 2-4).
- [161] IJsselsteijn, W., & Riva, G. (2003). Being there: The experience of presence in mediated environments. In G. Riva, F. Davide, & W. A. IJsselsteijn (Eds.), *Being there: Concepts, effects and measurements of user presence in synthetic environments* (pp. 3– 16). IOS Press.
- [162] Akin, D.L., Minsky, M.L., Thiel, E.D., & Kurtzman, C.R. (1983). Space applications of automation, robotics and machine intelligence systems (ARAMIS), phase 2. volume 1: Telepresence technology base development (No. NAS 1.26: 3734).
- [163] Slater, M. (2002). Presence and the sixth sense. *Presence*, 11(4), 435-439.
- [164] Slater, M., Steed, A., McCarthy, J., & Maringelli, F. (1998). The influence of body movement on subjective presence invirtual environments. Human Factors, 40(3), 469–477.
- [165] Sheridan, T.B. (1992). Musings on Telepresence and Virtual Presence. *Presence: Teleoperators and Virtual Environments*, 1 (1), 120–126. doi: https://doi.org/10.1162/pres.1992.1.1.120.

- [166] Murray, C.D., Fox, J., & Pettifer, S. (2007) Absorption, dissociation, locus of control and presence in virtual reality. *Computers in Human Behavior*, 23(3), 1347-1354.
- [167] Mantovani, G., & Riva, G. (1999). "Real" presence: how different ontologies generate different criteria for presence, telepresence, and virtual presence. *Presence*, 8(5), 540-550.
- [168] Schubert, T.W. (2003). The sense of presence in virtual environments: A threecomponent scale measuring spatial presence, involvement, and realness. Z. für Medienpsychologie, 15, 69-71.
- [169] Lessiter, J., Freeman, J., Keogh, E., & Davidoff, J. (2001). A cross-media presence questionnaire: The ITC-Sense of Presence Inventory. *Presence: Teleoperators & Virtual Environments*, 10(3), 282-297.
- [170] Skarbez, R., Brooks, Jr, F. P., & Whitton, M. C. (2017). A survey of presence and related concepts. *ACM Computing Surveys (CSUR)*, *50*(6), 1-39.
- [171] Nisbett, R.E., & Wilson, T.D. (1977). Telling more than we can know: Verbal reports on mental processes. *Psychological Review*, 84(3), 231– 259. https://doi.org/10.1037/0033-295X.84.3.231
- [172] Hendrix, C., & Barfield, W. (1996). Presence within virtual environments as a function of visual display parameters. *Presence: Teleoperators and Virtual Environments*, 5(3), 274–289. https://doi.org/10.1162/pres.1996.5.3.274
- [173] Freeman, D., Slater, M., Bebbington, P. E., Garety, P. A., Kuipers, E., Fowler, D., (...) & Vinayagamoorthy, V. (2003). Can virtual reality be used to investigate persecutory ideation?. *The Journal of nervous and mental disease*, 191(8), 509-514.
- [174] Coelho, H., Monteiro, P., Gonçalves, G., Melo, M., & Bessa, M. (2022). Authoring tools for virtual reality experiences: A systematic review. *Multimedia Tools and Applications*, *81*(19), 28037-28060.
- [175] Witmer, B.G., Jerome, C.J., & Singer, M.J. (2005). The factor structure of the presence questionnaire. *Presence: Teleoperators & Virtual Environments*, 14(3), 298-312.
- [176] Slater, M., Usoh, M., & Steed, A. (1995). Taking steps: The influence of a walking technique on presence in virtual reality. *ACM Transactions on Computer Human Interaction (TOCHI)*, 2(3), 201-219.
- [177] Usoh, M., Arthur, K., Whitton, M., Bastos, R., Steed, A., Slater, M., & Brooks,
 F. (1999). Walking > walking-in-place > flying in virtual environments. In Proc. of
 ACM SIGGRAPH 99. ACM Press/ ACM SIGGRAPH.
- Baños, R. M., Botella, C., Garcia-Palacios, A., Villa, H., Perpiñá, C., & Alcaniz, M. (2000). Presence and reality judgment in virtual environments: a unitary construct?. *CyberPsychology & Behavior*, 3(3), 327-335.

- [179] Schubert, T., Friedmann, F., Regenbrecht, H. (2001). The Experience of Presence: Factor Analytic Insights. *Presence: Teleoperators and Virtual Environments*, 10(3), 266–281. doi: https://doi.org/10.1162/105474601300343603
- [180] Kalawsky, R. (1998) A tool for evaluation of contributory factors associated withpresence in spatially immersive environments. Paper presented at the BT Presence Workshop,10-11 June 1998.
- [181] Berto, R. (2014). The role of nature in coping with psycho-physiological stress: A literature review on restorativeness. *Behavioral sciences*, 4(4), 394-409.
- [182] Serrano, B., Baños, R.M., & Botella, C. (2016). Virtual reality and stimulation of touch and smell for inducing relaxation: A randomized controlled trial. *Computers in Human Behavior*, 55, 1-8.
- [183] Anderson, A.P., Mayer, M.D., Fellows, A.M., Cowan, D.R., Hegel, M.T., & Buckey, J.C. (2017). Relaxation with immersive natural scenes presented using virtual reality. *Aerospace medicine and human performance*, 88(6), 520-526.
- [184] Riches, S., Azevedo, L., Bird, L., Pisani, S., & Valmaggia, L. (2021). Virtual reality relaxation for the general population: a systematic review. *Soc Psychiatry Psychiatr Epidemiol.* 56(10), 1707-1727. doi: 10.1007/s00127-021-02110-z.
- [185] Riches, S., Jeyarajaguru, P., Taylor, L., Fialho, C., Little, J., Ahmed, L., O'Brien, A., van Driel, C., Veling, W., & Valmaggia, L. (2023). Virtual reality relaxation for people with mental health conditions: a systematic review. *Soc Psychiatry Psychiatr Epidemiol.* 58(7), 989-1007. doi: 10.1007/s00127-022-02417-5.
- [186] Riches, S., & Smith, H. (2022). Taking a break in the 'new normal': virtual reality relaxation for a stressed workforce. *Mental Health Rev. J.* 27(2).
- [187] Gaggioli, A., et al. (2014).
 Experiential Virtual Scenarios With Real-Time Monitoring (Interreality) for the Management of Psychological Stress: A Block Randomized Controlled Trial J. Med. Internet Res. 16(7), e167 doi: 10.2196/jmir.3235
- [188] Pretsch, J., Pretsch, E., Saretzki, J., Kraus, H., & Grossmann, G. (2021). Improving employee well-being by means of virtual reality–REALEX: an empirical case study.
- [189] Riva, G., Mantovani, F., Capideville, C.S., Preziosa, A., Morganti, F., Villani, D., Gaggioli, A., Botella, C., & Alcañiz, M. (2007). Affective interactions using virtual reality: the link between presence and emotions. *Cyberpsychol Behav.* 10(1), 45-56. doi: 10.1089/cpb.2006.9993.
- [190] Bittner, L., Mostajeran, F., Steinicke, F., Gallinat, J., & Kühn, S. (2018). Evaluation of flowvr: A virtual reality game for improvement of depressive mood. *Biorxiv*, 451245.

- [191] Liszio, S., Graf, L., & Masuch, M. (2018). The relaxing effect of virtual nature: Immersive technology provides relief in acute stress situations. *Annual Review of CyberTherapy and Telemedicine*, *16*, 87–93.
- [192] Seabrook, E., Kelly, R., Foley, F., Theiler, S., Thomas, N., Wadley, G., & Nedeljkovic, M. (2020). Understanding How Virtual Reality Can Support Mindfulness Practice: Mixed Methods Study. *J Med Internet Res. 22*(3), e16106. doi: 10.2196/16106.
- [193] Gross, J.J., & Levenson, R.W. (1995). Emotion elicitation using films. Cognition and Emotion, 9(1), 87– 108. https://doi.org/10.1080/02699939508408966.
- [194] Pavic, K., Vergilino-Perez, D., Gricourt, T., & Chaby, L. (2022). Because I'm happy—an overview on fostering positive emotions through virtual reality. *Frontiers in Virtual Reality*, *3*, 788820.
- [195] Browning, M.H.E.M., Mimnaugh, K.J., van Riper, C.J., Laurent, H.K., LaValle, S.M. (2020). Can Simulated Nature Support Mental Health? Comparing Short, Single-Doses of 360-Degree Nature Videos in Virtual Reality With the Outdoors. *Front Psychol.* 10, 2667. doi: 10.3389/fpsyg.2019.02667.
- [196] Mattila, O., Korhonen, A., Pöyry, E., Hauru, K., Holopainen, J. & Parvinen, P. (2020). Restoration in a Virtual Reality Forest Environment. *Computers in Human Behavior*, 107, 106295. https://doi.org/10.1016/j.chb.2020.106295.
- [197] Villani, D., & Riva, G. (2012). Does interactive media enhance the management of stress? Suggestions from a controlled study. *Cyberpsychology, Behavior, and Social Networking*, 15(1), 24–30. https://doi.org/10.1089/cyber.2011.0141
- [198] Annerstedt, M., Jönsson, P., Wallergård, M., Johansson, G., Karlson, B., Grahn, P., Hansen, A.M., & Währborg, P. (2013). Inducing physiological stress recovery with sounds of nature in a virtual reality forest--results from a pilot study. *Physiol Behav.* 118, 240-50. doi: 10.1016/j.physbeh.2013.05.023.
- [199] Chirico, A., & Gaggioli, A. (2019). When virtual feels real: Comparing emotional responses and presence in virtual and natural environments. *Cyberpsychology, Behavior, and Social Networking, 22*(3), 220– 226. https://doi.org/10.1089/cyber.2018.0393.
- [200] Valtchanov, D., Barton, K.R., & Ellard, C. (2010). Restorative effects of virtual nature settings. *Cyberpsychol Behav Soc Netw.* 13(5), 503-12. doi: 10.1089/cyber.2009.0308.
- [201] Chirico, A., Clewis, R. R., Yaden, D. B., & Gaggioli, A. (2021). Nature versus art as elicitors of the sublime: A virtual reality study. *PloS one*, *16*(3), e0233628.
- [202] Yu, C.P., Lee, H.Y., & Luo, X.Y. (2018). The effect of virtual reality forest and urban environments on physiological and psychological responses. *Urban forestry & urban greening*, *35*, 106-114.

- [203] Kern, A.C., & Ellermeier, W. (2020). Audio in VR: Effects of a soundscape and movement-triggered step sounds on presence. *Frontiers in Robotics and AI*, 7, 20.
- [204] Watson, D., Clark, L.A., & Tellegen, A. (1988). Development and validation of brief measures of positive and negative affect: the PANAS scales. *J Pers Soc Psychol.* 54(6), 1063-70. doi: 10.1037//0022-3514.54.6.1063.
- [205] Baños, R.M., Espinoza, M., García-Palacios, A., Cervera, J.M., Esquerdo, G., Barrajón, E., & Botella, C. (2013). A positive psychological intervention using virtual reality for patients with advanced cancer in a hospital setting: a pilot study to assess feasibility. *Support Care Cancer*, 21(1), 263-70. doi: 10.1007/s00520-012-1520-x.
- [206] Felnhofer, A., Kothgassner, O. D., Schmidt, M., Heinzle, A. K., Beutl, L., Hlavacs, H., & Kryspin-Exner, I. (2015). Is virtual reality emotionally arousing? Investigating five emotion inducing virtual park scenarios. *International journal of human-computer studies*, 82, 48-56.
- [207] Chirico, A., Cipresso, P., Yaden, D. B., Biassoni, F., Riva, G., & Gaggioli, A. (2017). Effectiveness of immersive videos in inducing awe: an experimental study. *Scientific reports*, 7(1), 1218.
- [208] Marín-Morales, J., et al. (2018). Affective computing in virtual reality: emotion recognition from brain and heartbeat dynamics using wearable sensors. *Scientific reports*, 8(1), 13657.
- [209] Malbos, E., Chichery, N., Borwell, B., Seimandi, J., Weindel, G., & Lancon, C. (2020). Virtual reality for relaxation in the treatment of generalized anxiety disorder: a comparative trial. *Annu Rev Cyberther Telemed*, 18, 183-187.
- [210] Mistry, D., Zhu, J., Tremblay, P., Wekerle, C., Lanius, R., Jetly, R., & Frewen, P. (2020). Meditating in virtual reality: Proof-of-concept intervention for posttraumatic stress. *Psychological Trauma: Theory, Research, Practice, and Policy*, 12(8), 847.
- [211] Shah, L.B.I., Torres, S., Kannusamy, P., Chng, C.M.L., He, H.G., & Klainin-Yobas, P. (2015). Efficacy of the virtual reality-based stress management program on stress-related variables in people with mood disorders: the feasibility study. *Archives* of psychiatric nursing, 29(1), 6-13.
- [212] Veling, W., Lestestuiver, B., Jongma, M., Hoenders, H.J.R., van Driel, C. (2021). Virtual Reality Relaxation for Patients With a Psychiatric Disorder: Crossover Randomized Controlled Trial. *J Med Internet Res.* 23(1), e17233. doi: 10.2196/17233.
- [213] Geraets, C.N.W., van der Stouwe, E.C.D., Pot-Kolder, R., Veling, W. (2021). Advances in immersive virtual reality interventions for mental disorders: A new reality? *Curr Opin Psychol.* 41, 40-45. doi: 10.1016/j.copsyc.2021.02.004.
- [214] Bossenbroek, R., Wols, A., Weerdmeester, J., Lichtwarck-Aschoff, A., Granic, I., & van Rooij, M. M. (2020). Efficacy of a virtual reality biofeedback game (DEEP) to reduce anxiety and disruptive classroom behavior: single-case study. *JMIR mental health*, 7(3), e16066.

- [215] Jeong, H. S., Oh, J., Paik, M., Kim, H., Jang, S., Kim, B. S., & Kim, J. J. (2022). Development and feasibility assessment of virtual reality-based relaxation self-training program. *Frontiers in Virtual Reality*, 2, 722558.
- [216] Tan, H.L.E., Chng, C.M.L., Lau, Y., & Klainin-Yobas, P. (2021). Investigating the effects of a virtual reality-based stress management programme on inpatients with mental disorders: A pilot randomised controlled trial. *International Journal of Psychology*, 56(3), 444-453.
- [217] Diener, E., & Chan, M. Y. (2011). Happy people live longer: Subjective wellbeing contributes to health and longevity. *Applied Psychology: Health and Well-Being*, 3(1), 1–43. https://doi.org/10.1111/j.1758-0854.2010.01045.x
- [218] Kalantari, S., Bill Xu, T., Mostafavi, A., Lee, A., Barankevich, R., Boot, W.R., & Czaja, S.J. (2022). Using a Nature-Based Virtual Reality Environment for Improving Mood States and Cognitive Engagement in Older Adults: A Mixed-Method Feasibility *Study. Innov Aging*, 6(3), igac015. doi: 10.1093/geroni/igac015.
- [219] Brimelow, R.E., Dawe, B., & Dissanayaka, N. (2020). Preliminary research: virtual reality in residential aged care to reduce apathy and improve mood. *Cyberpsychology, Behavior, and Social Networking*, 23(3), 165-170.
- [220] Brimelow, R.E., Thangavelu, K., Beattie, R., & Dissanayaka, N.N. (2022). Feasibility of group-based multiple virtual reality sessions to reduce behavioral and psychological symptoms in persons living in residential aged care. *Journal of the American Medical Directors Association*, 23(5), 831-837.
- [221] Moyle, W., Jones, C., Dwan, T., & Petrovich, T. (2018). Effectiveness of a virtual reality forest on people with dementia: A mixed methods pilot study. *The Gerontologist*, 58(3), 478-487.
- [222] Liu, Q., Wang, Y., Tang, Q., & Liu, Z. (2020). Do you feel the same as i do? differences in virtual reality technology experience and acceptance between elderly adults and college students. *Frontiers in Psychology*, *11*, 573673.
- [223] Etchemendy, E., Baños, R. M., Botella, C., Castilla, D., Alcañíz, M., Rasal, P., & Farfallini, L. (2011). An e-health platform for the elderly population: The butler system. *Computers & Education*, 56(1), 275-279.
- [224] Huygelier, H., Schraepen, B., Van Ee, R., Vanden Abeele, V., & Gillebert, C.
 R. (2019). Acceptance of immersive head-mounted virtual reality in older adults. *Scientific reports*, 9(1), 4519.
- [225] Navarro-Haro, M.V., et al. (2017). Meditation experts try Virtual Reality Mindfulness: A pilot study evaluation of the feasibility and acceptability of Virtual Reality to facilitate mindfulness practice in people attending a Mindfulness conference. *PloS one*, *12*(11), e0187777.

- [226] Chandrasiri, A., Collett, J., Fassbender, E., & De Foe, A. (2020). A virtual reality approach to mindfulness skills training. *Virtual Reality*, *24*, 143-149.
- [227] Seabrook, E., Kelly, R., Foley, F., Theiler, S., Thomas, N., Wadley, G., & Nedeljkovic, M. (2020). Understanding how virtual reality can support mindfulness practice: mixed methods study. *Journal of medical Internet research*, *22*(3), e16106.
- [228] Triberti, S., & Barello, S. (2016). The quest for engaging AmI: Patient engagement and experience design tools to promote effective assisted living. *Journal of biomedical informatics*, 63, 150-156.
- [229] Tusek, D.L., & Cwynar, R.E. (2000). Strategies for implementing a guided imagery program to enhance patient experience. *AACN Clin Issues*. 11(1), 68-76. doi: 10.1097/00044067-200002000-00009.
- [230] Vempati, R.P., & Telles, S. (2002). Yoga-based guided relaxation reduces sympathetic activity judged from baseline levels. *Psychol Rep. 90*(2), 487-94. doi: 10.2466/pr0.2002.90.2.487.
- [231] Xie, B., et al. (2021). A review on virtual reality skill training applications. *Frontiers in Virtual Reality*, *2*, 645153.
- [232] Palazzo, C., et al., (2016). Barriers to home-based exercise program adherence with chronic low back pain: Patient expectations regarding new technologies. *Annals of physical and rehabilitation medicine*, *59*(2), 107-113.
- [233] Pardini, S., Gabrielli, S., Dianti, M., Novara, C., Zucco, G.M., Mich, O., Forti, S. (2022). The Role of Personalization in the User Experience, Preferences and Engagement with Virtual Reality Environments for Relaxation. *Int J Environ Res Public Health*. 19(12), 7237. doi: 10.3390/ijerph19127237.
- [234] Li, M., Wang, L., Jiang, M., Wu, D., Tian, T., Huang, W. (2020). Relaxation techniques for depressive disorders in adults: a systematic review and meta-analysis of randomised controlled trials. *Int J Psychiatry Clin Pract.* 24(3), 219-226. doi: 10.1080/13651501.2020.1764587.
- [235] Ussher, M., Spatz, A., Copland, C., Nicolaou, A., Cargill, A., Amini-Tabrizi, N., McCracken, L.M. (2014). Immediate effects of a brief mindfulness-based body scan on patients with chronic pain. *J Behav Med.* 37(1), 127-34. doi: 10.1007/s10865-012-9466-5.
- [236] Schultchen, D., Messner, M., Karabatsiakis, A. et al. (2019). Effects of an 8-Week Body Scan Intervention on Individually Perceived Psychological Stress and Related Steroid Hormones in Hair. *Mindfulness*, 10, 2532–2543. https://doi.org/10.1007/s12671-019-01222-7.
- [237] Wren, A.A., et al., (2021). Mindfulness-based virtual reality intervention for children and young adults with inflammatory bowel disease: A pilot feasibility and acceptability study. *Children*, 8(5), 368.

- [238] Zhang, Y., Liu, H., Kang, S.C., & Al-Hussein, M. (2020). Virtual reality applications for the built environment: Research trends and opportunities. *Automation in Construction*, *118*, 103311.
- [239] Sutcliffe, A., & Gault, B. (2004). Heuristic evaluation of virtual reality applications. *Interacting with Computers*, 16(4), 831–849, https://doi.org/10.1016/j.intcom.2004.05.001.
- [240] Stanney, K.M., Hale, K.S., Nahmens, I., & Kennedy, R.S. (2003). What to expect from immersive virtual environment exposure: Influences of gender, body mass index, and past experience. *Human factors*, *45*(3), 504-520.
- [241] Czajkowski, S.M., Powell, L.H., Adler, N., Naar-King, S., Reynolds, K.D., Hunter, C.M., Laraia, B., Olster, D.H., Perna, F.M., Peterson, J.C., Epel, E., Boyington, J.E., & Charlson, M.E. (2015). From ideas to efficacy: The ORBIT model for developing behavioral treatments for chronic diseases. *Health Psychol.* 34(10), 971-82. doi: 10.1037/hea0000161.
- [242] Serrano, B., Baños, R. M., & Botella, C. (2016). Virtual reality and stimulation of touch and smell for inducing relaxation: A randomized controlled trial. *Computers in Human Behavior*, 55, 1-8.
- [243] White, M.P., et al. (2021). Associations between green/blue spaces and mental health across 18 countries. *Sci Rep. 11*, 8903. https://doi.org/10.1038/s41598-021-87675-0.
- [244] Kabat-Zinn, J. (2003). Mindfulness-based interventions in context: Past, present, and future. *Clinical Psychology: Science and Practice*, 10(2), 144–156. https://doi.org/10.1093/clipsy.bpg016.
- [245] Spielberger, C.D. (1989). *State-Trait Anxiety Inventory: Bibliography* (2nd ed.). Palo Alto, CA: Consulting Psychologists Press.
- [246] Spielberger, C.D., Gorsuch, R.L., Lushene, R., Vagg, P.R., & Jacobs, G.A. (1983). *Manual for the State-Trait Anxiety Inventory*. Palo Alto, CA: Consulting Psychologists Press.
- [247] Lovibond, P.F., & Lovibond, SH. (1995). The structure of negative emotional states: comparison of the Depression Anxiety Stress Scales (DASS) with the Beck Depression and Anxiety Inventories. *Behav Res Ther.* 33(3), 335-43. doi: 10.1016/0005-7967(94)00075-u.
- [248] Bottesi, G., Ghisi, M., Altoè, G., Conforti, E., Melli, G., & Sica, C. (2015). The Italian version of the Depression Anxiety Stress Scales-21: Factor structure and psychometric properties on community and clinical samples. *Compr Psychiatry*. 60, 170-81. doi: 10.1016/j.comppsych.2015.04.005.
- [249] Ames, S.L., Wolffsohn, J.S., & McBrien, N.A. (2005). The development of a symptom questionnaire for assessing virtual reality viewing using a head-mounted display. *Optom Vis Sci.* 82(3), 168-76. doi: 10.1097/01.opx.0000156307.95086.6.

- [250] Bradley, M.M., & Lang, P.J. (1994). Measuring emotion: The Self-Assessment Manikin and the semantic differential. *Journal of Behavior Therapy and Experimental Psychiatry*, 25(1), 49–59. https://doi.org/10.1016/0005-7916(94)90063-9.
- [251] Morris, J.D. (1995). Observations: SAM: The self-assessment manikin: An efficient cross-cultural measurement of emotional response. *Journal of Advertising Research*, 35(6), 63–68.
- [252] IBM Corp. Released 2020. IBM SPSS Statistics for Windows, Version 27.0. Armonk, NY: IBM Corp.
- [253] George, D., & Mallery, P. (2010). SPSS for Windows step by step : A simple guide and reference, 17.0 update. Allyn & Bacon. Gomez, M. A., & Catan, M. M. (2021). Factors leading to limited researches conducted by Philippine public school teachers. *Innovare Journal of Education*, 9(3), 1–7. https://doi.org/10.22159/ijoe.2021v9i3.41272.
- [254] Khan, A.F. Ph.D Thesis. Aligarh Muslim University; Aligarh, India: 2015. Assessment of Midlife career Stress on Indian Managers.
- V., & Clarke, V. (2006). thematic [255] Braun, Using analysis in psychology. *Qualitative* 77-Research in Psychology, 3(2),101. https://doi.org/10.1191/1478088706qp063oa.
- [256] Wilkinson, M., Brantley, S., & Feng, J. (2021). A mini review of presence and immersion in virtual reality. *Proc. Hum. Factors Ergon. Soc. Annu. Meet.* 65, 1099– 1103. doi: 10.1177/1071181321651148.
- [257] Villani, D., Riva, F., & Riva, G. (2007). New technologies for relaxation: The role of presence. *International Journal of Stress Management*, 14(3), 260– 274. https://doi.org/10.1037/1072-5245.14.3.260
- [258] Rotter, J.B. (1966). Generalized expectancies for internal versus external control of reinforcement. *Psychol Monogr.* 80(1), 1-28.
- [259] Iachini, T., Maffei, L., Masullo, M., Senese, V.P., Rapuano, M., Pascale, A., Sorrentino, F., & Ruggiero, G. (2019). The experience of virtual reality: are individual differences in mental imagery associated with sense of presence? *Cogn Process.* 20(3), 291-298. doi: 10.1007/s10339-018-0897-y.
- [260] Pardini, S., Gabrielli, S., Olivetto, S., Fusina, F., Dianti, M., Forti, S., Lancini, C., & Novara, C. (2023). Personalized, Naturalistic Virtual Reality Scenarios Coupled With Web-Based Progressive Muscle Relaxation Training for the General Population: Protocol for a Proof-of-Principle Randomized Controlled Trial. *JMIR Res Protoc.* 12, e44183. doi: 10.2196/44183.
- [261] Bayram, N. & Bilgel, N. (2008). The Prevalence and Socio-Demographic Correlations of Depression, Anxiety and Stress among a Group of University Students. *Social Psychiatry and Psychiatric Epidemiology*, 43, 667-672. http://dx.doi.org/10.1007/s00127-008-0345-x.

- [262] Wiegner, L., Hange, D., Björkelund, C., & Ahlborg, G. Jr. (2015). Prevalence of perceived stress and associations to symptoms of exhaustion, depression and anxiety in a working age population seeking primary care--an observational study. *BMC Fam Pract, 19*, 16-38. doi: 10.1186/s12875-015-0252-7.
- [263] Son, C., Hegde, S., Smith, A., Wang, X., & Sasangohar, F. (2020). Effects of COVID-19 on College Students' Mental Health in the United States: Interview Survey Study. *J Med Internet Res.* 22(9), e21279. doi: 10.2196/21279.
- [264] Komariah, M., Ibrahim, K., Pahria, T., Rahayuwati, L., & Somantri, I. (2023). Effect of Mindfulness Breathing Meditation on Depression, Anxiety, and Stress: A Randomized Controlled Trial among University Students. *Healthcare*, 11(1), 26. https://doi.org/10.3390/healthcare11010026.
- [265] Ozamiz-Etxebarria, N., Santa María, M.D., Munitis, A.E., & Gorrotxategi, M.P. (2020). Reduction of COVID-19 Anxiety Levels Through Relaxation Techniques: A Study Carried Out in Northern Spain on a Sample of Young University Students. *Front Psychol.* 11, 2038. doi: 10.3389/fpsyg.2020.02038.
- [266] Lang, P.J. (1979). A bio-informational theory of emotional imagery. *Psychophysiology*, *16*(6), 495–512. doi: 10.1111/j.1469-8986.1979.tb01511.x.
- [267] Benson, H., Beary, J.F., & Carol, M.P. (1974). The relaxation response. *Psychiatry: Journal for the Study of Interpersonal Processes*, 37(1), 37–46. PMID:4810622.
- [268] Browning, M.H.E.M., Shin, S., Drong, G. et al. (2023). Daily exposure to virtual nature reduces symptoms of anxiety in college students. Sci Rep. 13, 1239. https://doi.org/10.1038/s41598-023-28070-9.
- [269] Syrjala, K.L., & Abrams, J.R. Hypnosis and imagery in the treatment of pain. In: Gatchel RJ, Turk DC, editors. Psychological approaches to pain management: A practitioner's handbook. New York, NY: Guilford Press; 1996:231-258. ISBN: 9780898622928.
- [270] O'Neill, L.M., Barnier, A.J., & McConkey, K. Treating anxiety with selfhypnosis and relaxation. *Contemporary Hypnosis* 16(23), 68-80 (1999). doi: 10.1002/ch.154.
- [271] Arena, J.G., & Blanchard, E.B. Biofeedback and relaxation therapy for chronic pain disorders. In: Gatchel RJ, Turk DC, editors. Psychological approaches to pain management: A practitioner's handbook. New York, NY: Guilford Press; 1996:179-230. ISBN: 9780898622928.
- [272] Bakke, A.C., Purtzer, M.Z., & Newton, P. The effect of hypnotic-guided imagery on psychological well-being and immune function in patients with prior breast cancer. *Journal of Psychosomatic Research* 15, 1131-1137 (2002). doi: 10.1016/S0022-3999(02)00409-9.

- [273] Frazier, P., Liu, Y., Selvey, A., Meredith, L., & Nguyen-Feng, V.N. (2023). Randomized controlled trials assessing efficacy of brief web-based stress management interventions for college students during the COVID pandemic. *J Couns Psychol. 23*. doi: 10.1037/cou0000652.
- [274] Kemper, K.J., Lynn, J., & Mahan, J.D. (2015). What Is the Impact of Online Training in Mind-Body Skills? *J Evid Based Complementary Altern Med.* 20(4), 275-82. doi: 10.1177/2156587215580882.
- [275] Jung, Y.H., Ha, T.M., Oh, C.Y., Lee, U.S., Jang, J.H., Kim, J., Park, J.O., & Kang, D.H. (2016). The Effects of an Online Mind-Body Training Program on Stress, Coping Strategies, Emotional Intelligence, Resilience and Psychological State. *PLoS One, 11(*8), e0159841. doi: 10.1371/journal.pone.0159841.
- [276] Pizzoli, S.F.M., Marzorati, C., Mazzoni, D., & Pravettoni, G. (2020). Web-Based Relaxation Intervention for Stress During Social Isolation: Randomized Controlled Trial. *JMIR Ment Health*, 7(12), e22757. doi: 10.2196/22757.
- [277] Vincelli, F., & Riva, G. (2000). Virtual reality as a new imaginative tool in psychotherapy. In *Medicine Meets Virtual Reality 2000* (pp. 356-358). IOS Press.
- [278] Muhaiyuddin, N.D.M., & Rambli, D.R.A. (2018). An Interactive Image-based Virtual Reality Application for Guided Imagery Therapy. *Journal of Telecommunication, Electronic and Computer Engineering*, 10(2), 1-8.
- [279] Ioannou, A., Paikousis, L., Papastavrou, E., Avraamides, M.N., Astras, G., & Charalambous, A. (2022). Effectiveness of Virtual Reality Vs Guided Imagery on mood changes in cancer patients receiving chemotherapy treatment: A crossover trial. *Eur J* Oncol Nurs. 61, 102188. doi: 10.1016/j.ejon.2022.102188.
- [280] Grassini, S., Laumann, K., & Rasmussen Skogstad, M. (2020). The use of virtual reality alone does not promote training performance (but sense of presence does). *Frontiers in psychology*, *11*, 1743.
- [281] Ling, Y., Nefs, H. T., Morina, N., Heynderickx, I., & Brinkman, W. P. (2014). A meta-analysis on the relationship between self-reported presence and anxiety in virtual reality exposure therapy for anxiety disorders. *PloS one*, *9*(5), e96144.
- [282] Sica, C., Magni, C., Ghisi, M., Altoè, G., Sighinolfi, C., Chiri, L.R., & Franceschini, S. (2008). Coping Orientation to Problems Experienced-Nuova Versione Italiana (COPE-NVI): uno strumento per la misura degli stili di coping. *Psicoterapia cognitiva e comportamentale*, 14(1), 27-53.
- [283] Marks, D.F. (1989). Construct validity of the vividness of visual imagery questionnaire. *Percept Mot Skills*, 69(2), 459–465. https://doi.org/10.2466/pms.1989.69.2.459.
- [284] Antonietti, A., & Crespi, M. (1995). Analisi di tre questionari per la valutazione della vividezza dell'immagine mentale (Analysis of three questionnaires for assessing

the vividness of mental image). Department of Psychology, Catholic University of The Sacred Heart. Technical report.

- [285] Basso, J.C., McHale, A., Ende, V., Oberlin, D.J., & Suzuki, W.A. (2019). Brief, daily meditation enhances attention, memory, mood, and emotional regulation in nonexperienced meditators. *Behav Brain Res.* 356, 208-220. doi: 10.1016/j.bbr.2018.08.023.
- [286] Rooks, J.D., Morrison, A.B., Goolsarran, M. et al. (2017). We Are Talking About Practice: the Influence of Mindfulness vs. Relaxation Training on Athletes' Attention and Well-Being over High-Demand Intervals. *J Cogn Enhanc. 1*, 141–153 (2017). https://doi.org/10.1007/s41465-017-0016-5.
- [287] Faul, F., Erdfelder, E., Buchner, A., & Lang, A.G. (2009). Statistical power analyses using G*Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*. 41, 1149–1160. doi: 10.3758/BRM.41.4.1149
- [288] https://glimmpse.samplesizeshop.org/design/CALCULATE
- [289] Dupuy, H.J. The Psychological General Well-being (PGWB) Index. In: Wenger NK, Mattson ME, Furburg CD, Elinson J, editors. Assessment of Quality of Life in Clinical Trials of Cardiovascular Therapies. New York, NY: Le Jacq Publishing; 1984:170-83. ISBN: 09377716227.
- [290] Grossi, E., Mosconi, P., Groth, N., Niero, M., & Apolone, G. (2002). Il Questionario Psychological General Well Being. Questionario per la valutazione dello stato generale di benessere psicologico. Versione Italiana. Milano, Mi: Istituto di Ricerche Farmacologiche "Mario Negri".
- [291] Grossi, E., Groth, N., Mosconi, P., Cerutti, R., Pace, F., Compare, A., & Apolone, G. (2006). Development and validation of the short version of the Psychological General Well-Being Index (PGWB-S). *Health Qual Life Outcomes*. 4(88). doi 10.1186/1477-7525-4-88.
- [292] Chirico, A. et al. (2020). Virtual reality and music therapy as distraction interventions to alleviate anxiety and improve mood states in breast cancer patients during chemotherapy. *J Cell Physiol.* 235(6), 5353-5362. doi: 10.1002/jcp.29422.
- [293] IBM Corp. Released 2022. IBM SPSS Statistics for Windows, Version 29.0. Armonk, NY: IBM Corp.
- [294] R Core Team (2021). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.Rproject.org/.
- [295] Wickham et al., (2019). Welcome to the Tidyverse. Journal of Open Source Software, 4(43), 1686. https://doi.org/10.21105/joss.01686
- [296] Wickham, H., & Francois, R. (2016). dplyr: A Grammar of Data Manipulation. https://cran.r-project.org/web/packages/dplyr/index.html

- [297] Lenth, R.V. (2016). Least-Squares Means: The R Package Ismeans. *Journal of Statistical Software*, 69(1), 1-33. https://doi.org/10.18637/jss.v069.i01
- [298] Kuznetsova, A., Brockhoff, P. B., & Christensen, R. H. B. (2016). ImerTest: Tests in Linear Mixed Effects Models. https://doi.org/10.18637/jss.v082.i13.
- [299] Tremblay, A., & Ransijn, J. (2013). LMERConvenienceFunctions: A suite of functions to back-fit fixed effects and forward-fit random effects, as well as other miscellaneous functions. *R package version 2*, 919-931. https://cran.rproject.org/web/packages/LMERConvenienceFunctions/LMERConvenienceFunction s.pdf
- [300] Wickham, H. (2009). ggplot2: Elegant Graphics for Data Analysis (2nd Ed.). New York, NY: Springer-Verlag. https://doi.org/10.1007/978-0-387-98141-3
- [301] Krueger, C., & Tian, L. (2004). A comparison of the general linear mixed model and repeated measures ANOVA using a dataset with multiple missing data points. *Biological research for nursing*, 6(2), 151-157. https://doi.org/10.1177/1099800404267682.
- [302] Muth, C., Bales, K. L., Hinde, K., Maninger, N., Mendoza, S. P., & Ferrer, E. (2016). Alternative models for small samples in psychological research: applying linear mixed effects models and generalized estimating equations to repeated measures data. *Educational and Psychological Measurement*, 76(1), 64-87. https://doi.org/10.1177/0013164415580432.
- [303] Boisgontier, M.P., & Cheval, B. (2016). The anova to mixed model transition. *Neuroscience* & *Biobehavioral Reviews*, 68, 1004-1005. https://doi.org/10.1016/j.neubiorev.2016.05.034.
- [304] Luke, S.G. (2016). Evaluating significance in linear mixed-effects models in R. Behavior Research Methods, 49(4), 1494-1502. https://doi.org/10.3758/s13428-016-0809-y.
- [305] Welch, B.L. (1947). The generalization of 'Student's' problem when several different population variances are involved. *Biometrika*, 34(1-2), 28-35. https://doi.org/10.1093/biomet/34.1-2.28.
- [306] Hair, J.F., Black, W.C., Babin, B.J., & Anderson, R.E. (2010). *Multivariate data analysis: A global perspective.* Prentice Hall.
- [307] Byrne, B.M. (2010). Structural Equation Modeling with AMOS: Basic Concepts, Applications, and Programming. Taylor and Francis Group Publication.
- [308] Grassini, S., & Laumann, K. (2020). Are modern head-mounted displays sexist? A systematic review on gender differences in HMD-mediated virtual reality. *Frontiers in psychology*, 11, 1604.

- [309] Munafo, J., Diedrick, M., & Stoffregen, T.A. (2017). The virtual reality headmounted display Oculus Rift induces motion sickness and is sexist in its effects. *Experimental brain research*, 235, 889-901.
- [310] Yang, J.C., Quadir, B., Chen, N.S., & Miao, Q. (2016). Effects of online presence on learning performance in a blog-based online course. *The Internet and Higher Education*, *30*, 11-20.
- [311] Lequeux, B., Uzan, C., & Rehman, M.B. (2018). Does resting heart rate measured by the physician reflect the patient's true resting heart rate? White-coat heart rate. *Indian Heart J.* 70(1), 93-98. doi: 10.1016/j.ihj.2017.07.015.
- [312] Kim, J., & Park, T. (2020). The onset threshold of cybersickness in constant and accelerating optical flow. *Applied Sciences*, 10 (21), 7808. https://doi.org/10.3390/app10217808.
- [313] Tian, N., Lopes, P., & Boulic, R. (2022). A review of cybersickness in headmounted displays: raising attention to individual susceptibility. *Virtual Reality*, *26*, 1409–1441. https://doi.org/10.1007/s10055-022-00638-2.
- [314] Kim, H., Kim, D.J., Kim, S., Chung, W.H., Park, K.A., Kim, J.D.K., Kim, D., Kim, M.J., Kim, K., & Jeon, H.J. (2021). Effect of Virtual Reality on Stress Reduction and Change of Physiological Parameters Including Heart Rate Variability in People With High Stress: An Open Randomized Crossover Trial. *Front Psychiatry*, 12, 614539. doi: 10.3389/fpsyt.2021.614539.
- [315] Hoag, J.A., Karst, J., Bingen, K., Palou-Torres, A., & Yan, K. (2022). Distracting Through Procedural Pain and Distress Using Virtual Reality and Guided Imagery in Pediatric, Adolescent, and Young Adult Patients: Randomized Controlled Trial. J Med Internet Res, 24(4), e30260. doi: 10.2196/30260.
- [316] McCaul, K.D., & Malott, J.M. (1984). Distraction and coping with pain. *Psychol Bull*, *95*(3), 516-533. [Medline: 6399756].
- [317] Fassbender, E., & Heiden, W. (2006). The virtual memory palace. *J Comput Inf Syst 2*(1), 457–464.
- [318] Legge, E.L., Madan, C.R., Ng, E.T., & Caplan, J.B. (2012). Building a memory palace in minutes: equivalent memory performance using virtual versus conventional environments with the method of loci. *Acta Psychol.* 141(3), 380–390.
- [319] Krokos, E., Plaisant, C. & Varshney, A. (2019). Virtual memory palaces: immersion aids recall. *Virtual Reality*, 23, 1–15. https://doi.org/10.1007/s10055-018-0346-3.
- [320] Knauf, M. (2013). *Space to reason: a spatial theory of human thought*. MIT Press, Cambridge.
- [321] Godden, D.R., & Baddeley, A.D. (1975). Context-dependent memory in two natural environments: on land and underwater. *Br J Psychol* 66(3), 325–331.

- [322] Slater, M. (2009). Place illusion and plausibility can lead to realistic behaviour in immersive virtual environments. *Philos Trans R Soc Lond B Biol Sci. 364*(1535), 3549–3557.
- [323] Barsalou, L.W. (2008). Grounded cognition. Annu Rev Psychol, 59, 617–645.
- [324] Shapiro, L. (2010). *Embodied cognition*. Routledge, New York. ISBN 978-0415773423.
- [325] Hartley, T., Lever, C., Burgess, N., & O'Keefe, J. (2014). Space in the brain: how the hippocampal formation supports spatial cognition. *Philos Trans R Soc B*. *369*(1635), 20120,510.
- [326] Tomasi, D. (2021). Olfactory virtual reality (OVR) for wellbeing and reduction of stress, anxiety and pain.
- [327] Pizzoli, S.F.M., Monzani, D., Mazzocco, K., Maggioni, E., & Pravettoni, G. (2022). The power of odor persuasion: The incorporation of olfactory cues in virtual environments for personalized relaxation. *Perspectives on Psychological Science*, 17(3), 652-661.
- [328] Abbott, R.W., & Diaz-Artiles, A. (2022). The impact of digital scents on behavioral health in a restorative virtual reality environment. *Acta Astronautica*, 197, 145-153.
- [329] Knight, W.E., & Rickard, N.S. (2001). Relaxing music prevents stress-induced increases in subjective anxiety, systolic blood pressure, and heart rate in healthy males and females. *J Music Ther. Winter*, *38*(4), 254-72. doi: 10.1093/jmt/38.4.254.
- [330] Dadashi, M., Birashk, B., Taremian, F., Asgarnejad, A.A., & Momtazi, S. (2015). Effects of Increase in Amplitude of Occipital Alpha & Theta Brain Waves on Global Functioning Level of Patients with GAD. *Basic Clin Neurosci.* 6(1), 14-20.
- [331] Shuda, Q., Bougoulias, M.E., & Kass, R. (2020). Effect of nature exposure on perceived and physiologic stress: A systematic review. *Complementary Therapies in Medicine*, *53*, 102514.
- [332] Maples-Keller, J.L., Bunnell, B.E., Kim, S.J. & Rothbaum, B.O. (2017). The Use of Virtual Reality Technology in the Treatment of Anxiety and Other Psychiatric Disorders. *Harv. Rev. Psychiatry*, 25(3), 103-113. doi: 10.1097/HRP.00000000000138.
- [333] Oing, T. & Prescott, J. (2018). Implementations of Virtual Reality for Anxiety-Related Disorders: Systematic Review. *JMIR Serious Games*, 6(4), e10965. doi: 10.2196/10965.
- [334] Kim, O., Pang, Y., & Kim, J.H. (2019). The effectiveness of virtual reality for people with mild cognitive impairment or dementia: a meta-analysis. *BMC Psychiatry*, *19*(1), 219. doi: 10.1186/s12888-019-2180-x.

- [335] García-Betances, R.I., Jiménez-Mixco, V., Arredondo, M.T. & Cabrera-Umpiérrez, M.F. (2015). Using virtual reality for cognitive training of the elderly. Am J Alzheimers Dis Other Demen. 30(1), 49-54. doi: 10.1177/1533317514545866.
- [336] García-Betances, R.I., Arredondo, Waldmeyer, M.T., Fico, G. & Cabrera-Umpiérrez, MF. (2015). A succinct overview of virtual reality technology use in Alzheimer's disease. *Front Aging Neurosci.* 12(7), 80. doi: 10.3389/fnagi.2015.00080.
- [337] Manera, V. et al. (2016). A Feasibility Study with Image-Based Rendered Virtual Reality in Patients with Mild Cognitive Impairment and Dementia. *PLoS One*, *11*(3), e0151487. doi: 10.1371/journal.pone.0151487.
- [338] Ugliotti, F.M. (2022). Virtual Representations for Cybertherapy: A Relaxation Experience for Dementia Patients. In Handbook of Research on Implementing Digital Reality and Interactive Technologies to Achieve Society 5.0 (pp. 1-23). IGI Global.
- [339] Syed-Abdul, S. et al. (2019). Virtual reality among the elderly: a usefulness and acceptance study from Taiwan. *BMC Geriatr. 19*, 223. https://doi.org/10.1186/s12877-019-1218-8.
- [340] Niki, K. et al. (2020). Immersive Virtual Reality Reminiscence Reduces Anxiety in the Oldest-Old Without Causing Serious Side Effects: A Single-Center, Pilot, and Randomized Crossover Study. *Front Hum Neurosci.* 14, 598161. doi: 10.3389/fnhum.2020.598161.
- [341] Zhai, K. et al. (2021). Virtual Reality Therapy for Depression and Mood in Long-Term Care Facilities. *Geriatrics*, 6(2), 58. https://doi.org/10.3390/geriatrics6020058.
- [342] https://www.oculus.com/safety-center/quest-2/
- [343] Folstein, M.F., Folstein, S.E., & McHugh, P.R. (1975). "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. *J. Psychiatr. Res.* 12(3), 189-98. doi: 10.1016/0022-3956(75)90026-6..
- [344] Magni, E., Binetti, G., Bianchetti, A., Rozzini, R. & Trabucchi, M. (1996).
 Mini-Mental State Examination: a normative study in Italian elderly population. *Eur. J. Neurol.* 3(3), 198-202. doi: 10.1111/j.1468-1331.1996.tb00423.x.
- [345] Cummings, J.L. et al. (1994). The Neuropsychiatric Inventory: comprehensive assessment of psychopathology in dementia. *Neurology*, 44(12), 2308-14. doi: 10.1212/wnl.44.12.2308.
- [346] Lawton, M.P., Van Haitsma, K., & Klapper J. (1999). Observed Emotion Rating Scale. *Journal of Mental Health and Aging*, *5*, 69–81.
- [347] Roberts, A.E. et al. (2021). Evaluating the quality and safety of health-related apps and e-tools: Adapting the Mobile App Rating Scale and developing a quality assurance protocol. *Internet Interv. 24*, 100379. doi: 10.1016/j.invent.2021.100379.

- [348] Brooke, J. (1996). SUS: a 'quick and dirty' usability scale. In: Jordan PW, Thomas B, Weerdmeester BA, McClelland IL (eds) Usability evaluation in industry. Taylor & Francis, London, pp 189–194.
- [349] Bangor, A., Kortum, P. T. & Miller, J. T. (2008). An empirical evaluation of the System Usability Scale. *International Journal of Human-Computer Interaction*, 24(6), 574–594. https://doi.org/10.1080/10447310802205776.
- [350] Sauro, J. Measuring Usability with the System Usability Scale (SUS). Available online: https://measuringu.com/sus/ (accessed on 07 March 2023).
- [351] Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology, 3*(2), 77–101. https://doi.org/10.1191/1478088706qp0630a.
- [352] Onwuegbuzie, A.J., Dickinson, W.B., Leech, N.L. & Zoran, A.G. (2009). A Qualitative Framework for Collecting and Analyzing Data in Focus Group Research. *Int. J. Qual. Methods*, *8*, 1–21.
- [353] Gabrielli, S. et al. (2022). Multidimensional Study on Users' Evaluation of the KRAKEN Personal Data Sharing Platform. *Applied Sciences*, 12(7), 3270. https://doi.org/10.3390/app12073270.
- [354] Al Zayer, M., Adhanom, I.B., MacNeilage & P., Folmer, E. The effect of fieldof-view restriction on sex bias in VR sickness and spatial navigation performance. In *Proceedings of the 2019 CHI conference on human factors in computing systems*, CHI 2019, 354:1–354:12, New York, NY, USA, ACM.
- [355] Chang, E., Kim, H.T., & Yoo, B. (2020). Virtual reality sickness: a review of causes and measurements. *International Journal of Human–Computer Interaction*, 36(17), 1658-1682. https://doi.org/10.1080/10447318.2020.1778351.
- [356] Kim, J. & Park, T. (2020). The onset threshold of cybersickness in constant and accelerating optical flow. *Applied Sciences*, 10(21), 7808. https://doi.org/10.3390/app10217808.
- [357] Tian, N., Lopes, P., & Boulic, R. (2022). A review of cybersickness in headmounted displays: raising attention to individual susceptibility. *Virtual Reality*, *26*, 1409–1441. https://doi.org/10.1007/s10055-022-00638-2.
- [358] Appel, L., Appel, E., Kisonas, E., Lewis, S. & Qing Sheng, L. (2021). Virtual Reality for Veteran Relaxation: Can VR Therapy Help Veterans Living With Dementia Who Exhibit Responsive Behaviors? *Front. Virtual Real.* 2. https://doi.org/10.3389/frvir.2021.724020.
- [359] Healy, D., Flynn, A., Conlan, O., McSharry, J. & Walsh, J. (2022). Older Adults' Experiences and Perceptions of Immersive Virtual Reality: Systematic Review and Thematic Synthesis. *JMIR Serious Games*, *10*(4), e35802. doi: 10.2196/35802.

- [360] Lorenz, M., Brade, J., Klimant, P., Heyde, C.E., & Hammer, N. (2023). Age and gender effects on presence, user experience and usability in virtual environments-first insights. *PLoS One*, *18*(3), e0283565. doi: 10.1371/journal.pone.0283565.
- [361] https://www.picoxr.com/it/products/pico4/specs?gclid=EAIaIQobChMIsZyqw 5vR_QIVhNzVCh3L7Q5WEAAYASACEgKrU_D_BwE.
- [362] Recchia, G., Capuano, D.M., Mistri, N., & Verna, R. (2020). Digital therapeutics-what they are, what they will be. *Acta Sci Med Sci*, 4(3), 1-9.
- [363] https://www.ibm.com/topics/chatbots (accessed on 08 August 2023).
- [364] Bower, P., & Gilbody, S. (2005). Stepped care in psychological therapies: access, effectiveness and efficiency. Narrative literature review. *Br J Psychiatry*, *186*, 11-7. doi: 10.1192/bjp.186.1.11.
- [365] Eccher, C., Gios, L., Zanutto, A., Bizzarri, G., Conforti, D., & Forti, S. (2020). TreC platform. An integrated and evolving care model for patients' empowerment and data repository. *J Biomed Inform. 102*, 103359. doi: 10.1016/j.jbi.2019.103359.
- [366] Kruse, C. S., Argueta, D. A., Lopez, L., &, Nair, A. (2015). Patient and provider attitudes toward the use of patient portals for the management of chronic disease: a systematic review. *Journal of medical Internet research*, *17*(2), e40.
- [367] Fairbanks, R. J., & Wears, R. L. (2008). Hazards with medical devices: the role of design. *Annals of emergency medicine*, *52*(5), 519-521.
- [368] Gilardi, S., Guglielmetti, C., & Pravettoni, G. (2014). Interprofessional team dynamics and information flow management in emergency departments. *Journal of Advanced Nursing*, 70(6), 1299-1309.

ABBREVIATIONS

VR	Virtual Reality
PMR	Progressive Muscle Relaxation
GI	Guided Imagery
HMDs	Head Mounted Displays
ART	Attention Restoration Theory
APA	American Psychological Association
SRT	Stress Reduction Theory
PFA	Perceptual Fluency Account
CAVE	Cave Automatic Virtual Environment
PQ	Presence Questionnaire
ITC-SOPI	ITC- Sense of Presence Inventory
MAUVE	Multi-criteria Assessment of Usability for Virtual Environments
ORBIT	Obesity-Related Behavioral Intervention Trials
FBK	Fondazione Bruno Kessler
STAI	State-Trait Anxiety Inventory
DASS-21	Depressive Anxiety Stress Scale-21
VREs	Virtual Reality Environments
SAM	Self-Assessment Manikin
VAS	Visual Analogue Scale
VRSQ	Virtual Reality Symptom Questionnaire
cVRE	customized Virtual Reality Environment
ANOVA	Analysis of Variance
COPE-NVI	Coping Orientation to the Problems Experienced-New Italian Version
VVIQ	Vividness of Visual Imagery Questionnaire
TVIC	Test of Visual Imagery Control
PGWBI	Psychological General Well-Being Index
LME	Linear Mixed-Effects
FDR	False Discovery Rate
HR	Heart Rate
SDMs	Substitute Decision Makers
APSP	Azienda Pubblica di Servizi alla Persona
OERS	Observed Emotion Rating Scale
MMSE	Mini-Mental State Examination
NPI	U.C.L.A. Neuropsychiatric Inventory-NPI
A-MARS	Adapting-Mobile App Rating Scale
SUS	System Usability Scale
TReC	Trentino Citizens Clinical Record

ACKNOWLEDGMENTS

I would like to express my deepest gratitude to all those who have supported and contributed to the successful completion of my Ph.D. journey. This thesis represents the culmination of years of dedication and collaboration, and I am profoundly grateful to everyone who has played a part in this endeavour.

First, I extend my heartfelt thanks to my thesis advisors, Professors Caterina Novara and Silvia Gabrielli, for their guidance and insights throughout this research project. Their expertise and encouragement have been instrumental in shaping my ideas and refining my work.

I am grateful to Stefano Forti and Oscar Mayora Ibarra for providing the resources and infrastructure necessary for conducting my research, and for allowing me to continue my research activity in the following years. I extend my gratitude to Marco Dianti for building the VR apparatus and constantly working on my side to deploy the current thesis.

I'd also like to reserve a special thanks to Lucia Ronconi, the librarians, and technical personnel from the University of Padova, who have been instrumental in facilitating my work.

I want to acknowledge the financial support I received during my Ph.D. studies, which made this research possible. I am grateful to the Fondazione Bruno Kessler (Trento), in agreement with the Brain, Mind, and Computer Science Ph.D. program (University of Padova), and the Fondazione VRT (Valorizzazione Ricerca Trentina) that funded my studies and enabled me to focus on my research without financial constraints.

A special thanks to Professor Lora Appel for welcoming me into her research group during my visiting period abroad at York University and University Health Network (Toronto, Canada), and for continuing to give me the opportunity to collaborate together.

I thank my fellow students and colleagues at the University of Padova, Fondazione Bruno Kessler, for their shared experiences. Your support has made this academic journey both enjoyable and enriching.

I would like to express my gratitude to my soul mate's sister-in-law (and colleague), Silvia, for taking an active part in the deployment of the current thesis and for always being with me no matter what happens.

An immeasurable gratitude for always having my family next to me (even if far away). Your belief in me has been my constant motivation.

I want to dedicate this thesis to my brother Federico. You were my example, and you will always be with me.

Last but not least, no words are sufficient to express my gratitude to have Mirco as a partner in my life, whose patience, love, and sacrifices have sustained me during the most challenging phases of this journey (especially during the last months, we know why).

To all whose names I have mentioned and those who may remain unnamed but have contributed in some way, please accept my sincere thanks for being part of this remarkable chapter in my academic life.

With heartfelt gratitude,

Susanna September, 2023